8314732, 2023, 3; Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by Cochranetalia, Wiley Online Library on [0203/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

ADOPTED: 20 January 2023 doi: 10.2903/j.efsa.2023.7866



Risks for human health related to the presence of grayanotoxins in certain honey

EFSA Panel on Contaminants in the Food Chain (CONTAM), Dieter Schrenk, Margherita Bignami, Laurent Bodin, James Kevin Chipman, Jesús del Mazo, Bettina Grasl-Kraupp, Christer Hogstrand, Laurentius (Ron) Hoogenboom, Jean-Charles Leblanc, Carlo Stefano Nebbia, Elsa Nielsen, Evangelia Ntzani, Annette Petersen, Salomon Sand, Tanja Schwerdtle, Christiane Vleminckx, Birgit Dusemund, Andrew Hart, Patrick Mulder, Barbara Viviani, Maria Anastassiadou, Claudia Cascio, Francesca Riolo and Heather Wallace

Abstract

The European Commission asked EFSA for a scientific opinion on the risks for human health of the presence of grayanotoxins (GTXs) in 'certain honey' from Ericaceae plants. The risk assessment included all structurally related grayananes occurring with GTXs in 'certain' honey. Oral exposure is associated with acute intoxication in humans. Acute symptoms affect the muscles, nervous and cardiovascular systems. These may lead to complete atrioventricular block, convulsions, mental confusion, agitation, syncope and respiratory depression. For acute effects, the CONTAM Panel derived a reference point (RP) of 15.3 μ g/kg body weight for the sum of GTX I and III based on a BMDL₁₀ for reduced heart rate in rats. A similar relative potency was considered for GTX I. Without chronic toxicity studies, an RP for long-term effects could not be derived. There is evidence for genotoxicity in mice exposed to GTX III or honey containing GTX I and III, showing increased levels of chromosomal damage. The mechanism of genotoxicity is unknown. Without representative occurrence data for the sum of GTX I and III and consumption data from Ericaceae honey, acute dietary exposure was estimated based on selected concentrations for GTX I and III reflecting concentrations measured in 'certain' honeys. Applying a margin of exposure (MOE) approach, the estimated MOEs raised health concerns for acute toxicity. The Panel calculated the highest concentrations for GTX I and III below which no acute effects would be expected following 'certain honey' consumption. The Panel is 75% or more certain that the *calculated highest concentration* of 0.05 mg for the sum of GTX I and III per kg honey is protective for all age groups regarding acute intoxications. This value does not consider other grayananes in 'certain honey' and does not cover the identified genotoxicity.

© 2023 European Food Safety Authority. *EFSA Journal* published by Wiley-VCH GmbH on behalf of European Food Safety Authority.

Keywords: grayanotoxins, grayananes, Rhododendron honey, Ericaceae

Requestor: European Commission Question number: EFSA-Q-2020-00509 Correspondence: contam@efsa.europa.eu **Panel members:** Margherita Bignami, Laurent Bodin, James Kevin Chipman, Jesús del Mazo, Bettina Grasl-Kraupp, Christer Hogstrand, Laurentius (Ron) Hoogenboom, Jean-Charles Leblanc, Carlo Stefano Nebbia, Elsa Nielsen, Evangelia Ntzani, Annette Petersen, Salomon Sand, Dieter Schrenk, Tanja Schwerdtle, Christiane Vleminckx and Heather Wallace.

Declarations of interest: If you wish to access the declaration of interests of any expert contributing to an EFSA scientific assessment, please contact interestmanagement@efsa.europa.eu.

Acknowledgements: The CONTAM Panel wishes to thank Federico Cruciani for the support provided to this scientific output. The Panel wishes to acknowledge all Member State bodies and other organisations that provided data for this scientific output.

Suggested citation: EFSA CONTAM Panel (EFSA Panel on Contaminants in the Food Chain), Schrenk D, Bignami M, Bodin L, Chipman JK, del Mazo J, Grasl-Kraupp B, Hogstrand C, Hoogenboom L, Leblanc J-C, Nebbia CS, Nielsen E, Ntzani E, Petersen A, Sand S, Schwerdtle T, Vleminckx C, Dusemund B, Hart A, Mulder P, Viviani B, Anastassiadou M, Cascio C, Riolo F and Wallace H, 2023. Scientific Opinion on the risks for human health related to the presence of grayanotoxins in certain honey. EFSA Journal 2023;21(3):7866, 120 pp. https://doi.org/10.2903/j.efsa.2023.7866

ISSN: 1831-4732

© 2023 European Food Safety Authority. *EFSA Journal* published by Wiley-VCH GmbH on behalf of European Food Safety Authority.

This is an open access article under the terms of the Creative Commons Attribution-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited and no modifications or adaptations are made.

EFSA may include images or other content for which it does not hold copyright. In such cases, EFSA indicates the copyright holder and users should seek permission to reproduce the content from the original source.



The EFSA Journal is a publication of the European Food Safety Authority, a European agency funded by the European Union.



18314732, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [02/03/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [02/03/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [02/03/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [02/03/2023].

Summary

Following a request from the European Commission, the European Food Safety Authority (EFSA) Panel on Contaminants in the Food Chain (CONTAM Panel) evaluated the risks for human health related to the presence of grayanotoxins (GTXs) in honey from Ericaceae plants, herein reported as 'certain honey'. This risk assessment was extended to cover all structurally related grayananes co-occurring with GTXs in 'certain honey' and, if available, other foods.

Grayananes are polyhydroxylated diterpenes that are produced by some species of the Ericaceae family, particularly in the genera of *Agarista*, *Craibiodendron, Kalmia, Leucothoe, Lyonia, Pieris* and *Rhododendron*. Presently, approximately 180 naturally occurring grayananes have been identified. Oral exposure to grayananes is associated with acute intoxication in humans. Acute symptoms affect the muscles and the nervous and cardiovascular systems with bradycardia and hypotension being most prominent. These may lead to complete atrioventricular block, convulsions, mental confusion, agitation, syncope and respiratory depression.

Grayananes can be present in the nectar of the flowering plants and when collected by honeybees, they can accumulate in honey. Some grayanane-producing plant species are used for commercial honey production, the most important for this opinion being the *Rhododendron* species *R. ponticum*, *R. luteum* and *R. ferrugineum*. The first two species are particularly abundant in Turkey and the latter species occurs in the Alps and Pyrenees. Other *Rhododendron* or Ericaceae species may be used for honey production as well but not all of them have been described to contain grayananes (e.g. Heather honey). Non-*Rhododendron* honey from *Agarista salicifolia* is associated with grayanane acute intoxications in humans on the French island of La Réunion.

Approximately 20 grayananes have been identified in or associated with honey. GTX I and GTX III are the grayananes most studied with respect to toxicological effects as well as their occurrence in 'certain honey', but other analogues, such as rhodojaponin III, may also contribute to the effects observed.

Oral toxicokinetic studies with GTX I and III in experimental animals are scarce and do not allow conclusions on the bioavailability of GTX I and III following exposure to 'certain honey'. In case reports on human intoxications related to consumption of 'certain honey', GTX I and GTX III were detected in human blood and urine, showing that they are absorbed in the GI-tract.

The Panel noted that most toxicodynamic studies on experimental animals use i.p. injection. However, no toxicokinetic study dealing with i.p. injection of grayananes in experimental animals was identified and, therefore, it was not possible to compare the grayanane bioavailability after oral and *i.p.* exposure to `certain honey' or the pure substance(s).

No information on the biotransformation of grayananes in mammals is available to EFSA. Based on data on the *ent*-kaurane diterpenoid oridonin, no metabolic cleavage of the grayanane rings is anticipated.

Acute animal toxicity studies reported cardiovascular effects, decreased respiratory rate and impairment of the nervous system shortly after exposure to grayananes or 'certain honey'. Among these, blood pressure reduction and bradycardia were most representative of the symptoms of human intoxication.

Based on their structure–activity relationships, grayananes can be divided into four groups. The two most potent groups showed three structural alerts, and the number of alerts is inversely proportional to the LD_{50} values. Relative potency factors (RPFs) were derived for each group as follows: 2 for Group 1A, 1 for Group 1B, 0.1 for Group 2 and < 0.01 (negligible potency) for Group 3. A potency factor of 1 was assigned to GTX I and III which have similar toxicity.

The limited information from acute, subacute and subchronic studies in rodents suggests that several biochemical parameters are altered following exposure to individual grayananes and/or 'certain honey', sometimes in a dose-dependent manner. None of these changes, however, were considered as toxicologically relevant because the magnitude of the changes was small. In most of these studies, when examined, no histological changes were reported.

From single dose studies in experimental animals, a $BMDL_{10}$ of 15.3 μ g/kg bw was derived, based on the reduction of heart rate after i.p injection of GTX III in rats. This reference point (RP) was applied to the sum of GTX I and III, considering that these two substances have similar toxic potency.

No RP for subacute and/or subchronic exposure could be identified.

Histological findings in rat testis from repeated dose i.p. injection of GTX III at doses from 0.1 to 0.8 μ g/kg bw suggest potential effects on the testes, but the available data are limited.

Induction of chromosomal damage as shown in an *in vivo* micronucleus (MN) test in male mice indicates genotoxicity for GTX III and honey containing GTX I and III. In the same honey sample, the analysis of chromosomal aberrations confirmed this result. The mechanism of genotoxicity is unknown.

With regard to observations in humans, case reports are available on intoxications due to ingestion of grayananes containing honeys, plants, plant parts or their preparations. Focus is put on reports with relevance for the risk assessment giving information on dose effect relationships after oral intake of 'certain honey'.

Acute intoxications reported in the last decades from European countries are mainly associated with imported honey from Turkey or Nepal, but some have also been caused by honey from other countries. No intoxication cases have been reported for honeys from EU origin.

Rhododendron honeys have a traditional use as folk medicine, e.g. against gastrointestinal disorders, hypertension, heart failure, diabetes, asthma, cough and as an aphrodisiac.

Known amounts of ingested *Rhododendron* honey associated with acute intoxication in humans vary and are reported to range from 5 to 150 g with levels of GTX I and/or GTX III ranging from 0.5 to 54 mg/kg. The majority of intoxications are reported to refer to males aged between 40 and 65 years, the higher prevalence being explained by the higher use of 'certain honey' as a sexual stimulant as well as a higher rate of hypertension in males of this age group. It was noted, however, that not all reported intoxications relate to the use of *Rhododendron* honey as folk medicine but cover also consumption of honey as conventional food.

Resulting symptoms on the nervous and cardiovascular systems occur within minutes and up to 5 hours. Poisoning symptoms may persist from a few hours to a few days. In general, patients recover within 1–2 days.

Even though more than 1,000 cases of intoxication after ingestion of 'certain' Ericaceae honeys by individuals were retrieved in published literature only a few of these case reports provide quantitative information on the content of grayananes in the honey.

In acute poisoning cases observed with *Rhododendron* honeys, the estimated intake on a single occasion of GTX I or GTX III as part of the mixture of grayananes present in the honey range from 0.6 to 17 μ g/kg bw or 3.9 to 24 μ g/kg bw, respectively, and from 4.8 to 40 μ g/kg bw for the sum of GTX I and GTX III. A possible contribution of other grayananes to the development of adverse effects cannot be quantitatively assessed due to insufficient information on their occurrence and on their toxicological profile. From cases of intoxications following the ingestion of *Rhododendron* honey, in which GTX I and III could not be detected, it can be concluded that other grayananes also cause adverse effects on the muscles and the nervous and cardiovascular systems. An NOAEL for any GTX or for total grayananes could not be identified from the data available for human case studies.

For human chronic exposure, reported symptoms do not differ from those of acute exposure. However, information on dose–effect relationships for long-term intake of grayananes could not be identified.

Results from one study suggest that patients with cardiovascular diseases, when receiving antiischaemic or antihypertensive therapy (e.g. beta-blockers), are at higher risk of developing adverse cardiovascular effects when exposed to grayanotoxins compared to healthy subjects.

The mechanism of acute intoxication relates to grayanane binding to sodium voltage gated channels in their open state and preventing their inactivation. This results in a persistent activation of excitable cells like neurons, muscle cells, cardiomyocytes and is consistent with the promotion of acetylcholine release from the vagus nerve. The use of specific sodium channel blockers and antagonists of acetylcholine receptors counteracts grayananes toxicity, supporting the involvement of these mechanisms in their toxicity.

As a result of the indicated *in vivo* genotoxicity for the pure GTX III and for one honey sample containing GTX I and III, a health-based guidance value (HBGV) cannot be derived for grayananes. Due to the lack of information on the underlying mode of action of genotoxicity and the lack of data on chronic toxicity and carcinogenicity, the Panel was unable to assess the risk related to chronic/ repeated exposure.

In addition, due to the lack of specific occurrence data on grayananes, the lack of consumption data for *Rhododendron* honey and/or other honeys containing grayananes in the EFSA databases and the scarcity of data in the literature, dietary exposure was estimated based only on selective concentrations for GTX I and III in 'certain honey'.

No information was available regarding GTX occurrence in foods other than honey in the EU.

Regarding the risks for acute toxicity, the Panel assessed the health concern using the MOE approach. MOEs were calculated based on the RP of 15.3 μ g/kg bw for the sum of GTX I and III and

18314732, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [0203/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

the exposure at selected concentrations reflecting the highest measured concentration for GTX I in a honey sample of EU origin (0.1 mg/kg) and concentrations measured for the sum of GTX I and III in 'certain honeys' (1 and 10 mg/kg). The estimated MOEs were below 100, and thus raised health concerns for acute toxicity.

To estimate the *highest concentration(s)* for the sum of GTX I and III in 'normal' honey below which no acute effects on heart rate and blood pressure would be expected following acute consumption of 'certain honey', the highest reported P95 intake of honey consumers was combined with the RP of 15.3 μ g/kg bw for the sum of GTX I and III and an MOE of 100 or greater was applied.

For toddlers, the lowest value was estimated as 0.05 mg GTX I and III/kg honey based on the P95 consumption of 3 g honey per kg bw. This should protect also other age groups with equal or lower relative honey consumption.

When *Rhododendron* honey is consumed by adults as traditional folk medicine, the Panel calculated *highest concentrations* for the sum of GTX I and III based on portions of *Rhododendron* honey from labels of products sold on the Internet and originating from Nepal or Turkey. The *highest concentration* was estimated to be 0.16 mg GTX I and III/kg honey for the general adult population and corresponds to consumption of 3 tablespoons (66 grams) of honey. It was noted that the honey portions on the products' labels used in the calculations are specific to these products and are not necessarily representative of all commercially available products. In addition, it is uncertain if all consumers follow the recommendations on the products' labelling.

Uncertainty analysis was performed to identify and quantify sources of uncertainty affecting the risk assessment and combine them to assess the overall certainty of the main conclusions. Sources of uncertainties related to hazard identification and characterisation of grayananes and consumption of honey containing grayananes in the EU were listed and described. Since the Panel did not identify non-standard uncertainties in the genotoxicity assessment, no further uncertainty analysis was undertaken for this.

The Panel quantified by expert judgement the combined impact of the identified sources of uncertainty on the RP for acute toxicity, the relative potency of other grayananes and the P95 of honey consumption for EU toddlers and adults.

Based on these judgements, the Panel is 85% certain that the RP is protective, i.e. would not be lower if the uncertainties affecting it were resolved, and 70–95% certain that the proposed RPFs are conservative.

The Panel is 75% or more certain for toddlers and adults that the MOEs presented in the Opinion would not be lower if the uncertainties relating to the RP for acute toxicity and exposure based on selected concentrations and the 95th percentile of acute consumption in the EFSA database were resolved. This conclusion is also applicable for all the other age groups.

The Panel is also 75% or more certain that the *highest concentration* of 0.05 mg/kg in honey calculated for toddlers is protective for all age groups, i.e. will not result in a concern for acute effects at the 95th percentile of acute consumption. The Panel is 85% certain that, when *Rhododendron* honey is consumed as folk medicine, the *highest concentration* of 0.16 mg/kg, based on a portion size of 66 g, would not lead to a concern for acute effects. It was noted that larger portion sizes would result in proportionately lower *highest concentrations*.

These conclusions do not cover the genotoxicity endpoint. In addition, they do not consider the impact of other co-occurring grayananes in 'certain honey' which cannot be analysed presently. When analytical standards for other grayananes become available, it may be possible to refine the assessment and include those.

Recommendations

The following needs have been identified to improve the risk assessment for humans and reduce the uncertainties:

- Analytical standards are required for grayananes other than GTX I and III to enable accurate measurement of these substances in 'certain honey' and other foods, in particular for rhodojaponin III.
- Validated analytical methods should be developed for the quantification of total grayananes in 'certain honey' with LOQs per grayanane at 0.01 mg/kg or lower.
- There is a need for occurrence data on grayananes in honey and commodities thereof from Ericaceae species. Quantitative data should be collected for at least GTX I and GTX III.

18314732, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.efsa.2023,7866 by CochraneItalia, Wiley Online Library on [0203/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

- Integration of the monitoring of grayananes in *Rhododendron* honeys on the EU market in the national control activities is recommended, especially from specific regions of production (e.g. Alps and Pyrenees).
- Investigations of the market share of *Rhododendron* honeys in the EU market are needed.
- Mechanistic studies are required to understand the mode of action of genotoxicity.
- The short- and long-term effects of orally administered grayananes on the reproductive system need to be investigated with single and repeated-dose toxicity studies across a broader dose range (> 0.8μ g/kg bw).
- Experimental studies via the oral route on the effects of acute and chronic exposure/ carcinogenicity to grayananes are required.
- Observations/surveys in humans on the acute effects of grayananes in the low-dose range and on the NOAEL are desirable.

Table of contents

Abstract		1
Summary	/	3
Recomme	endations	5
1.1.	Background and Terms of Reference as provided by the requestor	9
111	Interpretation of the Terms of Reference	9
1 2	Supporting information for the assessment	ģ
121	Chemistry	ģ
122	Sources	13
1221	Gravananes in plants	13
1 2 2 2	Gravananes in Joney	15
1 2 2 3	Gravananes in hotery sources other than honey	16
12.2.3.	Sampling and methods of analysis	17
1231	Extraction and clean-un	17
1 2 2 2	Detection / quantification methods	17
1.2.3.2.	Provious rick accossments	17
1.2.7.		10
1.2.5. ว	Data and Methodologies	10
2.	Hazard identification and characterication	10
2.1.	Additu iueiittiinduoni dhu thididtteriisduon	19
2.2.		19
2.3.		20
2.4.		21
2.5.	Exposure assessment	21
2.6.	Risk characterisation	21
3.	Assessment	21
3.1.	Hazard identification and characterisation	21
3.1.1.	Ioxicokinetics	21
3.1.2.	Toxicity in experimental animals	24
3.1.2.1.	Acute single-dose toxicity studies	24
3.1.2.2.	Repeated dose toxicity studies	31
3.1.2.3.	Genotoxicity studies	32
3.1.3.	Observations in humans	33
3.1.3.1.	Case reports on acute intoxications due to ingestion of 'certain honey'	33
3.1.3.2.	Studies on cases of acute intoxications due to ingestion of 'certain honey'	40
3.1.3.3.	Case reports on acute intoxications due to ingestion of certain plants, their parts or preparations	41
3.1.3.4.	Clinical studies	42
3.1.3.5.	Summary of observations in humans	42
3.1.4.	Mode of action	43
3.1.5.	Relative potencies of grayananes for acute toxicity	44
3.1.6.	Considerations of critical effects and dose-response analysis	45
3.1.6.1.	RP based on acute/single-dose exposure to grayananes	45
3.1.6.2.	RP based on repeated dose exposure to grayananes	46
3.1.7.	Consideration of a Health-Based Guidance Value (HBGV) and a margin of exposure	47
3.2.	Occurrence data	47
3.2.1.	Previously reported occurrence data in the open literature	47
3.2.1.1.	Grayanane occurrence in honeys from the EU	47
3.2.1.2.	Grayanane occurrence in non-European honeys	48
3.2.1.3.	Grayanane occurrence in other foods	49
3.2.2.	Processing of honey	54
3.3.	Dietary exposure assessment for humans	54
3.4.	Risk characterisation	54
3.4.1.	Risk characterisation of acute effects	54
3.4.1.1.	Margin of exposure approach	54
3.4.2.	Risk characterisation for genotoxic/chronic effects	55
3.4.3.	Non-dietary sources of exposure	55
3.5.	Estimation of the highest concentrations of the sum of GTX I and III in 'certain honey' below which	
	no acute effects are expected	55
3.5.1.	Scenario A: Consumption of <i>Rhododendron</i> honey as 'normal' honey	56
3.5.2.	Scenario B: Consumption of <i>Rhododendron</i> honey as traditional folk medicine in adults	56
3.6.	Uncertainty analysis.	56
4.	Conclusions.	59
4.1	General	59
		55

4.2.	Hazard identification and characterisation	59
4.2.1.	Toxicokinetics	59
4.2.2.	Single-dose toxicity studies in experimental animals	59
4.2.3.	Repeated-dose toxicity studies in experimental animals	60
4.2.4.	Genotoxicity	60
4.2.5.	Observations in humans	60
4.2.6.	Mode of action	60
4.2.7.	Critical effects and dose-response analysis	60
4.2.8.	Occurrence	61
4.2.9.	Dietary exposure assessment	61
4.2.10.	Risk characterisation	61
4.2.11.	Derivation of the highest levels in honey below which no acute effects are expected	61
4.2.12.	Uncertainty analysis	62
5.	Recommendations	62
Documen	tation as provided to EFSA	63
Reference	25	63
Abbreviat	ions	71
Appendix	A – Grayananes considered in the opinion	73
Appendix	B – Other toxicological studies in experimental animals	98
Appendix	C – BMD analysis	103
Appendix	D – Uncertainty analysis	108
Appendix	E – Lists of uncertainties	113
Annex A	- Protocol	120
Annex B	- Plant list of Ericaceae species	120
Annex C	- Consumption data statistics	120
Annex D	– Uncertainty analysis: EKE Protocol	120

1. Introduction

1.1. Background and Terms of Reference as provided by the requestor

BACKGROUND

Grayanotoxins are produced by plants in the Ericaceae family. While many species of this family contain grayanotoxins, only a few contain significant levels. Bees that collect pollen and nectar from these grayanotoxin-containing plants often produce honey that also contains grayanotoxins. Consumption of this honey may cause grayanotoxin poisoning in humans. As the presence of grayanotoxins is related to very specific honeys, not widely produced, measures taken in certain Member States at national level appear to be effective in avoiding poisoning cases in the EU. However, it is appropriate that EFSA performs a risk assessment in order to consider the need for taking additional risk management measures at EU level to ensure a high level of public health protection.

TERMS OF REFERENCE

In accordance with Art. 29 (1) of Regulation (EC) No 178/2002, the European Commission asks the European Food Safety Authority to provide an opinion on the risks for human health related to the presence of grayanotoxins in certain honey.

1.1.1. Interpretation of the Terms of Reference

The scientific opinion on grayanotoxins (GTXs) in certain honey shall cover the potential risks for human health following exposure to honey and honey products originated from Ericaceae family plants which contain GTXs and other structurally related substances of concern (i.e. grayananes of concern).

For the purpose of the present assessment, the term 'grayananes' shall refer to GTXs and other structurally related substances of concern and the term 'certain honey' shall refer to honey and honey products originated from Ericaceae family plants and containing grayananes. The terms 'mad honey' or 'toxic honey' shall refer to honey containing grayananes in sufficient amounts to cause intoxication in humans.

In particular, the CONTAM Panel should address the questions by the requestor as follows:

- Identify the grayananes of concern present in honey and honey products;
- Evaluate the toxicity of grayananes of concern present in 'certain honey' (i.e. honey and honey products originated from Ericaceae family plants);
- Evaluate the composition and levels of grayananes of concern in 'certain honey' based on data from monitoring and the literature, including information on occurrence and consumption of 'certain honey' from countries outside the EU;
- Estimate the dietary exposure of the EU population to grayananes of concern present in 'certain honey';
- Assess the human health risks for the EU population, including the consumption patterns of honey containing grayananes by specific groups of the population;
- If there is information on the presence of grayananes in other food products originating from the Ericaceae family, this will be assessed as appropriate.

In the absence of representative data for grayanane occurrence in honeys marketed in the EU, and providing the available data will allow the derivation of health-based guidance value(s) (HBGVs) for these substances, a backward calculation will be performed in order to estimate the *highest concentrations* of grayananes per eating portion that can be present in 'certain honey' below which no health effects are expected.

1.2. Supporting information for the assessment

1.2.1. Chemistry

Grayananes are polyhydroxylated diterpenes belonging to the chemical group of grayanoids. Grayanoids are present in Ericaceae plants. The most important genera producing these substances are *Agarista*, *Craibiodendron*, *Kalmia*, *Leucothoe*, *Lyonia*, *Pieris* and *Rhododendron* (Li et al., 2013, 2019; These et al., 2015). Presently, approximately 300 natural grayanoids have been isolated from Ericaceae plants (Li et al., 2013, 2019).

18314722, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023 7866 by CochraneItalia, Wiley Online Library on [0203/023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

The biogenetic precursor of grayanoids is assumed to be *ent*-kaurane (Li et al., 2013, 2019; Hanson, 2016). Studies on the biosynthesis of grayanoids suggest at least six possible bio-transformations of *ent*-kaurane resulting in the formation of six groups of substances: grayananes, 3,4-seco-grayananes, 9,10-seco-grayananes, 1,5-seco-grayananes, kalmanes and leucothanes (Li et al., 2013) (Figure 1).



Figure 1: Simplified scheme for the formation and diversification of grayanoids in plants (Adapted from Li et al., 2013)

The substances of interest for this assessment belong to the group of grayananes and constitute the largest group with over 180 structures described (Li et al., 2013, 2019). Grayananes have as basic structure an A-nor-B-homo-*ent*-kaurane skeleton. Their structure is characterised by a *trans* junction between rings A and B and a *cis* junction between rings B and C. Their stereochemistry is shown in Figure 2. Grayananes may also contain additional structural groups which influence their toxicological properties and potency (see Appendix A and Section 3.1.5).

Based on their structure, grayananes are further divided to the following subgroups: non-epoxidised grayananes, 2,3-epoxy-grayananes and 5,9-epoxy-grayananes (Li et al., 2013). GTXs belong to the subgroup of non-epoxidised grayananes. The detailed structures, synonyms, sources and chemical properties of grayananes are shown in Appendix A.

efs IOURNAL



 $\label{eq:response} \begin{array}{l} \mathsf{R} = \mathsf{H}, \, \mathsf{OH}, \, \mathsf{OCOR'} \; (\mathsf{R'=ester}) \\ \mathsf{R}^* = \mathsf{CH}_3 \text{+} \mathsf{OH} \; \mathsf{or} \; \text{=} \mathsf{CH}_2 \\ \mathsf{R}^{**} \text{=} \; \mathsf{H}, \; \mathsf{OH} \; \mathsf{or} \; \text{epoxy ring} \end{array}$

Figure 2: Generalised structure and stereochemistry of grayananes under assessment

The nomenclature of grayananes is complex and ambiguous. Thus, structurally similar grayananes are reported with various names such as grayanotoxins, asebotoxins, rhodojaponins, andromedotoxins, andrometodols, pieristoxins, etc. (Li et al., 2013; These et al., 2015). The naming in most cases refers to the plant species from which the compound was first isolated and structurally described. Almost all numbered grayananes have been first described for *Leucothoe grayana* (Li et al., 2013).

Some grayanane names, such as andromedotoxin/rhodotoxin, lyoniatoxin or rhomotoxin, have been discontinued, because later research found them to be identical to grayananes described earlier from other species. GTX I was the first grayanane isolated in pure form (Tallent et al., 1957). Its molecular structure was elucidated in 1970 by X-ray crystallography (Narayanan et al., 1970).

It has been reported that oral exposure to plants or plant parts containing grayananes may result in acute intoxications in animals and humans. Bees pollinating grayanane-containing Ericaceae may also produce honey containing grayananes, causing intoxication in humans (see Section 3.1.3).

Additionally, grayananes have been described to exhibit antifeedant and insecticidal properties (Li et al., 2017). The Panel noted that patent applications for the use of Chinese azalea extract (CN1218635C, 2001¹) and a preparation of rhodojaponin III with chlorpyrifos (CN101124911B, 2007²) for use as insecticide have been previously submitted in China. In 2016, an application for the use of grayanane-containing honey as rodenticide was submitted to the European Commission and considered by EFSA (EFSA, 2017). This is discussed in Section 1.2.4.

For the purpose of the present assessment only grayananes which occur in honey and honey products will be evaluated. If, however, there is information on the presence of grayananes in other food products originating from Ericaceae, this will be assessed where appropriate.

Appendix A reports the compounds of Ericaceae origin identified to date and includes, for each compound, the trivial, IUPAC names and synonyms, physical properties and Ericaceae species. For most of these substances, no data on occurrence in honey or other dietary sources are available. Substances which have been identified or tentatively associated with honey (see Section 3.2) are shown in Table 1.

The assessment does not consider synthetic compounds derived from the grayanane backbone.

Grayanane ^(a) Molecular formula CAS no.	Structure	Grayanane ^(a) Molecular formula CAS no.	Structure
A. Grayananes for w	which reference standards are com	mercially available ^(b)	
GTX I C ₂₂ H ₃₆ O ₇ 4720-09-6		GTX III C ₂₀ H ₃₄ O ₆ 4678-45-9	

Table 1:	Grayananes	identified or	tentatively	associated	with	honey

18314722, 2023, 3, Downloaded from https://efas.onlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library on [02/03/202]. See the Terms and Conditions (https://nlinelibrary.wiejc.com/doi/10.2903j.efas.2023.7866 by Cochaneltaia, Wiejc Online Library.wiejc.2023.7866 by Cochaneltaia, Wiej

¹ https://patents.google.com/patent/CN1218635C/en#citedBy

² https://patents.google.com/patent/CN101124911B/en

Grayanane ^(a)		Grayanane ^(a)	
Molecular formula CAS no.	Structure	Molecular formula CAS no.	Structure
B. Grayananes for w honey ^(c)	hich reference standards are not	commercially available,	but which have been identified in
GTX II ^(d) C ₂₀ H ₃₂ O ₅ 4678-44-8		GTX VII C ₂₀ H ₃₀ O ₄ 30460-59-4	
GTX VIII C ₂₀ H ₃₀ O ₄ 30460-60-7		GTX XVII C ₂₀ H ₃₀ O ₆ 59740-27-1	
GTX XVIII C ₂₀ H ₃₂ O ₄ 70474-76-9		Rhodomollein XIX $C_{20}H_{30}O_5$ 629615-33-4	HO HOHOHOH
Rhodojaponin III ^{(e),(f)} $C_{20}H_{32}O_6$ 26342-66-5	HO H HO OH OH		
C. Grayananes for w honey ^(c)	hich reference standards are not	commercially available,	but which are associated with
GTX IV C ₂₂ H ₃₄ O ₆ 30272-17-4	HO HO HO OAc	GTX VI C ₂₀ H ₃₂ O ₅ 30460-36-7	
GTX XVI C ₂₂ H ₃₄ O ₆ 59236-87-2		Craiobiotoxin I C ₂₀ H ₃₂ O ₅ 864679-89-0	но н
Craiobiotoxin II C ₂₀ H ₃₂ O ₅ 864679-90-3	HO H HO OH HO OH	Craiobiotoxin V C ₂₀ H ₃₂ O ₅ 864679-93-6	
Craiobiotoxin VIII $C_{20}H_{32}O_5$ 864680-12-6	HO H HO HOH	Kalmitoxin I C ₂₀ H ₃₄ O ₇ 56663-60-6	

Grayanane ^(a) Molecular formula CAS no.	Structure	Grayanane ^(a) Molecular formula CAS no.	Structure
Rhodojaponin VI C ₂₀ H ₃₄ O ₇ 37720-87-9	HO,	Rhodomollein XIII C ₂₂ H ₃₆ O ₇ 54781-72-5	
Rhodomollein XVIII $C_{20}H_{34}O_7$ 292164-20-6		Rhodomollein XIV C $_{20}H_{34}O_7$ 798558-84-6	

(a): For information on the botanical sources of grayananes and their physico-chemical properties, see Appendix A.

(b): Recently, a reference standard on rhodojaponin II became commercially available.

(c): Based on the elementary composition calculated for compounds present in a sample of contaminated *Rhododendron* honey from the Black Sea region in Turkey, These et al. (2015) suggest a list of possible grayanane analogues: In group B, grayanane analogues are shown that have an elementary composition matching with that of one or more of the chromatographic peaks obtained by liquid chromatography coupled with high-resolution mass spectrometry (LC-HRMS) analysis of the sample and for which in the literature only one plausible grayanane analogue is known for that specific elementary composition (or two analogues in case two peaks with the same elementary composition are present). In group C, compounds are shown that have an elementary composition matching with that of one or more of the peaks, but for which more than one plausible grayanane analogue is known. For these compounds, more information is required to ascertain which analogue is actually present in the sample. See also Appendix A for information on grayananes identified in Ericaceae spp.

(d): Structure confirmed based on Lee et al. (2008).

- (e): Synonyms: Rhomotoxin, desacetylpieristoxin B.
- (f): Based on Scott et al. (1971). These et al. (2015) did not identify this substance in the honey sample they analysed.

1.2.2. Sources

1.2.2.1. Grayananes in plants

Grayananes have been found in plants or parts of plants (leaves, berries, flower nectar and pollen) of some, but not all, genera of Ericaceae, such as *Agarista, Craibiodendron, Kalmia, Leucothoe, Lyonia, Pieris* or *Rhododendron* (Gunduz et al., 2008; EFSA, 2009; Li et al., 2013, 2019; These et al., 2015; Hanson, 2016). Many of these species are also cultivated as ornamental plants (Koca and Koca, 2007; Gunduz et al., 2008). *Rhododendron* is a very large genus, comprising over 1,000 species³ (Egan, 2015; Shrestha et al., 2018). The other genera comprise between 5 and 40 species.

Geographical origin of grayanane-containing species

An overview (non-exhaustive list) of Ericaceae species that have been investigated with respect to the presence of grayananes is presented in Annex B. The large majority of toxic Ericaceae species are found in the northern hemisphere, in Europe, Asia and North America. Only a limited number of grayanane-containing species are found in Africa, South America and Australia. The largest diversity of grayanane-containing species is present in China and the Himalayan region, followed by south-east Asia and Japan (Shrestha et al., 2018).

Several *Kalmia* species that occur in North America, such as *K. latifolia* (mountain laurel) and *K. angustifolia* (sheep laurel) are well known to cause intoxications in sheep and goats when foraged (Furbee and Wermuth, 1997; Puschner et al., 2001).

In the French island of La Réunion, grayananes have been found in a number of endemic plant species, especially *Agarista salicifolia*.⁴

In Turkey, *Rhododendron* species *R. luteum* and *R. ponticum* are the most common toxic species (Gunduz et al., 2008) and both grow in the Black Sea region (Turkey, 2022 under Documentation as provided to EFSA). *R. ponticum* is particularly invasive (Koca and Koca, 2007).

18314732, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by Cochranettalia, Wiley Online Library on [02/03/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons

³ Global Biodiversity Information Facility (GBIF) website, https://www.gbif.org/species/2883026

⁴ Préfet de la Réunion. Direction de l'alimentation, de l'agriculture et de la foret, Service de l'alimentation. Arrêté préfectoral n° 2013-1977. Réglementant la vente de miel en gaufres à la Réunion. http://administration.reunion.gouv.fr/spip.php?article2460

In Europe, *R. ponticum* grows naturally in the high and humid mountains of Spain and Portugal. In France, *R. ponticum* mostly develops in the north-western part of the country and Aquitaine (Manceau, 2015). In the UK and Ireland, *R. ponticum* was introduced as ornamental plant in the late 1800s (Egan et al., 2016). Although the native *R. ponticum* in Spain and Portugal are endangered species, the introduced populations in Great Britain and Ireland are invasive.

Levels of grayananes in plants

The nature and concentrations of grayananes vary considerably from one species to another, but also within and between populations of the same species.

Sahin (2014) studied the GTX III content of *R. ponticum* and *R. luteum* leaves and flowers originating from the Turkish Black Sea region and found that *R. luteum* contained considerably higher GTX III concentrations than *R. ponticum*, in both flowers and leaves (See Table 2 and data submitted by Turkey, 2022 under Documentation as provided to EFSA).

Table 2: GTX III concentrations in flowers and leaves of *R. ponticum* and *R. luteum* in different locations of the Black Sea Turkish region

Rhododendre	on ponticum	Rhododendre	on luteum
Flower (mg/kg dw ^(a))	Leaf (mg/kg dw)	Flower (mg/kg dw)	Leaf (mg/kg dw)
< LOQ (=0.04) – 22.9	< LOQ (=0.04) – 36.3	1.74–760	3.35–154

(a): dry weight.

R. ponticum cultivated in botanical gardens in Germany is reported to contain GTX I from 1,300 to 2,300 mg/kg fresh weight (Lechtenberg et al., 2014).

In a study on *R. ponticum* nectar from different geographical origins, the GTX I concentrations in the Iberian nectars were found up to 5,670 mg/kg on a dry weight basis, which are much higher than concentrations reported in the Turkish honey from the same species (Egan et al., 2016). According to Stout (2022, see Documentation as provided to EFSA), the lower levels of GTX I in the Turkish *Rhododendron* nectar might explain why honeybees are able to tolerate the toxins and produce honey in Turkey. Another explanation may relate to the ability of Turkish subspecies of honeybees to metabolise/detoxify the toxins. There is presently no evidence, however, to confirm these statements.

When comparing the GTX concentrations between nectars of *R. ponticum* from introduced populations in Ireland (mostly chemotype B, GTX III dominant) and the native populations in Portugal and Spain (chemotype A, GTX I dominant), no statistically significant variations in the levels of GTX III were observed. On the other hand, the concentrations of GTX I in the introduced populations were found to be significantly lower than in the native populations (mean 810 mg/kg dw for the Irish populations vs. 1,460 mg/kg dw for the Iberian populations) (Egan et al., 2016).

EFSA did not identify studies on the occurrence of grayananes in *Rhododendron* pollen; therefore, the grayanane content in *Rhododendron* pollen is unknown.

Rhododendron species with low or not known content of grayananes

R. ferrugineum grows at high altitudes in the Alps and Pyrenees and is used for the production of honey in the EU. Seephonkai et al. (2021) did not detect grayananes in *R. ferrugineum* extracts analysed by thin layer chromatography (TLC) (limit of detection (LOD) not known). On the TLC plate, the spots of the chloroform-soluble fraction of the plant did not coincide with those of solutions of GTXs I-IV, XIV, XVIII, grayanoside A and C, the aglycone of grayanoside C and kalmitoxin I.

Analysis by GC-MS of extracts of *R. ferrugineum* and *R. hirsutum* (which also flourishes in the Alps) did not show the presence of GTX I. The presence of grayananes cannot be completely excluded in this case, considering the low sensitivity of the method (LOD of 30 mg/kg) (Lechtenberg et al., 2014).

Lucatello et al. (2022) recently reported the presence of low levels of GTX I in Italian alpine *Rhododendron* honey. The presence of GTX I in these honeys was attributed to *R. ferrugineum*, which is an abundant species in the areas of collection.

Other *Rhododendron* species found in Europe are $R. \times$ *intermedium* which is a hybrid of R. *ferrugineum* and R. *hirsutum* and flourishes in the Alps (Bruni et al., 2016; Sosnovsky et al., 2017), and R. *myrtifolium* in the Carpathians (Sosnovsky et al., 2017). No information on the presence of grayananes in these species is available.

1.2.2.2. Grayananes in honey

For the purpose of the present assessment, the term 'certain honey' shall refer to honey and honey products originated from Ericaceae family plants and containing grayananes. The terms 'mad honey' or 'toxic honey' shall refer to honey-containing grayananes in sufficient amounts to cause intoxication in humans (see Section 3.1.3). These definitions will be used throughout this opinion.

The Panel is aware that the terms 'mad honey' or 'toxic honey' are not exclusive to the honey rich in grayananes but have been previously associated with honeys containing hazardous substances other than grayananes (Zhang et al., 2016). As these other honeys are not derived from Ericaceae species, these are out of the scope of the present assessment and will not be discussed further.

The most commonly studied GTXs in honey are GTXs I, II and III. In a study by These et al. (2015) and more recently by Mulder and de Vries (2022 under Documentation as provided to EFSA), several other grayananes were tentatively identified in honey which may also contribute to the overall risk following oral consumption of 'mad honey' (see Table 1 and Section 3.2).

The presence of other substances in 'mad honey' which might exhibit similar effects to grayananes was considered by the Panel, but no evidence supporting the presence of other potential toxins in Ericaceae honey showing toxicity in humans was identified (Machado et al., 2020; Alkan et al., 2020).

Qualitative characteristics of Rhododendron honeys

Data on the qualitative characteristics of *Rhododendron* honeys suggest that these honeys meet the EU standard⁵ requirements for honey (Oddo et al., 2004; Akgün et al., 2021; Alkan et al., 2020).

Although 'mad honey' is reported to have a sharp and bitter taste (Gunduz et al., 2007; Mayda et al., 2018), honey produced from EU *Rhododendron* species has not (Oddo et al., 2004; Italy, 2022 under Documentation as provided to EFSA).

Rhododendron honeys can be monofloral (i.e. containing at least 45% *Rhododendron* pollen; Mayda et al., 2018) or polyfloral. Monofloral honeys such as buckwheat honey and chestnut honey may also contain *Rhododendron* pollen (Panseri et al., 2013, Mayda et al., 2018). Results from analysis of chestnut honey samples submitted by Turkey in the framework of the EFSA *call for stakeholder inputs in relation to the mandate on grayanotoxins in 'certain honeys'* did not show the presence of grayanotoxins in any of the samples analysed (Turkey, 2022 under Documentation as provided to EFSA).

Pollen from *Rhododendron* species has characteristic shape and size and can be distinguished by scanning electron microscopy (SEM). It presents in the form of relatively large tetrads (four individual pollen fused together) of 30–70 μ m (Alkan et al., 2020; Sarwar and Takahashi, 2013). Pollen analysis supports the identification of territorial origin of the honey based on co-occurring 'accompanying species' and the degree of purity of the honey samples (Stakeholder and focal points inputs on grayanotoxins, 2022, under Documentation as provided to EFSA). The amount of pollen transported by bees depends on the size of the flower (which can vary geographically, even within species), the bee body size and foraging behaviour (Stout et al., 2015).

Honeys known to contain grayananes

Most 'mad honey' intoxications reported in the literature relate to consumption of *Rhododendron* honeys from the Turkish Black Sea region or Nepal (see Section 3.1.3). In the French island of La Réunion, honeys associated with grayanane intoxications in humans are produced from foraging of endemic plant species, especially *Agarista salicifolia*.⁶

In 370 BC, the Greek historian Xenophon was the first to report intoxications from 'mad honey' consumed by soldiers in their returning journey through Turkey (401–400 BC) (Furbee and Wermuth, 1997; Gunduz et al., 2008; Dusemund et al., 2012).

In Turkey, honey produced in springtime is generally more toxic and may contain higher concentrations of grayanotoxins than the honey produced in other seasons. The ratio of GTXs in honey depends on the temperature (> 12°C in the night) and the precipitation during the flowering period which may increase the water content in the nectar but decreases its yield. Another important factor is the wind direction: the northern humid air from the Black Sea increases the yield; the southern dry air flow instead, decreases the nectar density (Turkey, 2022 under Documentation as provided to EFSA).

18314732, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [0203/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

⁵ Council Directive 2001/110/EC of 20 December 2001 relating to honey. OJ L 10, 12.1.2002, p. 47–52.

⁶ Préfet de la Réunion. Direction de l'alimentation, de l'agriculture et de la foret, Service de l'alimentation. Arrêté préfectoral n° 2013-1977. Réglementant la vente de miel en gaufres à la Réunion. http://administration.reunion.gouv.fr/spip.php?article2460

Data from the 2022 European Commission's Honey Market Presentation show that in 2021, 173,284 tons of honey of all origins were imported from third countries in the EU. Most imports were from Ukraine (31.1%) and China (27.7%), followed by Mexico (8.9%) and Argentina (8.3%) (European Commission, 2022).

From 2016 to 2020, only a small amount of honey (7–41 tons) was exported from Nepal worldwide (FAOSTAT data), and only 2.7% (4,676 tons) of the 2021 total honey imports in the EU was from Turkey (European Commission, 2022). Although information on the ratio of *Rhododendron* honey to the total imported honey from all sources is not available to EFSA, the total imports of honey from Turkey suggest that the amount of *Rhododendron* honey reaching the EU market should be low. This should also be the case for honey imported from Nepal.

The absence of intoxication cases from *Rhododendron* honeys of EU origin suggests that *R. ponticum* is not among the species intentionally used in apiculture in Europe. This is also the case for Portugal and Spain, despite Spain being the country with the most hives in Europe (European Commission, 2022), and despite grayanane-containing *R. ponticum* flourishing in the Iberian mountains (Egan et al., 2016).

EFSA is not aware if grayananes are toxic to honeybees foraging native *R. ponticum* species in Portugal and Spain but is aware that honeybees do not forage on the introduced species of *R. ponticum* in Ireland, because of the toxicity of the nectar. Therefore, it is unlikely that the Irish honeybees are exposed to the nectar toxins (Stout et al., 2006; Tiedeken et al., 2016) which may explain why honey from north-western Europe has not raised concerns related to GTX occurrence.

Finally, in a recent search for products labelled as 'mad honey' in the Mintel Global New Products database (GNPD),⁷ EFSA did not find the marketing of such products in the EU. 'Mad honey' was found to be marketed, though, outside the EU and on the internet.

Rhododendron and other Ericaceae honeys with low or not known amounts of grayananes

Honey from *R. ferrugineum* which flourishes in the Pyrenees and the Alps is a domestic *Rhododendron* honey, produced and marketed in the EU. Italian beekeepers bring bees above 1,600 metres in the period between mid-June and the first days of July when the *Rhododendrons* blossom. EU *Rhododendron* honey is a seasonal product with a high degree of purity which presents a delicate and slightly floral/fruity fragrance (Stakeholder and focal points inputs on grayanotoxins, 2022, under Documentation as provided to EFSA).

Based on the available information, it was long thought that honey from *R. ferrugineum* would not contain grayananes (see Section 1.2.2.1; Lechtenberg et al., 2014; Seephonkai et al., 2021). However, a recent study by Lucatello et al. (2022) quantified GTX I and semi-quantified GTX III in Italian alpine *Rhododendron* honey. Of the 125 samples analysed by LC-MS/MS, 30% contained GTX I in a concentration ranging from 0.012 to 0.104 mg/kg (LOQ: 0.01 mg/kg). Some regional variation in the percentage of positive samples was observed: honey originating from Valle d'Aosta and Piemonte contained GTX I above the LOQ in 60% and 49% of the samples, respectively, while for honey from Lombardi and Trentino Alto Adige, this was the case for 14% and 21%, respectively. Most of the samples containing GTX I in levels exceeding 0.04 mg/kg were from the Piemonte and Valle d'Aosta regions. In these areas, *R. ferrugineum* was the most abundant *Rhododendron* species.

No intoxications were reported from honey produced from *Calluna vulgaris* (Heather honey) (Hanson, 2016). In Romania, honey produced by *Calluna vulgaris* in the Apuseni Carpathians area at 1,000–1,500 metres altitude is called Live Black Grass Honey. The honey is produced and collected in small quantities during the last summer period where the highest seasonal temperatures are recorded (Stakeholder and focal points inputs on grayanotoxins, 2022; see Documentation as provided to EFSA).

1.2.2.3. Grayananes in dietary sources other than honey

Only a limited number of Ericaceae species are known to produce grayananes (Koca and Koca, 2007) (Annex B). Cranberries (*Vaccinium macrocarpon*), blueberries (*V. myrtillus*) and lingonberries (*V. vitis idaea*), strawberries from *Arbutus unedo* species are popular foodstuffs originating from plants of the Ericaceae family for which there is no indication of the presence of grayananes (Hanson, 2016). In a study on the safety of cranberry leaves to be used as extracts in beverages, no GTX III was detected (LOD of 0.2 mg/kg) in any of the 75 samples collected at different locations and time points in the USA (Booth et al., 2012).

⁷ https://www.gnpd.com/sinatra/gnpd/search/#

There is some evidence on the presence of GTXs in dietary supplements and processed foods (wine) from third countries (Republic of Korea; see Section 3.2) (Hwang et al., 2018), but no information is available regarding GTXs occurrence in similar foods in the EU.

Herbal teas originating from different *Rhododendron* species to which the EU consumers might be exposed may also contain grayananes. Many *Rhododendron* species have been used in alternative medicine, and other ethnopharmacological applications e.g. traditional Chinese or Ayurvedic medicine (Popescu and Kopp, 2013; Dampc and Luczkiewicz, 2015; Shi et al., 2021). See also Annex B. The investigation of the biological and potential medicinal properties of *Rhododendron species*, in particular their diterpenoid constituents, still constitutes a very active field of research (Li et al., 2013, 2019; Wang et al., 2014). Although the presence of such products in the EU market is unknown, EU consumers may still be exposed by purchase of such products online.

1.2.3. Sampling and methods of analysis

For reliable quantification of grayananes, reference standards of high purity are required. Only for GTXs I and III, a limited number of suppliers could be identified. Recently, a new reference standard became available for rhodojaponin II. For all other GTXs and related compounds (Appendix A), no reference standards are currently available.

1.2.3.1. Extraction and clean-up

Grayananes are often isolated from plants by methanol or chloroform extraction followed by further purification and crystallisation (Koca and Koca, 2007). GTXs can be extracted from honey using aqueous solutions, often in combination with methanol as organic modifier to improve extraction (Lee et al., 2008; Kaplan et al., 2014; Kurtoglu et al., 2014). Optionally an organic or inorganic acid is added (These et al., 2015; Aygun et al., 2018) and the sample may be heated to facilitate dissolution of the honey matrix (Sibel et al., 2014; Silici et al., 2016). Solid-phase extraction (SPE) on C18 or polymeric phase cartridges has been used to clean the sample (Lee et al., 2008; Kurtoglu et al., 2014; Sibel et al., 2015; Silici et al., 2016). Determination of GTXs in honey using a dilute-and-shoot approach has been described (Kaplan et al., 2014). Extraction of GTXs from biological matrices such as blood and urine require purification and concentration by SPE (Cho et al., 2014; Aygun et al., 2018). Hwang et al. (2018) extracted dietary supplements and homemade wine with methanol and used SPE for sample clean-up (Hwang et al., 2018).

1.2.3.2. Detection/quantification methods

Liquid chromatography coupled with tandem mass spectrometry (LC-MS/MS) is most commonly used for the analysis of GTXs in honey and in other biological matrices (Holstege et al., 2001; Lee et al., 2008; Kaplan et al., 2014; Kurtoglu et al., 2014; Sibel et al., 2014; Silici et al., 2016; Aygun et al., 2018; Hwang et al., 2018). Typically, the GTXs are analysed in positive electrospray ionisation mode (ESI), but analysis in negative ESI has also been reported (Zou et al., 2014; Sahin et al., 2015; Silici et al., 2016). Fragmentation spectra of GTXs generally exhibit several losses of water and sodium adducts are often reported (Holstege et al., 2001; Cho et al., 2014; These et al., 2015; Hwang et al., 2018). Quantitative LC-MS/MS analysis of GTXs and related compounds is hampered by the lack of available analytical standards. Most literature reports therefore only include GTXs I and III. Alternatively, LC-HRMS has been used (Zou et al., 2014; These et al., 2015; Hwang et al., 2018). Using LC-HRMS, it could be shown that a sample of 'mad honey' contained up to 15 grayanoid compounds that were tentatively identified and quantified (These et al., 2015). Kurtoglu et al. (2014) reported limits of detection (LOD) in positive ESI for GTXs I and III of 0.2 and 0.1 mg/kg, respectively, in honey. Limits of quantification (LOQ) reported were 0.6 and 0.3 mg/kg, respectively. Kaplan et al. (2014) reported a decision limit (CC α) of 0.012 and 0.010 mg/kg for GTXs I and III, respectively, in honey. An LOD of 0.013 mg/kg and an LOQ of 0.039 mg/kg in honey for grayanotoxin III in negative ESI has been reported (Sahin et al., 2015). For GTXs I and III, based on the lowest validated level, an LOQ of 1 μ g/L in rat whole blood could be established (Cho et al., 2014).

1.2.4. Previous risk assessments

In 2010, as a consequence of a case of poisoning reported to the Hessian ministry of consumer protection, the German Federal Institute for Risk Assessment (BfR) evaluated the health risks of honeys which contain GTXs (BfR 2010a and 2010b). Four hours after consumption of an unknown amount of a honey originating from the Turkish Black Sea region, a consumer had developed

disturbance of consciousness and bradycardia requiring medical treatment. Symptoms faded within 24 hours. An analysis performed by the Hessian food control authority revealed that the honey in question contained GTX III in a concentration of 43 mg/kg and pollen originating mainly from Castanea sativa and R. ponticum. The possible presence of GTX I had not been investigated. BfR only associated the part of the honey derived from R. ponticum, and not that derived from C. sativa, with the poisoning. Based on a review of the relevant literature on the toxicity in humans and experimental animals, the BfR concluded that a No-observed-adverse-effect level (NOAEL) for GTX III could not be identified. Also, the limit values for GTX I and III in honey, below which they would not be associated with health concerns, could not be derived due to missing information on dose-effect relationships of these substances and their interactions between each other and with other plant components in honeys. According to the evaluation of human data on intoxication due to honeys containing GTXs, intake amounts that have led to poisoning vary considerably and were as low as 5 g of these honeys per person. In recent years, intoxications with GTXs-containing honeys in German-speaking countries were exclusively associated with consumption of Turkish honeys. Based on the available knowledge at that time, the BfR drew the conclusion that *Rhododendron* honey from regions of the Turkish Black Sea coast cannot be regarded as safe. Furthermore, the BfR noted in view of the global occurrence of Rhododendron species and other species of the Ericaceae family which contain GTXs, that poisoning would only be expected through honeys collected where such plants dominate the vegetation and not where they are only cultivated as ornamental plants.

In 2013 and 2016, the European Commission submitted to EFSA requests related to an application on the approval of *Rhododendron* honey as 'basic substance' for plant protection purposes in line with Art. 23 of Regulation (EU) No 1107/2009. The applicant's intention was to use honey from *Rhododendron* containing GTXs as a rodenticide. Concentration ranges from 10 to 60 mg/kg for GTX I and 10–300 mg/kg for GTX III in *Rhododendron* honey were proposed by the applicant (EFSA, 2017).

Following consultation with member states, EFSA identified a number of limitations such as the variability of occurrence levels of GTXs and/or the lack of specifications for the product to be marketed which does not allow a consistent composition and/or identify impurities of potential concern to be established. At the time of the evaluation, no information on the marketing of GTXs-containing *Rhododendron* honey in the EU was available neither regarding the composition of such products nor provisions for maximum levels of GTXs in honey from *Rhododendron* (EFSA, 2017). Other limitations relate to the lack of data to support that the available methods of analysis are valid and fit for purpose, and the lack of a proper scientific report to demonstrate, among others, that after being exposed to *Rhododendron* honey in the field, mice die without undergoing unnecessary suffering.

With regard to the effects in humans, EFSA stressed that honey originating from *Rhododendron* spp. containing GTXs cannot be considered compliant with the EU food law since food poisoning was associated with the presence of GTXs in *Rhododendron* honey (also called 'mad honey') (EFSA, 2017). Additionally, note was taken that plant parts of such *Rhododendron* spp. contain natural substances of possible concern for human health as reported in the EFSA Compendium of botanicals (EFSA, 2012).

The Panel noted that in addition to *Rhododendron* spp., plant parts of *Andromeda* spp., *Kalmia latifolia*, *Lyonia* spp., *Pieris formosa* and *Pieris japonica* also contain GTXs and are reported in the EFSA Compendium of botanicals.⁸

1.2.5. Legislation

No maximum levels for GTXs in honey from Ericaceae are set in the EU and no specific regulatory provisions are applicable for GTXs and other grayanane analogues in 'certain honey'. Among Member States, France regulates comb honey ('miel en gaufre') from *Agarista salicifolia* and other endemic plants containing GTXs at regional level (La Réunion) and for the flowering season only (from 1 November to 31 March).⁹ In 2010, Germany (BfR) issued a recommendation discouraging consumption of *Rhododendron* honey originating from the Turkish Black Sea coast. This provision, however, is not implemented in a regulation.

18314732, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [0203/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

⁸ https://www.efsa.europa.eu/en/data-report/compendium-botanicals

⁹ Préfet de la Réunion. Direction de l'alimentation, de l'agriculture et de la foret, Service de l'alimentation. Arrêté préfectoral n° 2013-1977. Réglementant la vente de miel en gaufres à la Réunion. http://administration.reunion.gouv.fr/spip.php?article2460

2. Data and Methodologies

The EFSA risk assessment on grayananes in 'certain honey' was developed applying a structured methodological approach, which implied developing a priori the protocol or strategy of the full risk assessment and performing each step of the risk assessment in line with the strategy and documenting the process. The protocol in Annex A to this Opinion contains the method that was proposed for all the steps of the risk assessment process, including any subsequent refinements/ changes made.

Information on chemistry, formation in plants, natural sources, analytical methods, a previous BfR assessment, an EFSA assessment on the use of 'mad honey' as plant protection product and legislation was gathered from the literature, assessments by EFSA and other bodies (by checking their original websites of the relevant organisations), from current EU and Member State legislation and a *call for stakeholder and focal points inputs in relation to the mandate on grayanotoxins in 'certain honeys'* launched by EFSA in January 2022 (see Documentation as provided to EFSA). Literature searches were conducted to identify new information in reviews and peer-reviewed publications. Details of the literature searches are given in the external scientific report (DTU, 2020) and the protocol (Annex A).

2.1. Hazard identification and characterisation

Information relevant for the sections under hazard identification and characterisation was identified in the framework of a literature search performed on grayanotoxins (DTU, 2020) and additional literature searches conducted during the risk assessment in order to ensure that the most recent information on grayananes is considered in this Opinion (see also Protocol in Annex A).

The selection of the scientific papers for inclusion or exclusion was based on consideration of the extent to which a study was relevant to the assessment or on general study quality considerations (e.g. sufficient details on the methodology, performance and outcome of the study, on dosing, substance studied and route of administration and on statistical description of the results), irrespective of the results. Limitations in the information used are documented in this Scientific Opinion and considered in the uncertainty analysis.

Benchmark dose (BMD) analysis was carried out using the PROAST software package developed by RIVM (version 70.0) on the web-app 'EFSA-Proastplatform' based on the EFSA Scientific Committee guidance (2017a). To judge the width of the BMD credible interval, the set of criteria and cut-off values described in the latest EFSA Guidance on BMD modelling (EFSA Scientific Committee, 2022) were applied.¹⁰

2.2. Occurrence data

EFSA gathers data on the occurrence within the call for continuous collection of chemical contaminants occurrence data in food and feed that is issued every year. European national authorities and similar bodies, research institutions, academia, food business operators and other stakeholders were invited to submit analytical data on grayananes.

No data on grayananes occurrence in honey, honey products or other foods were submitted to EFSA following the continuous call for data on chemical contaminants, including grayanotoxins.

A literature review was carried out by an external contractor in order to collect published data (DTU, 2020) and results are presented in Section 3.2.1. Collection and appraisal of occurrence data were carried out as described in the EFSA Protocol (Annex A).

Quantitative results on the occurrence of GTX I, II, and III in 'certain honey' are available in the literature. Some authors analysed honey for one grayanane only (GTX I or GTX III), others for GTX I and III (see Section 3.2.1, Tables 14 and 15). The Panel noted that only eight samples from Nepal were analysed for GTX I, II and III. There are no studies that provide quantitative concentrations of other co-occurring grayananes in the honey samples. Presently, only semi-quantitative information from a limited number of studies is available for grayananes other than GTX I, II and III in 'certain honey' (Kerkvliet, 1981; These et al., 2015; Mulder and de Vries, 2022 under Documentation as provided to EFSA). These other grayananes may contribute to the overall toxicity of these particular

¹⁰ The latest EFSA Guidance on BMD modelling published in 2022 that introduced Bayesian model averaging was published on 25 October 2022. However, its implementation in the EFSA 'R4EU BMD analysis platform' was not available before 16 November 2022 (https://zenodo.org/record/7334435#.Y5osYXbMLD4). Due to the timelines for discussion and adoption of the present Opinion, it was not possible to perform the BMD modelling using this newer version.

honeys. In the absence of analytical standards for all grayananes (except for GTX I and III), the available information cannot be used for exposure assessment purposes. In addition, Nepalese honeys appear to have a different grayanane composition than honeys from the Turkish Black Sea Region which does not support the merging of analytical results for these two different types of honey.

2.3. Food consumption data

The EFSA Comprehensive European Food Consumption Database (Comprehensive Database) provides a compilation of existing national information on food consumption at individual level. It was first built in 2010 (EFSA, 2011; Huybrechts et al., 2011; Merten et al., 2011). Details on how the Comprehensive Database is used are published in the Guidance of EFSA (EFSA, 2011). The latest version of the Comprehensive Database updated in July 2021 was used.¹¹ Within the dietary studies, subjects are classified in different age classes as follows:

- Infants: > 16 weeks of age to < 12 months old
- Toddlers: \geq 12 months to < 36 months old
- Other children: \geq 36 months to < 10 years old
- Adolescents: \geq 10 years to < 18 years old
- Adults: \geq 18 years to < 65 years old
- Elderly: ≥ 65 years to < 75 years old
- Very elderly: \geq 75 years old

Additional surveys provided information on specific population groups: 'Pregnant women' (from Austria, Cyprus, Spain, Latvia, Portugal and Romania), 'Lactating women' (from Estonia and Greece) and Vegetarians (from Romania).

For the acute assessment, food consumption data were available for 51 surveys from 24 countries. The mean and P95 acute consumption (expressed in g per day and g/kg bw per day) were calculated from the distributions of the total daily consumption of each subject in each dietary survey and age class.

For the chronic assessment, food consumption data were available for 47 surveys from 22 countries surveys based on at least two survey days. The mean and P95 chronic consumption (expressed in g per day and g/kg bw per day) were calculated from the distributions of the total daily consumption of each subject in each dietary survey and age class.

In Annex C.1, these dietary surveys and the number of subjects available for the acute exposure assessment are described.

The food consumption data gathered by EFSA in the Comprehensive Database are the most complete and detailed data currently available in the EU. Consumption data were collected using single or repeated 24- or 48-h dietary recalls or dietary records covering from 3 to 7 days per subject. Because of the differences in the methods used for data collection, direct country-to-country comparisons can be misleading.

It was noted that honey from *Rhododendron* spp. is not specifically addressed in the EFSA food consumption database where most of the eating events relate to the general term 'honey'. Since consumption of *Rhododendron* honey is not reported, a proxy of the consumption of *Rhododendron* honey was used. The Panel decided to use consumption data for honey by merging all consumption events related to 'Honey', 'Honey, monofloral', 'Honey, polyfloral', 'Honey, blended' and 'Minor honey types' (as reported in FoodEx2).

Acute consumption data of 'normal honey' were used for risk characterisation based on a margin of exposure (MOE) and backward calculations under scenario A 'consumption of *Rhododendron* honey as 'normal' honey' (see Section 2.6, 3.4.1 and 3.5.1).

Additionally, a consumption scenario was based on recommended doses for consumption taken from 'mad honey' product labels. A search was made to investigate availability of 'mad honeys' accessible to the EU consumers via the internet market. Examples of recommended doses for consumption were taken from five product labels retrieved from the market (all produced outside the EU) and used in the risk assessment (see Section 3.5.2). The Panel is aware that these portions, as provided on the product labels, are specific to these products and are not necessarily representative of all commercially available 'mad honeys'. In addition, it is uncertain if all consumers follow the

¹¹ Available online: https://www.efsa.europa.eu/en/data-report/food-consumption-data#the-efsa-comprehensive-european-foodconsumption-database

recommendations on the labelling. On the labels, portions were reported in teaspoons or tablespoons; therefore, standard conversion factors of 8 g for 1 full teaspoon of honey and of 22 g for a full tablespoon of honey (LARN, 2014)¹² were applied. Portions in grams were divided by 70 kg as a standard adult body weight. These consumption data were used in backward calculations for risk characterisation under scenario B '*Rhododendron* honey as traditional folk medicine in adults ('mad honey')' (see Sections 2.6 and 3.5.2).

2.4. Food classification

Consumption data present in the EFSA comprehensive consumption database were classified according to the 'Exposure hierarchy' of the FoodEx2 classification system (EFSA, 2015). This is based on a food list of 2,673 entries, referred as 'basic FoodEx2 code', aggregated into food groups and broader food categories in a hierarchical parent–child relationship (up to seven levels). In addition, a catalogue of 28 'facets' is available in order to describe further characteristics of the foods, such as physical state (e.g. powder, liquid) or processing technology (e.g. grinding, milling, crushing).

2.5. Exposure assessment

Because of the lack of specific occurrence data on grayananes and because of lack of consumption data for *Rhododendron* honey and/or other honeys containing grayananes in the EFSA databases and the scarcity of data in the literature, dietary exposure was estimated based on selected concentrations for GTX I and III reflecting concentrations measured in 'certain honey' (see Section 3.4.1).

2.6. Risk characterisation

The general principles of the risk characterisation for chemicals in food as described by the WHO/ IPCS (2009) were applied as well as the different EFSA guidance documents relevant to this step of the risk assessment (see Annex A on the Protocol) and the EFSA guidance on uncertainty analysis in scientific assessments (EFSA Scientific Committee, 2018a).

The acute consumption P95 estimates and selected concentrations for the sum of GTX I and III reflecting the concentrations measured in 'certain honey' were used to derive MOEs based on a reference point (RP) for acute effects derived from BMD analysis. In addition, backward calculations were performed in order to estimate the *highest concentration* of grayananes in 'certain honey' that is unlikely to lead to acute health concerns within the two specific consumption scenarios A and B (see Sections 2.3 and 3.4). When reference is made to these concentrations, the terms *calculated highest concentration(s)* or *highest concentration(s)* will be used in *italics* throughout this opinion in order to distinguish them from the measured highest concentrations in Section 3.2.1.

A comprehensive risk assessment for chronic exposure could not be performed due to the lack of data (see Section 3.4).

3. Assessment

3.1. Hazard identification and characterisation

3.1.1. Toxicokinetics

Toxicokinetic data are scarce and, therefore, will be considered as a whole and not subdivided into Absorption, Distribution, Metabolism and Elimination (ADME). Four animal and three human studies were identified. The studies are reported per substance, starting with the most potent compound.

In a study using ³H-rhodojaponin III (rhomotoxin) injected intravenously (*i.v.*) into mice the highest radioactivity was found in the gall bladder and bile followed by the liver and kidney. The amount of radioactivity in the gall bladder and bile was four times that found in liver and five times that in the kidney. 52.4% of the dose was excreted by 6 h via the urine and faeces. Approximately 60% was found bound to plasma proteins at 30 min (Liu et al., 1986).

Dong et al. (2014) determined rhodojaponins I, II and III¹³ in the plasma of male Sprague-Dawley rats, following oral administration of 21 and 113 mg/kg bw extract of *Rhododendri mollis* flos

¹² https://sinu.it/wp-content/uploads/2019/07/20141111_LARN_Porzioni.pdf

¹³ Rhodojaponins I, II and III used as standards in the method of analysis were isolated from extract of *Rhododendri mollis* flos and identified by high-resolution mass spectroscopy and NMR. Their purity determined by LC was higher than 98%.

containing 0.37% rhodojaponin I, 0.53% rhodojaponin II and 0.11% rhodojaponin III, respectively. Blood samples were collected and analysed at 0, 10, 20, 30, 60, 90, 120, 240, 360, 480 and 720 min of exposure. It was noted that a solution of carboxy methyl cellulose was added as a vehicle which might impact the bioavailability of the substances. The kinetic data of rhodojaponin I, II and III in blood are reported in Table 3.

Table 3:Kinetic parameters for rhodojaponin I, II and III in blood of male SD rats (n = 12) (Dong
et al., 2014)

Parameter	Rhodojaponin I		Rhodojaponin II		Rhodojaponin III	
Dose (mg/kg bw) ^(a)	0.08	0.42	0.11	0.60	0.02	0.12
C _{max} (ng/mL)	50 ± 10	340 ± 50	50 ± 10	180 ± 50	40 ± 10	190 ± 30
T _{max} (min)	$\textbf{45.0} \pm \textbf{21.2}$	$\textbf{54.0} \pm \textbf{48.3}$	$\textbf{60.0} \pm \textbf{32.0}$	$\textbf{72.0} \pm \textbf{34.2}$	$\textbf{52.5} \pm \textbf{21.2}$	$\textbf{50.0} \pm \textbf{36.7}$
T _{1/2} (min)	$\textbf{63.7} \pm \textbf{18.2}$	147 ± 37.8	$\textbf{134} \pm \textbf{66.1}$	216 ± 164	$\textbf{83.7} \pm \textbf{39.6}$	$\textbf{220} \pm \textbf{91.1}$
AUC (ng min/mL)	$\textbf{4,350} \pm \textbf{480}$	56,700 ± 10,330	6,090 ± 2,680	47,030 ± 3,410	3,150 ± 860	$\textbf{46,040} \pm \textbf{8,210}$

(a): Dose calculated based on the data provided.

Assuming that $AUC_{12h} \approx AUC_{\infty}$ (Mei et al., 2006)¹⁴ and based on the results from Table 3, the Panel estimated the ingested doses for rhodojaponins I, II and III. The ratios of the AUC/dose per substance indicate that rhodojaponin III is approximately three times more bioavailable than rhodojaponin I and three to five times more bioavailable than rhodojaponin II. For rhodojaponin III, the % bioavailability based on AUC was estimated between 3 and 7%.

In addition, the Panel also compared the bioavailability of rhodojaponins I, II and III at the time window associated with the maximum concentration of the substances in blood, considering that the risk from acute effects of grayananes is of relevance to the present risk assessment. By comparing the C_{max} /dose, the Panel noted that rhodojaponin III, with an estimated bioavailability between 10% and 13% based on Cmax, is approximately two to three times more bioavailable than rhodojaponin I and four to five times more bioavailable than rhodojaponin II.

Cho et al. (2014) measured kinetic parameters for GTX I and GTX III¹⁵ in rat whole blood using LC-MS/MS with an LOQ of 1 ng/mL. Male SD rats were exposed to 'mad honey' at 11.5 mg/kg bw (corresponding to 0.30 mg/kg bw GTX I and 0.35 mg/kg bw GTX III), or to pure GTX III at 0.35 mg/kg bw, by gavage. Blood samples (approximately 0.5 mL) were taken at 0, 5, 10, 15, 20, 30, 45, 60, 90, 120, 240 and 360 min after dosing. The dose and kinetic data are presented in Table 4. The Panel noted that this implies GTX I and III levels in honey of 26 and 30 g/kg, respectively, which is orders of magnitude higher than normally observed. It is unclear whether the rats received more honey or whether the actual dose of GTX I and III was much lower.

. ,			
Parameter	GTX I (honey)	GTX III (honey)	GTX III (reference standard)
Dose (mg/kg bw)	$\textbf{0.30}\pm\textbf{0.00}$	0.35 ± 0.00	0.35 ± 0.00
C _{max} (ng/mL)	$\textbf{8.7} \pm \textbf{1.3}$	10.3 ± 1.4	$\textbf{23.0} \pm \textbf{1.0}$
T _{max} (min)	$\textbf{20.0} \pm \textbf{14.6}$	$\textbf{20.0} \pm \textbf{15.3}$	$\textbf{45.0} \pm \textbf{24.6}$
T _{1/2} (min)	$\textbf{256} \pm \textbf{50.2}$	$\textbf{580} \pm \textbf{18.2}$	351 ± 15.2
Area Under Curve (AUC _{6h}) (ng min/mL)	2,842 ± 46.8	6,836 ± 729	12,872 ± 1,372
Mean Residence Time (min)	$\textbf{367} \pm \textbf{56.6}$	$\textbf{701} \pm \textbf{152}$	511 ± 171
Cl (mL/min)	$\textbf{0.106} \pm \textbf{0.001}$	0.051 ± 0.030	$\textbf{0.027} \pm \textbf{0.017}$

 Table 4:
 Kinetic parameters for GTX I and III in whole blood of male SD rats (n=6) (Cho et al., 2014)

License

 $^{^{14}}$ The authors suggest that when C_{12h}/C_{max} < 80%, the AUC_{12h} can be considered representative of the AUC_{\infty} (Mei et al., 2006).

¹⁵ GTX III was administered in the form of a hemi (ethyl acetate) complex with formula $C_{20}H_{34}O_6 + \frac{1}{2}C_4H_8O_2$ and molecular weight 414.5. The molecular weight of GTX III is 370.5.

Due to the limitations identified in the study by Cho et al. (2014), no conclusions can be drawn on the bioavailability of GTX I and III following exposure to 'certain honey'.

Accidental poisoning of two mini pigs, kept as pets, with *Pieris japonica* plants resulted in quantifiable levels of GTX I in multiple tissues and body fluids. The highest content was observed in the gastric contents (26.8 ng/g) followed by the bile (19.9 ng/mL) and then the liver (16.8 ng/g) (Pischon et al., 2018).

Aygun et al. (2015, 2018) undertook two separate observational studies to determine the association between GTX levels in blood and urine and the clinical data. In the more recent study (Aygun et al., 2018), concentrations of GTX I and III were measured in 25 patients, presented to the Emergency Medicine Department with symptoms of intoxication (see Table 5 and Section 3.1.3). Due to few cases reported in the study by Aygun et al. (2015), the data reported for blood are not considered for toxicokinetics. Aygun et al. (2018) reported honey intakes between 3 and 30 g. Assuming an average honey intake of 10 g, the Panel estimated that the intake of GTX I and III would be, respectively, 87 and 276 μ g for a person weighing 70 kg, or 1 and 4 μ g/kg bw based on the mean concentrations of GTX I (8.73 μ g/g) and GTX III (27.6 μ g/g) in the honeys consumed. When assuming a urine volume of 0.25 L, the average amount of GTX I and III excreted in urine would be, respectively, 7.5 and 97.5 μ g, i.e. 9 and 35% of the ingested dose.

Table 5: Mean content of GTX I and III in blood and urine of humans after consumption of 'mad honey' by 25 patients (Aygun et al., 2018)

-	Blood	Urine	Honey
Toxin	(ng/mL)	(μg/mL)	(mg/kg)
GTX I	$\textbf{4.8} \pm \textbf{5.6}$	$\textbf{0.03} \pm \textbf{0.04}$	$\textbf{8.7} \pm \textbf{6.0}$
GTX III	$\textbf{6.6} \pm \textbf{8.4}$	0.39 ± 0.46	$\textbf{27.6} \pm \textbf{19.0}$

In a small retrospective observational study aimed at determining the relationship between blood toxin level and clinical response, Choi et al. (2017) measured GTX I and III in six patients (four male and two female) who had consumed *Rhododendron* liqueur (also known as Doo-Gyeon Ju). Initial blood and follow-up (20 h) levels were measured. In the liqueur, the content of GTX I and III was 1.44 and 16.9 ng/mL, respectively. It was estimated from the study that the levels of GTX I and III in blood required to cause hypotension were between 2.52 and 4.55 ng/mL and 17.5 and 27.3 ng/mL. While these concentrations in blood are consistent with other reported data for GTX I and III, there are inconsistencies in the concentrations of the GTX I and GTX III in the liqueur and the subsequent calculations.

No information on the biotransformation of grayananes in mammals is available to EFSA. Tian et al. (2015) investigated, and De Sousa et al. (2018) reviewed the metabolism of diterpenes. For the *ent*-kaurane diterpenoid oridonin, 16 phase I metabolites were identified, following oral administration of the substance in rats. Some of these metabolites were conjugated with glucuronic acid, but no metabolites with an altered diterpene backbone structure were identified.

Considering that grayananes and oridonin are polyhydroxylated diterpenes sharing the same biogenetic precursor (*ent*-kaurane), no metabolic cleavage of the grayanane rings is anticipated. This is supported by studies on the stability of grayananes in rat blood which suggest that the *ent*-kaurane structure is stable (Cho et al., 2014).

Summary on toxicokinetics

In summary, there are few studies on the toxicokinetics of grayananes and they are limited. Thus, the Panel could not draw conclusions on the oral bioavailability of grayananes following exposure to 'certain honey'.

In case reports on human intoxications related to consumption of 'mad honey', GTX I and GTX III were detected in human blood and urine, showing that they are absorbed from the GI-tract.

The Panel noted that most toxicodynamic studies on experimental animals use *i.p.* injection (see Section 3.1.2). However, no toxicokinetic study dealing with *i.p.* injection of grayananes in experimental animals was identified and, therefore, it was not possible to compare the grayanane bioavailability after oral and *i.p.* exposure to 'mad honey' or the pure substance(s).

No information on the biotransformation of grayananes in mammals is available to EFSA. Based on data on the *ent*-kaurane diterpenoid oridonin, no metabolic cleavage of the grayanane rings is anticipated.

3.1.2. Toxicity in experimental animals

This section provides information on both the toxicological and pharmacological effects related to the exposure of animals to grayananes alone, 'certain honey', extracts of 'certain honey' or other extracts. Data are organised in different subsections addressing acute toxicity (i.e. LD_{50} and single dose exposure), and repeated dose toxicity. Only information relevant for the risk assessment is reported in detail.

Among the studies focusing on 'certain honeys', their or other extracts, only those providing quantitative results on the occurrence of grayananes in the tested matrix have been considered further.

3.1.2.1. Acute single-dose toxicity studies

Grayanane acute LD₅₀ estimation

Nine acute single-dose studies compared the toxicity of different natural grayananes by the derived LD_{50} values. Results are summarised in Table 6. For each single grayanane analysed, the botanical origin of the substances, route of exposure and species exposed are also reported. Grayananes are grouped in the table according to the route of exposure, starting from oral administration and animal species. In each group, grayananes have been ranked from the most to the least potent.

 LD_{50} values have been derived in mice by oral (Hikino et al., 1979), intraperitoneal (*i.p.*) (Scott et al., 1971; Fukuda et al., 1974; Hikino et al., 1976) (Table 6) or subcutaneous (*s.c.*) injection (Kohanawa, 1956) and in guinea pigs by intravenous (*i.v.*) injection (Hotta et al., 1980) (see Table in Appendix B). EFSA did not identify studies deriving LD_{50} values for grayananes in rats.

The main route of exposure used to derive an LD_{50} in mice was *i.p.* Results obtained from *i.p.* studies by different authors analysing the same substance in mice are generally consistent (Table 6). Taking into account all these studies, asebotoxin III was found to be the most potent grayanane. Among grayananes unequivocally identified in honey (see Table 1), the most potent appears to be GTX III, followed by GTX I and III. The potency of GTX II, however, is one order of magnitude lower than the potencies of GTX I and III.

In two studies in mice, Hikino et al. (1976, 1979) used the same source (*L. grayana*), strain (dd) and vehicle (3% gum arabic solution) to derive LD_{50} s for GTX I and GTX III, based on two different routes of administration: *i.p.* and oral. The oral LD_{50} values appeared 3.8 and 5.8 times higher than the *i.p.* LD_{50} values for GTX I and GTX III, respectively (see Table 6). The Panel noted that use of a similar experimental design in the two studies by Hikino et al. reduces the uncertainty in comparing the oral vs. the *i.p.* LD_{50} values for the two grayananes. Such comparison, however, incorporates multiple uncertainties associated with testing male mice only, the gross endpoint assessed in the studies (mortality), the bioavailability of grayananes in the oral studies and the lack of toxicokinetic studies allowing comparison of oral vs. *i.p.* bioavailability (see Section 3.1.1). Because of these limitations and information from other studies reporting results from both *i.p.* injection and oral administration (Popescu and Kopp, 2013; Sibel et al., 2014), the Panel did not extrapolate from *i.p.* to oral dose but considered the information by Hikino et al. (1976, 1979) in the uncertainty analysis.

Disturbance of cardiac rhythm (ventricular fibrillation) and depression of respiration are considered the main causes of death (Scott et al., 1971; Fukuda et al., 1974). At lower doses, bradycardia was observed rather than ventricular fibrillation. Animals that did not die after 1 h recovered (Hikino et al. 1976). The signs of toxicity following GTX I and GTX III administration were similar and independent of the route of administration (Hikino et al., 1979).

After *i.v.* exposure to grayananes, guinea pigs experienced respiratory failure, hypersecretion obstructing the airways and convulsions (Hotta et al., 1980) (see Table B.1 in Appendix B). These effects are consistent with those observed in mice.

The Panel noted that these experiments have been performed mainly in males. Thus, LD_{50} values might be overestimated assuming that females are slightly more sensitive to the acute toxic effect than males (OECD Guidelines 420, 2022). Finally, the purity of the compounds tested in the studies is often not reported.

'Mad honey' acute LD₅₀ estimation

Only one study provides an indication of an LD_{50} value for 'mad honey' in mice: It relates to *i.p.* injection of 34 g/kg bw of 'mad honey' extract (from British Columbia) in mice (Scott et al., 1971). The grayanane concentrations in honey were estimated to be 3 and 7 mg/kg for GTX II and III, respectively, together with an unknown amount of rhodojaponin III.¹⁶ Using the data reported by Hikino et al. (1976) (Table 6), the Panel calculated the total grayanane content in honey to be approximately 8.4 mg/kg (range: 7.3–9.3 mg/kg). No GTX I was identified in the 'mad honey' extract (Scott et al., 1971). It was noted that the purity of the compounds tested is not reported and the number of animals in the study is small (three animals).

Compound				LD ₅₀	
Trivial name	Origin	Species/strain/sex	Route	(mg/kg bw)	Reference
GTX III	L. grayana	Mice dd, M	Oral	4.9	Hikino 1979 [§]
GTX I	L. grayana	Mice dd, M	Oral	5.1	Hikino 1979 [§]
Asebotoxin III	P. japonica	Mice dd, M	i.p.	0.1 (0.09–0.11)	Hikino 1976 ^{(a),§}
Rhodojaponin III	R. japonicum	Mice, Kunming, M&F	i.p.	0.27 (0.18–0.35)	Li et al. 2015
		Mice dd, M	i.p.	0.4 (0.36–0.43)	Hikino 1976 ^{(a),§}
Asebotoxin I	P. japonica	Mice dd, M	i.p.	0.51 (0.47–0.55)	Hikino 1976 ^{(a),§}
Rhodojaponin V	R. japonicum	Mice dd, M	i.p.	0.75 (0.71–0.83)	Hikino 1976 ^{(a),§}
GTX III	Not provided	Mice ddY, M	i.p.	0.8	Fukuda et al. 1974
	L. grayana	Mice dd, M	i.p.	0.84 (0.77–0.9)	Hikino 1976 ^{(a),§}
	Standard	Mice, M	i.p.	0.908 (0.805–1.03)	Scott 1971 ^(a)
Lyoniol B	Not provided	Mice dd, M	i.p.	0.61 (0.56–0.67)	Hikino 1976 ^{(a),§}
	Not provided	Mice ddY, M	i.p.	0.8	Fukuda et al. 1974
GTX VI	L. grayana	Mice dd, M	i.p.	0.83 (0.75–0.89)	Hikino 1976 ^{(a),§}
GTX I	Standard	Mice, M	i.p.	1.28 (1.11–1.49)	Scott 1971 ^(a)
	L. grayana	Mice dd, M	i.p.	1.31 (1.19–1.44)	Hikino 1976 ^{(a),§}
	Not provided	Mice ddY, M	i.p.	1.5	Fukuda et al. 1974
Rhodojaponin VI	R. japonicum	Mice dd, M	i.p.	1.23 (1.21–1.26)	Hikino 1976 ^{(a),§}
		Mice, Kunming, M&F	i.p.	1.79 (1.32–2.17)	Li et al. 2015
Kalmitoxin I	P. japonica	Mice dd, M	i.p.	2.28 (2.09–2.49)	Hikino 1976 ^{(a),§}
Lyoniol A	L. ovalifolia	Mice dd, M	i.p.	3.01 (2.70–3.25)	Hikino 1976 ^{(a),§}
	Not provided	Mice ddY, M	i.p.	5.8	Fukuda et al. 1974
Pieristoxin G	P. japonica	Mice dd, M	i.p.	6.52 (5.82–7.17)	Hikino 1976 ^{(a),§}
GTX II	Standard	Mice, M	i.p.	No toxicity up to 4	Scott 1971 ^(a)
	Not provided	Mice ddY, M	i.p.	10	Fukuda et al. 1974
	L. grayana	Mice dd, M	i.p.	26.1 (23.7–28.7)	Hikino 1976 ^{(a),§}
GTX V	L. grayana	Mice dd, M	i.p.	> 100	Hikino 1976 ^{(a),§}
3-epi-GTX III	L. grayana	Mice dd, M	i.p.	> 100	Hikino 1976 ^{(a),§}
GTX XVI	L. grayana	Mice dd, M	i.p.	> 100	Hikino 1976 ^{(a),§}
Rhodomollein XIII	L. grayana	Mice dd, M	i.p.	> 100	Hikino 1976 ^{(a),§}
Leucothol B	L. gravana	Mice dd. M	i.p.	> 100	Hikino 1976 ^{(a),§}

Table 6: LD₅₀ values for naturally occurring grayananes of potential concern in mice. Compounds are ranked in the table based on their potency

§: extract, purity not provided.(a): 48 h.

 $^{^{\}rm 16}$ In the paper referred to as desacetyl pieristoxin B.

Other single-dose toxicity studies with individual grayananes

Symptoms after ingestion of honey containing grayananes in humans occur after minutes to hours and are attributable to effects on the cardiovascular and nervous system (see Section 3.1.3). The Panel paid particular attention to animal models addressing endpoints representative of human intoxication and affected shortly after the exposure, i.e. effects on the nervous system (motor activity and coordination, paraesthesia, muscular activity, salivation) and the cardiovascular system (decreased blood pressure and heart rate) (Moran et al., 1954; Cotten et al., 1956; Li et al., 1982; Mao et al., 1982; Ohgaki et al., 1988; Turkmen et al., 2013; Doğanyiğit et al., 2020).

The effects on the nervous and cardiovascular system are summarised in Tables 7 and 8, respectively. Both tables provide the major characteristics of the experimental design, type and characteristics of grayananes analysed, the most effective dose for each time of exposure selected, the ED_{50} value (when available) and the effects observed. Observations were grouped per representative endpoint and results were ranked from the most to the least potent grayanane.

Potential pharmacological effects of grayananes have also been investigated. However, these studies were considered not relevant to the present assessment and will not be considered further (Ohgaki et al., 1987; Gunduz et al., 2012a; Kuru et al., 2014; Li et al., 2015; Niu et al., 2016, 2018; Sun et al., 2019; Zheng et al., 2019).

Neurotoxicity studies (Table 7)

In two studies by Ohgaki et al. (1988), male mice were exposed via *i.p.* to two different doses of GTX III per animal group and per study: 0.1 and 0.25 mg/kg bw. The results showed a significant decrease in locomotor activity (Ohgaki et al., 1987) and coordination (Ohgaki et al., 1988). The effect occurred within 30 min after administration of both doses and remained significant up to 1 h at the highest dose only (Ohgaki et al., 1987). Higher levels of GTX I (0.25 mg/kg bw, i.p.) were required to induce the same reduction in locomotor activity as GTX III (0.1 mg/kg bw). The effects observed were considered significant within a timeframe of up to 1 h following exposure to GTX I (or GTX III). The highest effect was observed within 30 min of exposure. After 1 h of exposure, mice had fully recovered. When GTX II was administered i.p. at 0.25 mg/kg bw, no effect was observed (Ohgaki et al., 1987).

Other effects following i.p. exposure to GTX I, II, III and Iyoniol A and B in rodents are piloerection, increased salivation and retching, all occurring between 5 and 30 min, depending on the dose (Fukuda et al., 1974; Fukumoto et al., 1993). In mice, increased grayanane exposure results in washing, grooming, abnormal gait and salivation (Fukuda et al., 1974). According to Fukuda et al. (1974), ataxia is also observed with increasing doses in mice (doses not reported). Salivation was observed in rats exposed to GTX III (Fukumoto et al., 1993).

The studies reporting effects on the nervous system did not allow the identification of an NOAEL since adverse effects were observed at all tested doses, and part of the data is reported as ED_{50} only (Fukuda et al., 1974; Ohgaki et al., 1987, 1988; Fukumoto et al., 1993).

Experimental design	Grayanane	Dose mg/kg bw	Time	Effects on the nervous system	Reference	
Locomotor activi	Locomotor activity					
Mice, Std-ddy, M <i>i.p.</i> n = 3 <i>per</i> group Vehicle: saline	GTX III ^(b)	0.1; 0.25	Within 30 min	Reduction of locomotor activity (0.1 vs. 0.25 mg/kg bw, ns) and coordination The effect is reverted at 60 min, only for 0.1 mg/ kg bw (see below)	Ohgaki et al. (1987) Ohgaki et al. (1988)	
		0.1	60 min	No effect		
		0.25		Reduction of locomotor activity		

Table 7:	Summary of	of aravana	anes affecting	the nervous	svstem	in animal	studies ^(a)



Experimental design	Grayanane	Dose mg/kg bw	Time	Effects on the nervous system	Reference
Mice, Std-ddy, M <i>i.p</i> .	GTX I ^(b)	0.25	Within 30 min	Reduction of locomotor activity	Ohgaki et al. (1987)
n = 3 <i>per</i> group Vehicle: saline			60 min	No effect	
Mice, Std-ddy, M <i>i.p.</i> n = 3 <i>per</i> group Vehicle: saline	GTX II ^(b)	0.25	Up to 60 min	No effect	Ohgaki et al. (1987)
Cholinergic effect	t				
Mice, ddY <i>i.p</i> . n = 10–15	Lyoniol B	0.11	Within 30 min	Salivation (ED_{50}) and retching	Fukuda et al. (1974)
Mice, ddY <i>i.p.</i> n = 10–15	GTX III	0.15	Within 30 min	Salivation (ED_{50}) and retching	Fukuda et al. (1974)
Mice Std-ddy, M <i>i.p.</i> n = 3 <i>per</i> group Vehicle: saline	GTX III ^(b)	0.1; 0.25	Not reported	Salivation, vomiting, paralysis of the hind paw	Ohgaki et al. (1988)
Wistar rats, M	GTX III ^(c)	0.8	8–15 min	Salivation	Fukumoto et al. (1993)
<i>i.p.</i> n = 4 <i>per</i> group; control n = 3 vehicle: EtOH/ saline		2.8	5–10 min	Salivation plus loss of power, strength and respiratory inhibition after 20 min	
Mice, ddY, M <i>i.p.</i> n = 10-15	GTX I	0.22	Within 30 min	Salivation (ED_{50}) and retching	Fukuda et al. (1974)
Mice, ddY, M <i>i.p.</i> n = 10–15	Lyoniol A	0.7	Within 30 min	Salivation (ED_{50}) and retching	Fukuda et al. (1974)
Mice, ddY, M <i>i.p.</i> n = 10–15	GTX II	10–15	Within 30 min	Salivation (ED $_{50}$) and retching	Fukuda et al. (1974)
Mice ICR, M <i>i.p.</i> n = 3 <i>per</i> group	GTX II ^(d)	10	Immediately after dosing	Piloerection also coupled with sedation and ptosis	Terai et al. (2003)

M: male; vehicle: absent when not reported; ns: not significant.

(a): All studies include control groups.

(b): Standard provided by another researcher. Standard was identified by NMR, MS and IR.

(c): Standard purified from leaves of *L. grayana*.

(d): Standard derived from dehydration of GTX III.

Cardiovascular toxicity studies (Table 8)

The main cardiovascular effects reported after administration of GTX I (Moran et al., 1954; Cotten et al., 1956), GTX III (Turkmen et al., 2013) and rhodojaponin III (Li et al., 1982; Mao et al., 1982) are decreased blood pressure and bradycardia. Most studies have been performed via i.v. injection in dogs; only one study describes i.p. injection in rats (see Table 8). In order to measure blood pressure, heart rate and contractile force, animals were anesthetised to be cannulated. Grayananes tested in dogs were all injected through the femoral vein.

Reduction of blood pressure and heart rate after i.v. injection of rhodojaponin III in dogs was immediate (0.15–5 min), dose dependent and occurred at 0.004 mg/kg bw (Li et al., 1982; Mao et al., 1982). No decrease in blood pressure was observed at 0.002 mg/kg bw rhodojaponin III, while no data were provided for heart rate (Mao et al., 1982). Decreased heart rate and blood pressure were

License

also observed after i.v. injection of 0.005 and 0.008 mg/kg bw GTX I in dogs and i.p. injection of 0.2–0.8 mg/kg bw GTX III in rats (Moran et al., 1954; Cotten et al., 1956; Turkmen et al., 2013).

In a study by Turkmen et al., (2013), four groups of six male Sprague Dawley rats were exposed to a single i.p. dose of GTX III¹⁷ at levels 0 (control group), 0.2, 0.4 and 0.8 mg/kg bw. Blood pressure and heart rate were monitored continuously over 2 h with a 20 min baseline measurement before treatment. Values obtained at intervals of 30 min over a timeframe of 90 min after GTX III administration were taken into account to allow comparison with baseline values and between the different doses tested (Turkmen et al., 2013). The Panel noted that heart rate decreased with time in a dose-dependent manner. In particular, 0.4 and 0.8 mg/kg bw GTX III reached the maximal heart rate decrease after 40 and 20 min of exposure, respectively; this effect is mirrored at 60 and 30 min for each dose. This decrease was most notable in the two highest doses.

Yilmaz et al. (2017) studied the variations in respiratory rate, heart rate and arterial blood pressure in six groups of seven male Wistar albino rats exposed to a single i.v. dose of GTX III at levels 0 (control group), 0.005, 0.010, 0.020, 0.050 and 0.10 mg/kg bw. The three parameters were continuously monitored for 60 min. No effect on any parameter was observed at the lowest dose, i.v. (0.005 mg/kg bw). For all other doses, a dose-depended decrease was observed for the first 15 minutes of exposure. Thereafter, recovery to control values occurred for doses at 0.010 and 0.020 mg/ kg bw whereas the two highest doses were lethal. The Panel noted that the NOAEL in the study is one order of magnitude lower than the lethal dose. Limited statistical information was reported in the study.

Experimental design	Grayanane	Dose mg/kg bw	Time	Cardiovascular system	
Dogs <i>i.v.</i> n = 6 <i>per</i> group	Rhodojaponin III	0.0035; 0.008	Not reported	Reduces blood pressure (BP) and blood pressure elevation due to carotid pressor reflex (antihypertensive effect). The effect is dose dependent: BP: 0.0035 mg/kg bw: -18%* 0.008 mg/kg bw: -41%*	Li et al. (1982)
				BP elevation: 0.0035 mg/kg bw: -8%* 0.008 mg/kg bw: -41%*	
Dogs <i>i.v.</i> n = 3–10 according to experimental group	Rhodojaponin III	0.001; 0.002	0.15–5 min	Not significant decrease blood pressure (BP lowering average rate %: 4.6%; 12.4%, respectively) Heart rate not measured	Mao et al. (1982)
		0.0035; 0.008; 0.02		Decreased blood pressure (BP lowering: average rate %: 60.5%, 64 %, respectively) and heart rate (lowering average rate %: 39%, 70%, respectively). The effect does not significantly increase with the dose (up to 0.02 mg/kg bw)	

Table 8:	Summary of	grayananes	induced	acute	cardiovascular	toxicity	in animal	studies ^(a)
----------	------------	------------	---------	-------	----------------	----------	-----------	------------------------

18314722, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [0203/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons

 $^{^{17}}$ GTX III was administered in the form of a hemi (ethyl acetate) complex with formula $C_{20}H_{34}O_6\cdot 1/2C_4H_8O_2$ and molecular weight 414.5. The molecular weight of GTX III is 370.5.

Experimental design	Grayanane	Dose mg/kg bw	Time	Cardiovascular system	
Dogs <i>i.v.</i> n = not reported	GTX I	0.005	5 min	Decreased blood pressure, contractile force and heart rate	Cotten et al. (1956)
Mongrel dogs <i>i.v.</i> n = 2-6 according to experimental group	GTX I	0; 0.008	Not reported	Decreased blood pressure	Moran et al. (1954)
Sprague Dawley rats, M i.p. n = 4 per group Vehicle: 2% EtOH N.B. controls were injected only with saline	GTX III ^(b)	0; 0.2; 0.4; 0.8	30, 60 and 90 min	Decreased heart rate in a dose-response manner at 60 min of exposure	Turkmen et al. (2013)
Wistar albino rats, M i.v. n = 7 per group	GTX III ^(c)	0; 0.005; 0.010; 0.020; 0.050; 0.100	0, 2, 5,10,15, 20, 25, 30, 40, 50 and 60 min	Decreased heart rate, respiratory rate and blood pressure in a dose- dependent manner No effect observed at 0.005 mg/kg bw Animals died at doses 0.05 mg/kg bw and 0.10 mg/kg bw	Yilmaz et al. (2017)

M: male; i.v.: intravenous; vehicle: absent when not reported.

*: % of inhibition calculated on reported pressure average values (mmHg).

(a): Where separate controls are not reported, cardiovascular parameters were measured before and after treatment in the same animal.

(b): Standard obtained from Sigma-Aldrich.

(c): Standard obtained from Enzo Life Sciences.

Other toxicity studies

Lipid peroxidation, increase in malondialdehyde coupled to a decrease of catalase (CAT), glutathione peroxidase (GSH-Px), superoxide dismutase (SOD) were observed in mouse liver and kidney as well as in lung, heart, spleen, testes, epididymis, brain and erythrocytes, between 1 and 24 h after exposure to GTX III i.p at 0.01 mg/kg bw (Cakmak-Arsalan et al., 2020a,b) and at 0.015 mg/kg bw (Sibel et al., 2014). These observations were not coupled to clinical biochemistry parameters or to histological data.

Serum clinical biochemistry parameters indicative of liver damage are reduced after 1 h exposure to GTX I (0.25, 0.5, 1 mg/kg bw) in rats, with 0.25 mg/kg bw GTX I the most effective dose (Ascioglu et al., 2000). No effect was observed for GTX III up to 0.8 mg/kg bw both in rats and mice (Silici et al., 2016; Fukumoto et al., 1993). A significant increase in serum albumin and aminotransferases occurred after 1 h exposure to 2.8 mg/kg bw GTX III in mice (Fukumoto et al., 1993). Histopathological changes in the liver are described by Ascioglu et al. (2000) after exposure to 1 mg/kg bw GTX I. Neither Silici et al. (2016) or Fukumoto et al. (1993) provided data on liver histology. The lack of indication of incidence and severity grade and the absence of biochemical signs of liver toxicity and histopathological examinations are available from another study (Kukner et al. 2016), but no significant increase of incidence in the liver congestion, steatosis and inflammatory cells was observed after 24 h from a single dose of 0.01 mg/kg bw GTX III i.p in mice. Overall, examination of clinical biochemistry parameters and histological observations after GTX I and GTX III administration provided inconsistent data (Fukumoto et al., 1993; Ascioglu et al., 2000; Silici et al., 2016).

Effects on the reproductive system

Only one i.p. study was identified with GTX III in male Sprague Dawley rats (Doğanyiğit et al., 2019). Following 24 h of a single administration at doses 0, 0.1, 0.2, 0.4 and 0.8 μ g/kg bw, tissue irregularities in testis, appearance of apoptotic cells and decreased tubule diameter were documented but did not show a dose-response. No increase of male sex hormones was observed in the serum. These findings are incomprehensive, and no conclusions can be drawn. The Panel noted, however, that the potential effects of grayananes on the reproductive system require further investigation.

Other single-dose studies with 'mad honey' and other matrices

Details on studies with 'certain' Ericaceae honey and an extract from *Rhododendri Mollis* Flos are reported in Appendix B, Table B.2. Only those studies for which it was possible to estimate the exposure to grayananes are reported.

Single-dose toxicity studies with different doses of honey containing grayananes or extracts of grayanane-containing honey were performed in rats of both sexes (Onat et al., 1991a,b; Kuru et al., 2014; Sibel et al., 2014; Silici et al., 2016; Eraslan et al., 2018), male mice (Cakmak-Arsalan et al., 2020a,b; Kukner et al., 2016). The honey or honey extracts were exclusively derived from *Rhododendron* spp. (Sahin et al., 2018).

Apart from the Chinese drug Nao-Yang-Huaif extract, obtained at the Shanghai market, all other honeys came from the Turkish Black Sea region (Sahin et al., 2018; Onat et al., 1991a,b). Only five out of nine available studies provided quantitative results on the grayanane concentrations in honey, namely GTX I and GTX III. In summary, honey from the Black Sea region has been administered to animals by gavage with grayanane concentrations in honey ranging from 18 to 34 mg/kg for GTX I and from 5.9 to 39.9 mg/kg, for GTX III. The tested honeys have not been analysed for the presence of other grayananes.

Effects on the cardiovascular and respiratory system were observed in albino rats after i.p. injection of *Rhododendron* honey extracts obtained from *R. ponticum* honey which had previously led to human intoxication with severe bradycardia and hypotension. No indication of the grayanane content was provided (Onat et al., 1991a,b). Male and female albino rats exposed to 50, 1,000, and 5,000 mg/kg bw of the *R. ponticum* honey extract, via i.p. injection, displayed a dose-dependent reduction of both heart and respiratory rate. The effects were significantly different from the controls at dose levels of 1,000 and 5,000 mg/kg bw 15 min after exposure to the honey extract. No effects were observed at the lowest dose. The observed bradycardia and respiratory rate depression were restored by a subsequent i.p. injection of atropine (Onat et al., 1991a,b).

Effects in rodent liver from single-dose exposure to 'mad honey' involved altered lipid/protein ratio and lipid peroxidation (Sibel et al., 2014; Eraslan et al., 2018; Cakmak-Arslan et al., 2020a). Kukner et al. (2016) reported significant congestion and steatosis in the absence of necrosis. The data refer to a single sample of 'mad honey'. Oxidative stress has been observed in the kidney, lung, heart, spleen, testes, epididymis and brain (Sibel et al., 2014; Silici et al., 2016; Eraslan et al., 2018; Cakmak-Arslan et al., 2020b). It was noted that most of the studies assessing 'mad honey' also investigated the single grayananes (see subsection on single-dose studies on grayananes). Overall, these studies were considered of limited value due to experimental deficiencies and were, therefore, not useful for risk assessment. In addition, these limited observations in liver and kidney in rodents have not been confirmed in humans and will not be considered further.

Regarding matrices other than honey, Dong et al. (2014) investigated the effects of a mixture of rhodojaponins I, II and III¹⁸ in male Sprague-Dawley rats following oral administration of 21 and 113 mg/kg bw extract of *Rhododendri mollis* flos containing 0.37% rhodojaponin I, 0.53% rhodojaponin II and 0.11% rhodojaponin III, respectively. The Panel estimated the intake for the low-and high-dose groups to be 0.08 and 0.42 mg/kg bw rhodojaponin II. O.11 and 0.60 mg/kg bw rhodojaponin II and 0.02 and 0.12 mg/kg bw rhodojaponin III. Severe, reversible, poisoning symptoms were reported in the high-dose group (vomiting, muscle rigidity, convulsion). These effects were not noted in the low-dose group. Death occurred in 6 out of 20 rats treated with the highest dose. Heart rate, left ventricular systolic and left ventricular diastolic pressure decreased suggesting cardiac dysfunction. Plasma levels of LDH and CK-MB considerably increased in the treated animals whereas only a small increase was observed in AST levels compared to the control group.

Details of these studies are reported in Appendix B, Table B.2.

¹⁸ Rhodojaponins I, II and III used as standards in the method of analysis were isolated from extract of *Rhododendri mollis* flos and identified by high-resolution mass spectroscopy and NMR. Their purity determined by LC was higher than 98%.

Summary of acute animal toxicity studies after exposure to grayananes or preparations containing grayananes

Acute toxicity studies have been performed mainly in male rodents exposed to grayananes via i.p. injection or to 'mad honey'/'mad honey extracts'/other preparations by oral administration. Despite two different routes of exposure to grayananes have been used, similar acute effects were observed. These observations support a role for grayananes in 'mad honey' toxicity.

Among the different effects reported in acute animal toxicity studies, impairment of the nervous system and cardiovascular effects are those considered the most relevant, since occurring shortly after exposure (within 30/60 min) and are consistent with human intoxications due to 'mad honey' intake. In addition to these effects, different blood chemistry parameters indicative of kidney and liver damage were also altered. Nevertheless, these results were often inconsistent, not supported by histological changes and not representative of human intoxication.

Studies reporting effects on the nervous system (reduced locomotor activity and occurrence of cholinergic effects) do not allow the identification of an RP.

Available data allowed the calculation of an RP only for GTX III. The same value applies also for GTX I which appears to have similar toxicity. To derive RPs for the other grayananes relative potency factors (RPFs) should apply (see Section 3.1.5).

3.1.2.2. Repeated dose toxicity studies

Repeated dose studies with individual grayananes

Short-term exposure

Hikino et al. (1979) examined the following clinical parameters in rodents following exposure to GTX I and GTX III by oral gavage for 3 months at dose levels 0, 0.05, 0.25 and 1.0 mg/kg bw per day in mice, and at dose levels 0, 0.05 and 0.25 mg/kg bw per day in rats: total leucocyte and erythrocyte counts, serum glutamic–oxalacetic transaminase (GOT) and glutamic–pyruvic transaminase (GPT), alkaline phosphatase (AIP) and lactate dehydrogenase (LDH). Some of those parameters changed in a dose-dependent manner. Macroscopic examinations in livers, kidneys and spleens at necropsy did not show essential differences in the organ appearance and no notable organ alterations were observed in the histological examinations of the control group and groups of animals exposed to GTX I or III. Decrease in liver and spleen weights was observed following administration of *Rhododendron* extracts and the toxins at 1 mg/kg bw.

Ascioglu et al. (1996) examined the effects of i.p. injection of GTX I (0, 0.075 and 0.15 mg/kg bw per day) on Swiss albino male rats every day for 3 months on hepatic, renal and heart function. Urine analysis, serum analysis and histopathological examination of the liver, kidney and heart were performed. The highest dose of GTX I decreased the mean body weight gain, caused proteinuria and haematuria, increased glutamic pyruvate transaminase (GPT) and LDH4 activities and decreased total serum protein. No histopathological changes were observed.

In a study on male Sprague Dawley rats, the cardiotoxic effects (Doğanyiğit et al., 2020) and testicular effects (Doğanyiğit et al., 2019) of exposure to GTX III were investigated. Rats were exposed to GTX III via i.p. injection every day for 3 weeks. Doses administered were 0.1, 0.2, 0.4 and 0.8 μ g/kg bw per day. Changes in brain natriuretic peptide (BNP), interleukin IL-1 β , IL-6 and tumour necrosis factor- α (TNF- α) and apoptosis in heart tissue were reported. These changes, however, were small, variable and the dose–response relationship was unclear. Histological changes in the testes were evaluated by Johnsen's Testicular Biopsy Score (JTBS).¹⁹ After an initial decrease in the score, the response curve flattened. The diameter of the seminiferous tubules decreased at the two highest doses. Vacuolisation of the testes was seen at 0.1, 0.4 and 0.8 μ g/kg bw per day. No statistically significant changes in the levels of LH, FSH or testosterone were observed in the serum and the testes, respectively. Apoptosis was observed in the testicular cells.

No teratogenic effect was observed after i.p. injection of GTX I (0.15 mg/kg bw per day) in mice for three consecutive days (group 1: 6–8 Gestation Days (GD); group 2: 9–11 GD, group 3: 12–14 GD) and injection into the embryogenic coelom one day after egg incubation in chicks (Kobayashi et al., 1990).

18314732, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [0203/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

¹⁹ https://www.karger.com/Article/Abstract/178170

Repeated dose studies with Rhododendron Honey

Two repeated dose toxicity studies with 'mad honey' were identified.

Following exposure of Sprague-Dawley female rats to 'mad honey' by oral gavage for eight consecutive days at doses of 0, 0.3, 0.6, 1.2 and 2.4 g/kg bw per day (equivalent to 0.012, 0.024, 0.048 and 0.096 mg GTX III /kg bw per day) a dose-response related increase in a number of clinical chemistry parameters such as AST, ALT, LDH, ALP, CK and CK-MB was observed in the serum (Sahin et al., 2016). There were no further examinations to support that these alterations were toxicologically relevant.

In a study on male Wistar rats, Eraslan et al. (2018) found significant variations in the levels of oxidative stress markers in animals exposed to 'mad honey' by gavage at 5 g/kg bw per day for 60 days. The honey contained GTX I and III at 34 and 6.5 mg/kg, respectively. Thus, the Panel estimated the daily exposure to be 0.17 mg/kg bw per day GTX I and 0.03 mg/kg bw per day GTX III. Malondialdehyde and nitric oxide increased in the liver, kidney, brain, testes, heart and plasma. Superoxide dismutase increased in the same organs except for the brain where a decrease was seen. Catalase increased in the heart and erythrocytes whereas it decreased in the liver, kidney, brain and testes. GSH peroxidase increased in the liver, brain and erythrocytes and decreased in the kidney, testes and heart tissues. 4-Hydroxynonenal increased in the serum.

Summary of repeated-dose toxicity studies

The limited information from subacute and subchronic studies in rodents suggests that several biochemical parameters are altered following exposure to individual grayananes and/or 'mad honey', sometimes in a dose-dependent manner. None of these changes, however, were considered as toxicologically relevant because the magnitude of the changes was small. Histological examinations were only performed in some of the studies showing changes in biochemical parameters in the liver, kidney or heart, and no histological changes were observed in any of these organs.

Histological findings in the testes from repeated dose *i.p.* injection of GTX III in rats at doses from 0.1 to 0.8 μ g/kg bw per day for 3 weeks suggest effects on the testes.

Based on the above, no RP for subacute and/or subchronic exposure could be identified from these studies.

3.1.2.3. Genotoxicity studies

Cucer and Eroz (2010) investigated the mutagenic effects of GTX II and GTX III on cultured human lymphocytes. 10 male and 10 female subjects aged between 26 and 45 donated blood which was used for the micronucleus (MN) test and chromosome breakage analysis. GTX II and GTX III were added to the cultures at 0.35 and 3.52 mg/mL for 72 h. An increase in micronucleated cells was seen with age, but no genotoxic effects of GTX II and GTX III were observed in either the MN assay or in the number of cells with chromosomal breaks in comparison with the negative control. The positive control (mitomycin C) did show higher values in both assays.

In a study on male Wistar rats, Eraslan et al. (2018) evaluated the effect of 'mad honey' on oxidative stress, hepatic metabolism and genotoxic damage under acute (12.5 g honey/kg bw single dose, n = 12), subacute (7.5 g honey/kg bw per day for 21 days, n = 12) and subchronic (5 g honey/kg bw per day for 60 days, n = 12) dosing conditions. The concentration of GTX I and GTX III was, respectively, 34 and 6.5 mg/kg in the 'mad honey' used. Thus, the Panel estimated the daily exposure, given by gavage, in the acute study to be 0.43 mg/kg bw GTX I and 0.08 mg/kg bw GTX III, in the subacute study 0.26 mg/kg bw GTX I and 0.05 mg/kg bw GTX III, and in the subchronic study 0.17 mg/kg bw GTX I and 0.03 mg/kg bw GTX III. Multiple markers of oxidative stress were measured in liver, kidney, brain, heart, testes and plasma or erythrocytes (results reported in Section 3.1.2.2). The authors report positive results in the Comet assay expressed as increases in tail intensities. These were observed in the kidney and heart (acute study). There were no changes in the incidence of micronucleated cells with any treatment. The reporting of genotoxicity data was however poor, including major deviations from the recommended reporting of the primary comet assay descriptors. Overall, the evidence for genotoxicity was unclear.

A more recent study by Rasgele et al. (2021) measured the genotoxic effects of *Rhododendron* honey using three *in vivo* mammalian bioassays, e.g. chromosome aberration (CA) and micronucleus (MN) assays in bone marrow cells and induction of sperm abnormalities. Male mice (*M. musculus*) were exposed to honey at 25, 50 and 75 mg/kg bw by gavage for 24 and 48 h. *Rhododendron* honey

18314732, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [0203/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

contained 30 mg/kg GTX I and 7 mg/kg GTX III. Thus, the Panel estimated the exposure to GTX I from honey to be, respectively, 0.75, 1.50 and 2.25 µg/kg bw, and exposure to GTX III was, respectively, 0.18, 0.35 and 0.53 µg/kg bw. In addition, GTX III was also given at 0.01 mg/kg bw by i.p. injection. A significant increase in the number of cells with CAs was observed at all the tested concentrations of Rhododendron honey, while negative results were obtained with GTX III. The mitotic index was not changed in any experimental group. There was a concentration-related increase in the percentage of total sperm abnormalities, but this increase was not statistically significant compared to the control. In the MN assay, the highest dose of *Rhododendron* honey and the isolated GTX III increased the number of micronucleated polychromatic erythrocytes (MNPCE) with a decrease in the ratio of polychromatic/normochromatic erythrocytes (PCE/NCE) at both times indicating a genotoxic effect in the presence of bone marrow toxicity. The Panel noted that the study was clearly positive and there was sufficient evidence of bone marrow toxicity because exposure to GTX III (0.01 mg/kg) and Rhododendron honey (75 mg/kg) decreased the ratio of PCE to NCE for the two exposure periods analysed (EFSA Scientific Committee, 2017b). Although the study showed some deviation from the recommendations in the OECD TG 475 guideline (e.g. GTX III injected *i.p.*, reporting of the results), the data were clearly positive for (i) a dose-related increase in the percentage of bone marrow cells with DNA breaks, i.e. chromatid fusions, following in vivo exposure to RH; (ii) increase of micronuclei at the highest dose of *Rhododendron* honey and at 0.01 mg/kg bw GTX III following both a 24 and 48 h treatment.

Predictive tools were also used to identify structural alerts for genotoxicity and generate predictions on the mutagenic potential of grayananes. For this purpose, structures of three non-epoxidised grayananes (GTX I, GTX II and GTX III) and two epoxidised grayananes (asebotoxin III and pieristoxin G) were selected for assessment. No structural alerts for mutagenicity were identified in ToxTree and QSAR toolbox for the three non-epoxidised grayananes. Regarding GTX I and III, the predictions of the four VEGA models were inconsistent suggesting that GTX I and GTX III are non-mutagenic but with a low reliability (score of 0.53). The result for GTX II is that the substance is unlikely to be mutagenic with an overall consensus score of 0.75. The presence of the epoxide on the molecules of asebotoxin III and pieristoxin G was identified as a structural alert. Whereas the three non-epoxidised grayananes had negative (even though partially inconsistent) predictions, the potentially reactive epoxide moieties were flagged for two substances. The above pattern of QSAR predictions does not allow to derive conclusions on the genotoxicity of grayananes.

In summary, there is positive evidence for the induction of chromosomal damage by grayananes present in *Rhododendron* honey (given by gavage) and GTX III (given *i.p.*) from an *in vivo* MN study. This is confirmed by a chromosomal aberration assay performed by the same authors on the same honey sample (Rasgele et al., 2021). In line with the Scientific opinion on genotoxicity testing strategies applicable to food and feed safety assessment (EFSA, 2011), if the first *in vivo* test is positive, the substance should be considered as an *in vivo* genotoxin. Presently, there is no evidence allowing to conclude that grayananes act via a direct or indirect mode of action.

3.1.3. Observations in humans

The data presented here reflect the existing knowledge from case reports on pharmacological and toxic effects after ingestion of grayananes containing honeys, plants, plant parts or their preparations. Furthermore, studies in volunteers investigating the pharmacological effects of grayananes are addressed. In the following sections, focus is put on reports with relevance for the risk assessment of the intake of grayananes with food, e.g. giving information on dose–effect relationships.

3.1.3.1. Case reports on acute intoxications due to ingestion of 'certain honey'

Information from the literature

Acute intoxications in humans due to consumption of honey containing grayananes have been mainly associated with honeys originating from the Turkish Black Sea region also referred to as 'mad honey', Pontic honey or Turkish wild honey. In this area, toxic *Rhododendron* species (*R. ponticum* and *R. luteum*) containing grayananes are widespread (e.g. Gossinger et al., 1983; Koca and Koca, 2007; Gunduz et al., 2008; Demir and Kahveci, 2012; Jansen et al., 2012). However, acute poisoning cases have also been associated with ingestion of certain Ericaceae honey from other countries, e.g. in North America (Gunduz et al., 2008; Jansen et al., 2012), from Nepal (Jauhari et al., 2009; Shrestha et al., 2009; Jansen et al., 2012; Sohn et al., 2014; Lim et al., 2016) or from Reunion Island (honey collected from *Agarista salicifolia* (synonym: *Agauria salicifolia*)) (Weber et al., 2009). Acute intoxications

reported in the last decades from European countries or South Korea are associated with imported honey from Turkey or Nepal, respectively (e.g. BfR, 2010b; Jansen et al., 2012; Ullah et al., 2018).

Known doses of ingested *Rhododendron* honey causing acute intoxications vary (see also Table 9). In the study of Yilmaz et al. (2006), doses ranged between 5 and 30 g; however, honey intakes as high as 150 g leading to poisoning are also described (e.g. Aygun et al., 2015; Jauhari et al., 2009).

The majority of intoxications refer to males aged between 40 and 65 years (Silici and Atayoglu, 2015; Ullah et al., 2018). The higher prevalence may be explained according to Ullah et al. (2018) by the higher use of 'mad honey' as a sexual stimulant as well as a higher rate of hypertension in males of this age group. Paediatric case reports of *Rhododendron* honey intoxication are scarce. In the case of a 5-month-old male, vomiting and drowsiness occurred 4 h after he was given two teaspoons of 'mad honey' against coughing (Bilir et al., 2016). He recovered after fluid replacement. No further details on origin and composition of the honey ingested are reported. Yavuz et al. (1991) mention the intoxication of a 7-month-old infant without giving any details. Uzun et al. (2013) described that, 2 h after consumption of 150 mL of 'mad honey', a 15-year-old boy experienced hypotension (75/50 mm Hg) and sinus bradycardia (pulse: 45/min) apart from weakness and dizziness. He was treated in an emergency department and discharged one day later.

Resulting symptoms, which occur within minutes up to 5 hours, are mainly associated with the binding of grayananes to the voltage-dependent sodium channels leading ultimately to depolarisation of cell membranes (see Section 3.1.4) and thus affect nerves, muscle (skeletal and heart) and the central nervous system. Although the symptoms of 'mad honey' poisoning resemble those of the 'cholinergic toxidrome' (Ullah et al., 2018; Gunduz et al., 2012b) investigations in humans could not confirm that both syndromes share a common underlying mechanism (Gunduz et al., 2012b). Initial symptoms of acute poisoning are dizziness, nausea, vomiting, salivation, perspiration and weakness, followed by diarrhoea, paraesthesia, blurry vision and typically hypotension and bradycardia. Higher doses may lead to complete atrioventricular block, convulsions, mental confusion, agitation, loss of consciousness and respiratory depression (e.g. Koca and Koca, 2007; Gunduz et al., 2008; Bostan et al., 2010; Demir et al., 2011; Jansen et al., 2012; Erenler, 2016; Aygun et al., 2018; Ullah et al., 2018; Chest pain, hyperthermia, hypothermia and throat burn have also been reported (Yilmaz et al., 2006; Aygun et al., 2016; Tekinsoy et al., 2017; Ullah et al., 2018).

Hospitalisation in cases of *Rhododendron* honey acute intoxications is most frequently due to its cardiac manifestation. Sinus bradycardia is described to be the most frequent cardiac rhythm disorder that has been observed in *Rhododendron* honey poisoning (e.g. Okuyan et al., 2010; Bayram et al., 2012; Erenler, 2016). In addition, apart from varying degrees of atrioventricular blocks (e.g. Cagli et al., 2009; Erenler, 2016; Ullah et al., 2018), second-degree heart block (Weiss et al., 2010), nodal rhythms (Biberoglu et al., 1988, Gunduz et al., 2006; Aliyev et al., 2009a; Okuyan et al., 2010), atrial fibrillation (Bayram et al., 2012; Cakar et al., 2011; Kalkan et al., 2012; Osken et al., 2012; Sohn et al., 2014), blocked left bundle branch with extreme QT prolongation (Sayin et al., 2011) or with ST segment elevation (Sayin et al., 2012), asystole (Gunduz et al., 2007), interatrial block (Ösken et al., 2021), Wolff–Parkinson–White syndrome (Biberoglu and Komsuoglu, 1988) and myocardial infarction (Akinci et al., 2008; Ocak et al., 2013) have been reported following *Rhododendron* honey ingestion.

Severity of symptoms and the onset of signs were reported to depend on the amount of honey consumed (Jauhari et al., 2009; Bostan et al., 2010). Atropine is administered intravenously for emergency medical care in severe cases of poisoning (e.g. Özhan et al., 2004; Gunduz et al., 2007, 2008; Bostan et al., 2010; Jansen et al., 2012). Symptoms may persist from a few hours to a few days (e.g. Aygun et al., 2018). In general, patients recover within 1–2 days (e.g. Jauhari et al., 2009; Jansen et al., 2012). Though no fatalities from *Rhododendron* honey poisonings have been reported in modern times (Gunduz et al., 2008; Bostan et al., 2010; Silici and Atayoglu, 2015; Ullah et al., 2018), they may be life-threatening in individuals with cardiac complications (e.g. Koca and Koca, 2007; Gunduz et al., 2008; Ullah et al., 2018).

Rhododendron honey from the Black Sea region has a traditional use as folk medicine e.g. against gastrointestinal disorders, hypertension, heart failure, diabetes, asthma, cough and as an aphrodisiac (e.g. Ullah et al., 2018; Okuyan et al., 2010; Ugur et al., 2019). Gunduz et al. (2006, 2007, 2008) reported that in this area, inhabitants are familiar with these uses of *Rhododendron* honey and its symptoms of poisoning and are able to distinguish these honeys from others, since they induce a sharp burning sensation in the throat. However, acute intoxications with *Rhododendron* honeys have also been reported from consumptions which are in general typical for honey and not related to any uses as alternative medicine, such as at breakfast e.g. on slices of bread or in tea (Kerkvliet, 1981; Nassibou et al., 2020).

From the literature search on human observations more than 1,000 cases of intoxications after ingestion of certain Ericaceae honeys by individuals (e.g. Sohn et al., 2014; Silici and Atayoglu, 2015; Ullah et al., 2018), families or groups (e.g. Jauhari et al., 2009; Weber et al., 2009) were retrieved. Details will only be given below for reports in which analytical data on the presence of grayananes in the ingested honey were available or where systematic studies investigating the toxicological effects in patients were conducted.

Only a few of these published intoxication cases provide quantitative information on the content of grayananes in the honey (Kerkvliet, 1981; Aygun et al., 2015; These et al., 2015; Nassibou et al., 2020). Details of these reports are shown in Table 9.²⁰ Information on underlying diseases of the patients is only given in one of the cases, information on medication of the patients is completely missing.

²⁰ From the report by Aygun et al. (2015), the GTX content in the honey consumed was quoted as indicated in the text of the publication, since in its abstract and Table 5, the indication of the units was obviously subject to an error.



18314732, 2023

)23.7866 by Cochra

	•			, 5		5,		
Affected Typ subjects as o	Type of honey		Analytes, measured	Estimated single		Onset of	Symptoms/ Outcome	
(age, bw, medical history when reported)	as described (geographical and botanical origin when reported)	Intake of honey in g per person	concentration of grayananes in the honey (mg/kg)	dose of grayananes (μg/kg bw) ^(a)	Levels in blood (b) and urine (u)	symptoms (min/hours after consumption)	(blood pressure (bp, in mmHg) and heart rate (hr, in beats/ min) when reported)	Reference
4 male students	Honey from Eastern Nepal ^(b) , habitat of <i>R.</i> <i>arboreum</i> and <i>R.</i> <i>campanulatum</i>	Day 1: 10 g/ pers. Day 2: 20–100 g/pers. Day 3: 'a little in the tea'	GTX I, II, III: n.d. Two 'grayanotoxin analogues' (not identified): 30 mg/kg	Two 'grayanotoxin analogues' (not identified, other than GTX I, II, III): Day 1: 4.3 µg/kg bw Day 2: 8.6–43 µg/ kg bw Day 3: < 4.3 µg/kg bw	n.r.	Day 2: 25 min	Day 1: headache and feeling 'high' Day 2: all experienced pain in the chest and around the heart, tingling in the fingers and other extremities, looseness of some muscles, feeling 'high', dizziness, bradycardia (hr: 38), fainting. After 3 h, the symptoms disappeared Day 3: headache	Kerkvliet, 1981
1 male, 56-year-old	'Wild Turkish honey'	Two tablespoons, amounting to approx. 25 g/pers.	GTX III ^(d) : 54.0 mg/kg Total grayananes ^(e) : 358 mg/kg	GTX III: 19.3 µg/kg bw Total grayananes ^(e) : 128 µg/kg bw	n.r.	2 h	and feeling 'high'(C) Presented in hospital with severe bradycardia and loss of consciousness causing traffic accident	These et al., 2015

Table 9: Re	eports on cases of	acute intoxications with	Ericaceae hone	ey allowing	to estimate the	intake of grayananes
-------------	--------------------	--------------------------	----------------	-------------	-----------------	----------------------


18314732, 202

Affected subjects (age, bw, medical history when reported) 1 male,	Type of honey as described (geographical and botanical origin when reported) 'Mad honey' from	Intake of honey in g per person	Analytes, measured concentration of grayananes in the honey (mg/kg) GTX I: 0.61 mg/kg	Estimated single dose of grayananes (µg/kg bw) ^(a) GTX I: 0.94 µg/kg	Levels in blood (b) and urine (u) GTX I:	Onset of symptoms (min/hours after consumption)	Symptoms/ Outcome (blood pressure (bp, in mmHg) and heart rate (hr, in beats/ min) when reported) Presented in	Reference Aygun et al., 2015
67-year-old	Arsin, Turkish Black Sea region, habitat of <i>R.</i> <i>ponticum</i> and <i>R.</i> <i>luteum</i>		GTX III: 2.53 mg/ kg GTX I+III: 3.13 mg/kg	bw GTX III: 3.90 μg/kg bw GTX I+III: 4.84 μg/kg bw	b: 86.2 ng/mL u: 0.16 mg/mL GTX III: b: 8.19 ng/mL u: 1.90 mg/mL		emergency department with dizziness, nausea, hypotension (bp: 60/40), bradycardia (hr: 42)	. , , , ,
1 male, 51-year-old	'Mad honey' from Sürmene, Turkish Black Sea region, habitat of <i>R.</i> <i>ponticum</i> and <i>R.</i> <i>luteum</i>	150 g/pers.	GTX I: 7.77 mg/kg GTX III: 11.0 mg/ kg GTX I+III: 18.8 mg/kg	GTX I: 16.7 µg/kg bw GTX III: 23.6 µg/kg bw GTX I+III: 40.3 µg/kg bw	GTX I: b: 19.3 ng/mL u: 0.58 mg/mL GTX III: b: 2.77 ng/mL u: 1.91 mg/mL	2 h	Presented in emergency department with dizziness, nausea, hypotension (bp: 90/60), bradycardia (hr: 40)	Aygun et al., 2015
1 male, 70-year-old	'Mad honey' from Of, Turkish Black Sea region, habitat of <i>R. ponticum</i> and <i>R. luteum</i>	100 g/pers.	GTX I: 0.45 mg/kg GTX III: 2.98 mg/ kg GTX I+III: 3.44 mg/kg	GTX I: 0.64 µg/kg bw GTX III: 4.25 µg/kg bw GTX I+III: 4.90 µg/kg bw	GTX I: b: 8.13 ng/mL u: 0.19 mg/mL GTX III: b: 7.01 ng/mL u: 0.80 mg/mL	2 h	Presented in emergency department with dizziness, nausea, hypotension (bp: 80/50), bradycardia (hr: 35)	Aygun et al., 2015
1 female, 40-year-old	'Mad honey' from Yomra, Turkish Black Sea region, habitat of <i>R.</i> <i>ponticum</i> and <i>R.</i> <i>luteum</i>	16 g/pers.	GTX I: 9.90 mg/kg GTX III: 16.9 mg/ kg GTX I+III: 26.8 mg/kg	GTX I: 2.26 µg/kg bw GTX III: 3.86 µg/kg bw GTX I+III: 6.12 µg/kg bw	GTX I: b: 8.91 ng/mL u: 0.86 mg/mL GTX III: b: 1.69 ng/mL u: 3.38 mg/mL	30 min	Presented in emergency department with dizziness, nausea, hypotension (bp: 80/40), bradycardia (hr: 35)	Aygun et al., 2015



18

Affected	Type of honey		Analytes,			Onset of	Symptoms/ Outcome	
(age, bw, medical history when reported)	(geographical and botanical origin when reported)	Intake of honey in g per person	measured concentration of grayananes in the honey (mg/kg)	Estimated single dose of grayananes (µg/kg bw) ^(a)	Levels in blood (b) and urine (u)	symptoms (min/hours after consumption)	(blood pressure (bp, in mmHg) and heart rate (hr, in beats/ min) when reported)	Reference
1 female, 32-year-old, bw: 65 kg, medical history: ulcerative colitis	'Mad honey' of Nepalese origin, 'handmade without specific identification' ^(f)	20 g/pers. ^(g)	GTX I: 25.8 mg/kg ^(h)	GTX I: < 10 μg/kg bw ⁽ⁱ⁾	GTX I: b: 2.60 ng/mL	2 h (same on 2 nd exposure 2 days later)	Nausea, vomiting, flushing and perioral paraesthesia lasted about 6 hours. Two days later, the honey was consumed again with same symptoms appearing and presentation to the emergency department: sinus bradycardia (hr: 30), normal ECG, arterial hypotension	Nassibou et al., 2020
							(bp: 80/40). Symptoms spontaneously faded within 24 hours	

n.r.: Not reported; n.d.: Not detected.

(a): In cases, where the body weight is unknown, a body weight of 70 kg is assumed.

(b): The honey was rich in pollen and presumably obtained by pressing the combs; composition of pollen (excluding *Betula*): Rosaceae 54%, *Senecio* 16%, *Rhododendron* 14%, *Alnus* 6%. *Prunus, Daphne* and *Magnolia* contributed less than 1%.

(c): At the time of the analysis, only GTX III could be unambiguously identified in the honey since it was the only compound commercially available as a reference standard.

(d): Not considered in the assessment of acute dose-effect relationships, since it is unclear if the effects are only associated with the honey intake on the third day or also a consequence of the exposure in the previous 2 days.



- (e): For other grayananes than GTX III, a semi-quantitative approach was taken, assuming that their mass spectrometric response was comparable with that of GTX III. The following estimates were made (mg/kg honey): grayanotoxin XVII (6.2), C20H31O6 (12.1), rhodojaponin VI (1.4), rhodomollein XVIII or kalmitoxin I (9.5), rhodomollein XIX (8.3), GTX VII (116.3), GTX VIII or XIX (5.8), GTX II or VI (2.0), craiobiotoxin I, II, V or VIII (18.6), GTX XVIII (39.8), C20H33O4 (3.2), GTX I or rhodomollein XIII (58.3), GTX XIII or XIV (12.2), GTX IV, IX, X or XVI (10.5). Based on these levels and the content of GTX III, the estimates of the concentration in honey and of the resulting single dose for total grayananes were calculated.
- (f): The botanical origin of the Nepalese honey not defined but assumed to be derived from Ericaceae.
- (g): As estimated by the authors; unclear if the amount is consumed at a single occasion or in total on two occasions.

(h): As estimated by the authors.

(i): At the time of the analysis, only GTX I could be identified since it was the only compound commercially available as a reference standard.

Information from other sources

EFSA launched a *call for stakeholder inputs in relation to the mandate on grayanotoxins in 'certain honeys*²¹ requesting information on 'mad honey' intoxications registered in their countries. In Germany, five cases of acute intoxications associated with consumption of honey containing grayananes have been registered from 1990 to 2021 in the BfR National Case Database of Poisonings. Three of these honeys had Turkish origin and two were of unknown origin. Grayanotoxins were identified (but not quantified) in the five honey samples. All subjects were male adults (Germany, 2022, under documentation provided to EFSA). In France, five cases of 'mad honey' intoxications were reported from 2000 to 2020 and involved four male and one female adults. The two more recent cases were honeys from Nepal (2019–2020) and the other three from Turkey. None of the samples had been purchased in Europe (France, 2022, under documentation provided to EFSA). According to the Turkish Ministry of Health National Poison Information Centre, 54, 94 and 187 cases on honey and its derivatives were reported in 2018, 2019 and 2020, respectively (Turkey, 2022, under documentation provided to EFSA).

3.1.3.2. Studies on cases of acute intoxications due to ingestion of 'certain honey'

Acute exposure to 'mad honey'

Several studies were performed in Turkish hospitals to investigate the clinical picture in cases in which patients came to the emergency department due to 'mad honey' intoxication.

Sumerkan et al. (2013) analysed the data from 246 patients (58 females; age 56 \pm 15 years; 188 males, age 52 \pm 15 years) diagnosed with 'mad honey' intoxication between January and December 2011. On admission, 61 patients (28%) had experienced severe bradycardia (heart rate \leq 40 beats per min (bpm)), 33 patients (20%) severe hypotension (systolic blood pressure \leq 60 mmHg) and four patients (1.6%) presyncope or syncope.

From smaller studies, the multiple clinical manifestations are reported as summarised in Table 10 (Yavuz et al., 1991; Yilmaz et al., 2006; Okuyan et al., 2010; Uzun et al., 2013; Yaylaci et al., 2014, 2015; Bilir et al., 2017; Tekinsoy et al., 2017; Aygun et al., 2018).

Symptoms observed	Affected patients (% range)
Hypotension	87–100%
Bradycardia	43–95%
Nodal rhythm	3.7–21%
Atrial fibrillation	1.2–12.5%
First degree atrioventricular block	12.5–14.6%
Complete atrioventricular block	2.2–36%
Dizziness	51.7–100%
Nausea/vomiting	27.6–100%
Blurred vision	15.9–88%
Exhaustion/weakness/fatigue	35–100%
Change of consciousness	12.5–67%
Syncope	12-44%

Table 10:Range of symptoms in affected patients

In two of these studies the relationship between the poisoning symptoms and the grayanotoxin levels in the blood of the patients were analysed (Tekinsoy et al., 2017; Aygun et al., 2018). In the study of Tekinsoy et al. (2017), the blood analyses in the group of 36 patients with the diagnosis of 'mad honey' intoxication revealed concentrations of grayanotoxins (type not specified) with an average of 7.9 ng/mL (range: 0.00–30 ng/mL). No statistically significant relationship between the observed symptoms (dizziness, nausea, vomiting, weakness, deterioration, angina) and the blood grayanotoxins levels in the study group was seen. Similar results were reported by Aygun et al. (2018) from 25 cases of 'mad honey' poisoning with mean concentrations in blood of 4.8 ng GTX I/mL and 6.6 ng GTX III/ mL. No association was found between grayanotoxin levels in blood and clinical symptoms which included decreases of systolic blood pressure and heart rate.

²¹ https://www.efsa.europa.eu/en/partnersnetworks/eumembers#focal-points-eu-food-safety-interfaces

In addition to the well-known cardiovascular symptoms of 'mad honey' poisoning as reported in Table 10, a more recent retrospective study on 76 patients with 'mad honey' intoxication showed that an interatrial block was detected in 28 (36%) of them (Ösken et al., 2021).

In the study of Okuyan et al. (2010), the patients' histories of cardiovascular diseases and medications were considered when evaluating the effects of the 'mad honey' intoxication. From the 42 patients included in the trial, 13 had hypertension and five had coronary artery disease. Before admission to hospital, eight patients were on beta-blocker medication and 10 received an antihypertensive therapy,²² including administration of angiotensin-converting enzyme (ACE) inhibitors and of the calcium channel blocker (CCB) amlodipine. The mean heart rate and mean systolic blood pressure level of patients on beta-blocker therapy at admission were 35 \pm 6 bpm (25–54 bpm) and 68 \pm 14 mmHg (50–90 mmHg), respectively, and significantly lower than of patients without known beta-blocker administration (mean heart rate bpm: 43 \pm 8 bpm (28–59 bpm), mean systolic blood pressure: 78 \pm 11 mm Hg (50–100 mmHg)). No significant difference in heart rate and blood pressure were observed in patients receiving antihypertensive therapy compared to patients without this medication. The authors conclude that patients using beta-blockers and antihypertensive therapies seem to be more vulnerable to the cardiac effects of grayanotoxins than others. They recommend that physicians in the Turkish Black Sea area should warn their patients about adverse effects of 'mad honey' and antihypertensive drug combinations.

Repeated daily exposure to 'mad honey'

A prospective study was conducted in the department of cardiology of the Istanbul University (Aliyev et al., 2009b). In all patients presented with slow heart rate or atrioventricular conduction abnormalities between April 2008 and December 2008, the history of non-commercial honey intake was investigated. One hundred and seventy-three patients were referred to the institution with sinus bradycardia and atrioventricular block (various degrees) and/or for permanent pacemaker implantation. The subjects were asked for information on their honey intake. In five of the patients, a history of daily honey intake for a long period of time was revealed as either three to five teaspoons daily for 3 months, or 3–10 teaspoons daily for > 12 months. The honeys consumed originated from non-commercial beekeepers in the Eastern Black Sea region of Turkey. No further information with respect to the botanical origin or the chemical composition of the honeys was reported. All five patients experienced dizziness and presyncope. Discontinuation of the honey consumption resulted in rapid improvement of symptoms and of the electrocardiographic outcome. The Panel noted that data on the presence of grayananes in the honeys repeatedly consumed by the five patients are missing and that, therefore, the study is of limited relevance for the present assessment.

3.1.3.3. Case reports on acute intoxications due to ingestion of certain plants, their parts or preparations

The occurrence of grayananes in foods other than honey derived from certain Ericaceae plants has also led to human cases of intoxications. The same characteristic symptoms as for *Rhododendron* honey intoxications have been described from the consumption of *Rhododendron* liqueur (Choi et al., 2017), of herbal teas prepared from *Pieris japonica* (Aleguas et al., 2008) or the leaves of *Agarista salicifolia* (Martinet et al., 2005) or from the consumption of blossoms of *R. japonicum* (Koda et al., 2016) or *R. mucronulatum* (Lee et al., 2007) due to misinformation on their edibility. However, none of these publications allows an estimate of the dose of grayananes ingested. The publication of Choi et al. (2017) is not considered further due to inconsistencies in the reporting of the GTX I and GTX III concentrations in the liqueur and the subsequent calculated doses.

The large genus *Rhododendron* includes many species for which traditional medicinal uses, including those in Traditional Chinese Medicine and the Ayurvedic medical system have been described (e.g. Popescu and Kopp, 2013). However, acute intoxications may occur as a consequence of the misuse of *Rhododendron* species for medicinal purposes as observed by Poon et al. (2008). A 57-day-old infant was presented after vomiting in a cyanotic and unresponsive state with convulsions (twitching in all four limbs), sinus bradycardia, hypotension and shock to an emergency department in Hong Kong. Symptoms had started 20 min after administration of a bottle of milk mixed with a homemade decoction of *R. simsii* to treat the infant's bronchiolitis. The grandmother had collected the plant in an area nearby believing in the antitussive effect of the plant described in folk medicine. The infant was intubated and put on mechanical ventilation. Seizure activity ceased due to benzodiazepam

²² Number of patients receiving a combined medication of beta-blockers with ACE inhibitors or CCB is not indicated.

infusion. Discharge from hospital took place on day 8 after admission with full recovery confirmed 4 months later. GTX I was identified in residues of the plant and the decoction and in the patient's urine, without quantitative data being presented.

Accidental ingestion of parts (e.g. buds) of *Pieris japonica*, which is grown as a garden plant, is another cause for intoxications of young children with grayananes (Tschekunow et al., 1989; Aleguas et al., 2008; Masom et al., 2008). In a 21-month-old girl presented with pallor, main symptoms of the intoxication with *P. japonica* (unknown which plant parts were ingested) were vomiting and a decrease of the heart rate to 40 bpm (Aleguas et al., 2008). Similar symptoms, burning in the mouth, drooling, lethargy and a slight hypotension were also reported from a 2-year-old boy after ingestion of a 'cluster of *Pieris japonica* buds' (Tschekunow et al., 1989). Apart from bradycardia (71 bpm), lethargia and oral secretions, main complaints observed in a 2-year-old girl after intake of several leaves of *P. japonica* were nausea and vomiting (Masom et al., 2008). All three patients recovered after receiving atropine intravenously. The grayananes levels in the plant material ingested and in the children's blood or urine were not analysed.

3.1.3.4. Clinical studies

Clinical studies in humans investigating the pharmacological effects of grayananes after oral administration were not identified. Results of trials in Chinese hypertensive patients (Mao et al., 1981; Li and Mao, 1982; Chen et al., 1985), showing that intravenous infusion of rhodojaponin III (synonym: rhomotoxin) was effective in lowering systolic and diastolic blood pressures and influenced other cardiovascular parameters (e.g. reduction of left ventricular afterload, of heart rate and of myocardial oxygen consumption), did not allow conclusions to be drawn on dose–effect relationships of grayananes orally ingested with food. Therefore, these reports were not considered relevant for this risk assessment and are not further described.

3.1.3.5. Summary of observations in humans

Overall, it can be concluded that in regions where consumption of preparations originating from grayanane-containing Rhododendron species, such as 'mad honey', is popular, acute intoxications typical from the ingestion of grayanotoxins are consistently seen in emergency departments. Acute symptoms observed affect the muscles and the nervous and cardiovascular systems with bradycardia and hypotension being most prominent and without fatal cases being reported in the last decades. In acute poisoning cases observed with Rhododendron honeys from the Turkish Black Sea region, the estimated intakes on a single occasion of GTX I or GTX III as part of the mixture of grayananes present in the honey range from about 0.6 to 17 μ g/kg bw or 3 to 24 μ g/kg bw, respectively, and from 4.8 to 40 μ g/kg bw for the sum of GTX I and GTX III. A possible contribution of other grayananes present in Rhododendron honeys from the Turkish Black Sea region to the development of adverse effects cannot be quantitatively assessed due to insufficient information on their occurrence in these honeys and on their toxicological profile. From cases of acute intoxications following the ingestion of a Nepalese Rhododendron honey, in which GTX I and III could not be detected, it can be concluded that also grayananes other than GTX I and III present in Rhododendron species cause adverse effects on the muscles and the nervous and cardiovascular systems, when ingested in a single dose as low as 4.3 ug/kg bw. An NOAEL for GTX I, GTX III, for one of their analogues or for total gravananes could not be identified from the data available for human acute exposure to these compounds.

From the human case reports on poisonings, there are no indications that 'mad honey' or grayanane containing plants or their preparations or components induce liver toxicity as has been observed in experimental animal studies.

For human chronic exposure, reported symptoms do not differ from those of acute exposure. However, information on dose–effect relationships for long-term intake of grayananes could not be identified.

Available data are too limited to draw conclusions if children or patients with certain underlying diseases, are more vulnerable towards the toxicity of grayanotoxins than healthy adults. However, results from Okuyan et al. (2010) suggest that patients with cardiovascular diseases, when receiving anti-ischaemic or antihypertensive therapy (e.g. beta-blockers), are at higher risk of developing adverse cardiovascular effects when exposed to grayanotoxins compared to healthy subjects.

3.1.4. Mode of action

Regarding acute toxicity, the Panel decided to focus on the endpoint of most relevance for the hazard characterisation, i.e. the action of grayananes on the voltage-gated sodium (Nav) channels. The most relevant papers are addressed in this section.

Grayananes are so far considered the effectors of *Rhododendron* honey toxicity in humans, being able to reproduce similar symptoms, both in terms of quality and duration, in animal models. *Ex vivo* and *in vitro* studies on brain, muscle and cardiac preparation identify the voltage-gated sodium (Nav) channels as molecular target of grayanotoxins (Seyama and Narahashi, 1973; Narahashi and Seyama, 1974; Creveling et al., 1980; Matthews et al., 1981; Frelin et al., 1982; Kim et al., 2010). Nav involvement also comes from indirect evidence showing that Na⁺ channel blocker, procaine, counteracts reduction of blood pressure and heart rate induced by rhodojaponin III in animal models (Mao et al., 1982).

Nav channels are multimeric complexes consisting of the pore-forming α -subunit associated with auxiliary β -subunits (See Figure 3) that modulate the kinetics and voltage dependence of channel gating and are involved in channel localisation (Catterall, 2000). β -Subunits display tissue specificity, heart contain a mixture of β_1 - β_4 subunits, while skeletal muscles have only the β_1 subunit (Isom, 2001; Brackenbury and Isom, 2011). The pore-forming α -subunit is organised in four homologous domains (Iuphar: I-IV, literature: D1-D4), each one with six transmembrane hydrophobic domains (S1-S6) (Figure 3). S5 and S6 segments form the inner cavity of the pore, while S4 segments contain positive residues (usually arginine) which confer sensitivity of the channels to depolarisation initiating their activation (Catterall, 2000). Inactivation of the channel is ruled by the short intracellular loop in between domain III and IV (Catterall, 2000). Nine mammalian Nav channel isoforms (Nav1.1-Nav1.9) have been identified, which share greater that 50% identical amino acid sequence in transmembrane and extracellular domains but show different tissue presence and pharmacological features (Catterall, 2000).



Figure 3: Subunit structure of the voltage-gated sodium channels, based on Catterall, 2000. The binding site is indicated with a blue * in the figure

Agents that act on Nav bind to receptor sites on the pore-forming α -subunit (Cestèle and Catterall, 2000). Mutagenesis studies showed that critical replacement of amino acids located within S6 segment on all four domains abolishes GTX I effects on rat skeletal muscle and heart Nav (Ishii et al., 1999; Kimura et al., 2001). In addition, relative conductance results show that skeletal muscle channels are more sensitive to GTX I than cardiac channels due to the specific characteristics of the intracellular loop of domain I S4-S5 and Ile-433 in the transmembrane segment of domain I S6 (Kimura et al., 2001). GTX I binds only to the open state, driven by membrane depolarisation, and dissociates from the closed state (Ito et al., 1985; Maejima et al., 2002). Due to this mechanism, excitable cells (neurons, muscle, myocytes) appear to be sensitive to GTX (Hotta et al., 1994; Ishii et al., 1999; Kimura et al., 2001). GTX I binding to the four domains of the pore-forming α -subunit prolongs the open time of Na⁺ channels and shifts the sodium–conductance–voltage curve in the

direction of hyperpolarisation, thus inhibiting inactivation of voltage-gated Na channels (Maejima et al., 2002).

Since Nav channels activation and deactivation in excitable cells are responsible for action potential initiation and propagation, these changes in gating kinetics cause long-lasting depolarisation of excitable cells and increase the frequency of post-synaptic currents in primary neurons facilitated by pre-synaptic neurotransmitters release (Seyama, 1978, Hong and Chang, 1983; Ito et al., 1985; Kim et al., 2010). GTX I and rhodojaponin III-induced decreased blood pressure, contractile force and heart rate, are strongly reduced after vagotomy in animal models (Moran et al., 1954; Cotten et al., 1956; Mao et al., 1982). It is thus possible to hypothesise that these two grayananes also induce long-lasting depolarisation of the vagus nerve through Nav channels facilitating acetylcholine release. This might also explain the efficacy of atropine, a non-specific anti-muscarinic agent, to counteract the effects of rhodojaponin III on blood pressure and heart rate (Mao et al., 1982). In accordance with the role of grayananes in *Rhododendron* honey toxicity, both atropine and AF-DX 116, a selective M2-muscarinic receptor antagonist, restored bradycardia and atropine also respiratory rate depression induced by toxic *Rhododendron* honey extract in the rat (Onat et al., 1991a,b). Indeed, atropine is effective in the treatment of human intoxication with *Rhododendron* honey.

Nav channel subtypes display different kinetic, voltage-dependent and pharmacological properties (Catterall, 2000). The different distribution among cells thus contributes to distinct functional roles and sensitivity to targeting substances. Information on how different Nav isoforms respond to grayananes are sparse. GTX I exerts a greater effect on Nav 1.4 compared to Nav 1.5 (Kimura et al., 2001), although activity on other Nav channel subtypes remains to be determined.

No mechanistic data on genotoxicity and repeated dose toxicity were identified.

3.1.5. Relative potencies of grayananes for acute toxicity

The following discussion on differences in the potencies of grayananes relates only to acute toxicity. There is no data on the differences in potency related to the genotoxicity of grayananes.

Single-dose studies used to estimate LD_{50} values in experimental animals (see Section 3.1.2.1) describe similar symptoms to those reported in humans after consumption of 'mad honey'. Thus, the possibility of using the LD_{50} values in animals to rank the potency of grayananes and derive RPFs for acute toxicity was investigated.

It was noted that the studies deriving these LD_{50} values in rodents use mainly the i.p. or i.v. route of exposure instead of the oral route. Because of their structural similarity, it was assumed that the relative potency of substances was not markedly affected by the route of exposure.

Thus, the Panel derived RPFs for groups of grayananes and ranked them in Table 11 based on the LD_{50} values derived from i.p. studies in mice, as shown in Table 6 (when data were available). When data were not available, grayananes were ranked based on their structural alerts, as described in Tables A.1–A.3 in Appendix A and shown in Figure 4. In the absence of LD_{50} values, grayananes are allocated to groups 1–3 as shown in Table 11, based on the number of structural alerts. In case of grayananes with three structural alerts (group 1), the default allocation would be to group 1A.





The Panel noted that all grayananes with high potency share three structural features circled in Figure 4: They all contain (i) a 3β -hydroxy or a 2β , 3β -epoxy group, (ii) a 6β -hydroxy and (iii) a 10β -methyl group (Hikino et al., 1976; Li et al., 2013; Masutani et al., 1981). When one of these structural features is modified or missing, toxicity is greatly reduced, as in the case of GTX II and GTX IV where the 10β -methyl group is replaced by a C10 exocyclic double bond.

18314722, 2223, 3, Downloaded from https://efsa.onlinelibrary.wiey.com/doi/10.2903j.efsa.2023,7866 by Cochranetalia, Wiey Online Library on [02/03/2023]. See the Terms and Conditions (https://nlinelibrary.wiey.com/terms-and-conditions) on Wiey Online Library for rules of use; OA articles are governed by the applicable Creative Commons

License

Group	Indicativ range, (mg/kg	e LD ₅₀ <i>i.p</i> . bw)	RPF	Grayanane
Group 1, most potent (3 structural alerts)	Subgroup 1A	0.3 ^(a) –0.6	2	e.g. rhodojaponin III
	Subgroup 1B	0.6-2.0	1	e.g. GTX I and GTX III
Group 2, medium potency (2 structural alerts)	10–2	0	0.1	e.g. GTX II and GTX IV
Group 3, no observed toxicity (1 or no structural alert)	> 10	0	< 0.01 ^(b)	e.g. GTX XVI and GTX XVII

Table 11: Relative potency factors for acute toxicity: grouping of grayananes

(a): A lower LD_{50} value is estimated for asebotoxin III (0.1 mg/kg bw), but because the substance is not known to occur in 'certain honey', it was not included in the table.

(b): Assumed negligible contribution to risk assessment calculations (see Section 3.1.7).

Rhodojaponin III has been identified in a wide range of species, including *Rhododendron* spp. Its presence has been implicated in Grouse Mountain honey from British Columbia (Canada) (Scott et al., 1971), but not in 'mad honey' from the Black Sea region (These et al., 2015). Recent results from Nepalese honey samples suggest that rhodojaponin III may be present in these honeys (Mulder and de Vries, 2022 under Documentation as provided to EFSA).

Note was taken that asebotoxin III is approximately 10 times more potent than GTX I and III. However, asebotoxin III has not been identified in or associated with 'mad honey'. So far, asebotoxin III has only been isolated from *P. japonica and P. formosa*, which are not among the sources of 'mad honey' produced in Nepal or the Black Sea region. Based on the above, the Panel did not consider it relevant to derive a separate RPF for asebotoxin III. If new evidence suggests that asebotoxin III occurs in 'mad honey', the Panel will consider the possibility of deriving a separate RPF for the substance.

Table 12 reports the dose levels of GTXs I, II and III affecting locomotive activity and salivation. The results are consistent with those based on LD_{50} and support the derived RPFs.

Grayanotoxin	Locomotion, i.p.	Salivation (EC ₅₀), i.p.	LD ₅₀ , i.p.	RPF
GTX III	0.1 mg/kg	0.15 mg/kg	0.77–1.03 mg/kg	1
GTX I	0.25 mg/kg	0.22 mg/kg	1.1–1.5 mg/kg	1
GTX II	> 0.25 mg/kg	10–15 mg/kg	10–28.7 mg/kg	0.1

Table 12: Relative potencies of GTX I, GTX II and GTX III

3.1.6. Considerations of critical effects and dose-response analysis

3.1.6.1. RP based on acute/single-dose exposure to grayananes

RP based on animal data

The derivation of an RP from experimental animal studies was considered.

Some i.p. and i.v. studies with grayananes or 'mad honey' in experimental animals describe acute effects on heart rate and blood pressure which are consistent with those observed in humans (see Section 3.1.2.1). The Panel noted that the parameters in these studies are altered in a dose-dependent manner (Onat et al., 1991a,b; Turkmen et al., 2013; Yilmaz et al., 2017).

A BMD analysis was performed based on the reduction of heart rate after i.p. injection of GTX III in rats (Turkmen et al., 2013) (see Table 13). Instead of selecting the default benchmark response (BMR) of 5% for continuous effects, the Panel decided to select a BMR of 10% for this endpoint, i.e. a 10% decrease in the mean response compared to the controls, also considering the transient nature of this effect. Using model averaging, a BMDL₁₀ of 15.3 μ g/kg bw was estimated for GTX III. A detailed description of the BMD analysis is presented in Appendix C.

Table 13:	BMDL ₁₀ and BMDU ₁₀	calculated from th	e reduction of heart	rate observed after i.p.
	injection of GTX III in	rats (based on dose	e-response data by Tu	rkmen et al., 2013)

	$BMDL_{10}(\mu g/kg bw)$	$BMDU_{10}$ (µg/kg bw)	Lowest dose in the study (μ g/kg bw)
Results at 60 min	15.3	261	179

www.efsa.europa.eu/efsajournal

18314722, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [0203/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons

License

The estimated $BMDL_{10}$ of 15.3 µg/kg bw for GTX III (*i.p.*) based on Turkmen et al. (2013) was used as reference point (RP) for acute toxicity for the sum of GTX I and III considering that the two substances have similar potencies (see Tables 11 and 12).

Although data from an oral study might derive a higher RP than data from an i.p. study, the Panel decided to not extrapolate results from i.p. to oral exposure for the following reasons:

- i) Although the available toxicokinetic studies do not allow comparison of oral vs. i.p. bioavailability, there are studies in the scientific literature which compare i.p. to oral toxicokinetic values for different molecules. In a study by Turner et al. (2011), the authors report that because the primary route of absorption under i.p. is into the mesenteric vessels and guides the substance through the liver, substances administered i.p. may have similar or same order pharmacokinetics with those seen after oral administration. In addition, Al Shoyaib et al. (2020) reported that substances with low lipid-water partition coefficient (GTX III has a logP of -0.237 ± 0.514 based on SciFinder) are less easily released into the systemic circulation after i.p. injection than substances with higher lipid-water partition coefficient.
- ii) The data allowing comparison between i.p. and oral routes do not come from studies on toxicokinetics but studies on the potency of GTX III, following i.p. and oral administration in male mice (Hikino et al., 1976 and 1979). Both studies by Hikino et al. are subject to uncertainties mainly associated with testing male mice only and the gross endpoint used (mortality).

Although derived for the sum of GTX I and III, this group RP for acute toxicity may apply to other grayananes, such as GTX II, by making use of their RPFs (e.g. for GTX II with RPF = 0.1, concentrations in food samples would have to be converted by using the factor 0.1, see Table 11).

Uncertainties associated with the use of i.p. instead of oral studies, the oral bioavailability of GTX I and III, the effect of the matrix (honey), the higher and/or lower potency of other co-occurring grayananes in honey, etc., were considered in the uncertainty analysis (see Section 3.6).

Other effects observed in single-dose studies in experimental animals are small alterations in clinical parameters which are inconsistent and not supported by histopathological findings.

RP based on human data on acute intoxications

Consumption of dietary sources rich in grayananes is associated with acute effects on the nervous and cardiovascular systems, the most prominent ones being bradycardia and hypotension. As doseresponse data in humans relate to cases of poisoning from consumption of 'mad honey', an NOAEL for grayanotoxins or other grayananes could not be identified. In the absence of an NOAEL from human studies, the Panel referred to dose–response data derived from intoxication cases.

In two poisoning cases with 'mad honey' from Turkey reported by Aygun et al. (2015), the combined intake of GTX I and GTX III resulted in total doses of 4.84 and 4.90 μ g/kg bw (GTX I: 0.94 and GTX III: 3.90 μ g/kg bw or GTX I: 0.64 and GTX III: 4.25 μ g/kg bw, respectively) and led to presentation in an emergency department with dizziness, nausea, hypotension and bradycardia (see Table 9).

In a separate publication, acute intoxications with symptoms of headache and feeling 'high' have been associated with intake of Nepalese honey containing two grayananes that were different from GTX I, II, III and their structures were not identified. The doses were estimated to be 4.3 μ g/kg bw on the first day and < 4.3 μ g/kg bw on the third day of 'mad honey' consumption (Kerkvliet, 1981). It was noted, however, that the results from this publication were semi-quantitative and therefore can only be used as supportive data.

From the data by Aygun et al. (2015), the lowest observed acute intoxication level identified is 4.8 μ g/kg bw for the sum of GTX I and GTX III following single exposure in humans. In the absence of an NOAEL, the Panel estimated an RP of 0.48 μ g/kg bw for the sum of GTX I and GTX III, at which no adverse effects would be expected, by applying an uncertainty factor of 10 to the lowest observed poisoning dose. This approach is prone to a number of additional uncertainties, including the unknown contribution of other grayananes to the poisoning effect of the 'mad honey' sample (see Section 3.6).

3.1.6.2. RP based on repeated dose exposure to grayananes

In an *in vivo* micronucleus (MN) assay in male mice, both, grayanane-containing *Rhododendron* honey and the isolated GTX III, showed a clear genotoxic effect in the presence of bone marrow exposure (Rasgele et al., 2021).

No studies in experimental animals on chronic toxicity/carcinogenicity are available.

Effects observed in subacute and subchronic studies in experimental animals were small alterations in clinical parameters which were inconsistent and not supported by histopathological findings.

Histological findings in the testes from repeated dose i.p. injection of GTX III in rats at doses from 0.1 to 0.8 μ g/kg bw per day for 3 weeks suggest effects on the testes. Other critical effects could not be confirmed because of the lack of repeated-dose toxicity data in both animals and humans.

Therefore, no RP could be identified from repeated dose toxicity studies on grayananes.

3.1.7. Consideration of a Health-Based Guidance Value (HBGV) and a margin of exposure

Due to the indicated *in vivo* genotoxicity for the pure GTX III and for one honey sample containing GTX I and III, a health-based guidance value (HBGV) cannot be derived for grayananes (EFSA Scientific Committee, 2017b).

However, considering that single exposure to 'certain honey' may lead to intoxication in humans, the Panel decided to carry out an acute risk assessment.

Regarding the risks for acute toxicity, the Panel considered the overall data situation to be limited for the following reasons:

- i) No adequate animal studies with oral administration of grayanotoxins are available,
- ii) The few quantitatively evaluable cases of poisoning in humans do not allow conclusions to be drawn about possible adverse effects of GTX I and III in the low-dose range,
- iii) Animal and human data on the toxicological characterisation of grayananes other than GTX I and III are sparse or missing,
- iv) No adequate data to perform an acute exposure assessment are available.

Although exposure data on acute consumption were limited, the Panel performed a margin of exposure (MOE) calculation for acute toxicity based on exposure estimated from selected concentrations for the sum of GTX I and III reflecting the concentrations measured in `certain honeys' and the RP of 15.3 μ g/kg bw derived from one study in rats (Turkmen et al., 2013), in order to assess the health concern (See Section 3.4.1.1).

Based on the above considerations, the Panel calculated *highest concentration(s)* of grayanane below which no acute effects on heart rate and blood pressure would be expected following acute consumption of honey containing grayananes by applying a factor of 100 for inter- and intraspecies variability (WHO/ IPCS, 2005; EFSA Scientific Committee, 2012) (see Section 3.5). This assumes that the two substances have equivalent potency (see Hikino et al., 1976 and 1979) and considers that GTX I and III are the only two grayananes consistently identified and quantified in 'certain honey' (see Section 3.2).

3.2. Occurrence data

3.2.1. Previously reported occurrence data in the open literature

A limited number of surveys on GTXs in honey and other food products has been published in the open literature. Table 14 reports results from these surveys, where analyses were performed for GTX I, and/or GTX III, and/or the sum of GTX I and III in honeys and other foods. In some of the surveys, (semi-)quantitative results for grayananes are also presented. The analytical methodologies supporting the results for GTX I and III were conducted using LC-MS-based techniques and were based on fit-for-purpose protocols with regard to the availability of standards, accuracy and reproducibility. The sensitivity of some methods, however, is not low enough to quantify GTX I and III in honeys at concentrations as low as 0.05 mg/kg (see Section 3.5).

3.2.1.1. Grayanane occurrence in honeys from the EU

Recently, Lucatello et al. (2022) analysed 125 samples of *Rhododendron* honey collected in the Italian alpine regions of Valle d'Aosta, Piemonte, Lombardia, Trentino-Alto Adige and Veneto, for the content of GTX I (LOQ: 0.01 mg/kg) as well as for pollen and different sensory parameters. The samples were collected over a 3-year period (2017–2019). Concentrations for GTX I ranged from < LOQ to 0.10 mg/kg honey and, in total, 38 out of 125 (30%) of the samples contained measurable amounts of GTX I (see Table 14). The samples were also screened for the presence of GTX III, but only semi-quantitative information is reported in the paper suggesting that GTX III is present in 74% of samples with quantifiable levels of GTX I, at levels similar to or lower than GTX I.

These et al. (2015) analysed 49 samples from the German retail market, including beekeeper honeys, from various sources, but no GTXs could be detected in any of the samples (LOQ \leq 0.17 mg/kg). The origin of these samples is reported in Table 14. The Panel noted that only one sample of 'Heather honey' was of Ericaceae origin but is aware that polyfloral honeys may also contain a contribution from *Rhododendron* species. As the LOQ for the method used is not specified in the study, the Panel cannot assess whether it could identify and quantify GTX I and/or GTX III at concentrations lower than 0.17 mg/kg.

Mulder and de Vries (2022 under Documentation as provided to EFSA) also screened honeys from the EU for grayanane occurrence (LOQ: 0.1 mg/kg). These included European 'Heather honey' from the Netherlands and France, one *Rhododendron* honey from Spain and one Arbutus honey from Portugal. No grayananes were identified in any of these honeys, but the method could not quantify grayananes at concentrations below 0.1 mg/kg.

3.2.1.2. Grayanane occurrence in non-European honeys

Rhododendron honey samples collected from the Turkish Black Sea region were found to contain grayananes. In 2017, Dönmez and Kaya conducted a survey in the Ducze area, where honey is often sold as 'mad honey' by local people (Dönmez and Kaya, 2020). Eighty-one samples were analysed for their GTX III content, which ranged from < LOQ (0.001 mg/kg) to 0.07 mg/kg. Kurtoglu et al. (2014) collected the largest number of samples (187) over the course of 3 years (LOQ: 0.1 mg/kg). The samples were pooled per year and only average concentrations were published. GTX I was present in higher concentrations than GTX III (Table 14) and there was some yearly variation with higher levels present in honey in 2011 and 2012 compared with 2010. In his Ph.D. thesis, Sahin (2014) reported results for GTX III in six 'suspect' honey samples, Rhododendron honey (16 samples) and royal jelly (bee milk) (8 samples) collected from different locations in the Black Sea region (Sahin, 2014, see also Sahin et al., 2015). All samples contained GTX III, the highest levels being found in the 'suspect' honeys, while the royal jelly contained the lowest amounts (LOQ: 0.01 mg/kg). Smaller surveys were reported by Kaplan et al. (2014) (10 samples; LOQ: 0.01 mg/kg) and Silici et al. (2016) (6 samples; LOQ not reported); the latter survey only included GTX III. Kaplan et al. (2014) reported GTX I to be present in higher concentrations than GTX III. From these surveys, it can be concluded that there is a wide variation in total GTX concentration in the samples originating from the Black Sea region, ranging from < 0.001 mg/kg to 74 mg/kg.

Lee et al. (2008) analysed a set of wild honeys and comb honeys produced in the Republic of Korea as well as imported products. Most honey types from domestic and foreign origin did not contain measurable amounts of GTX I, II and III (LOD: 0.15 mg/kg). However, four out of eight wild honeys imported from Nepal contained substantial amounts of GTXs, ranging from 3.4 to 17 mg/kg. GTX I was the dominant compound in the samples (3.1–13 mg/kg). GTX II (< LOD-1.1 mg/kg) and GTX III (0.25–3.3 mg/kg) were present at lower levels in the four positive samples.

Recently, Ahn et al. (2022) reported results for GTX I and III in 60 samples of 'mad honey' from Nepal confiscated at airports in the Republic of Korea. Thirty-three samples contained GTX I and III (LOQ: 0.25 mg/kg) in concentrations ranging from 0.8 to 65 mg/kg for GTX I and from 0.25 to 64 mg/kg for GTX III. The sum of GTX I and III ranged from 1.0 to 100 mg/kg. In the positive samples, the average concentration of GTX I (25 mg/kg) was somewhat higher than the average concentration of GTX III (17 mg/kg).

Mulder and de Vries (2022 under Documentation as provided to EFSA) recently analysed two *Rhododendron* honey samples from Turkey (including one 'mad honey' sample purchased on the internet) and nine *Rhododendron* honey samples from Nepal (Table 14). In two samples from Nepal and one sample from Turkey, GTX I, II, III and rhodojaponin III were (semi-)quantified. In the absence of an analytical standard of rhodojaponin III, the substance was prepared from commercially available rhodojaponin II. This allowed confirmation of the presence of rhodojaponin III in the 'mad honey' samples. However, only semi-quantitative concentrations were obtained for rhodojaponin III since the purity of the compound was not checked.

Semi-quantitative results from the studies by Mulder and de Vries (2022 under Documentation as provided to EFSA) and These et al. (2015) suggest that grayananes other than GTX I, II and III cooccur in *Rhododendron* honeys from the Turkey Black Sea Region and Nepal. The total grayanane concentrations in Turkish honeys may be up to three times higher than the sum of GTX I and III (These et al., 2015). In Nepalese honeys, seeming to be richer in other grayananes, notably rhodojaponin III, the total grayanane concentrations may be up to seven times higher than the sum of GTX I and III (Mulder and de Vries, 2022 under Documentation as provided to EFSA) (see Table 15). It

was noted that the levels of rhodojaponin III in Nepalese honey could not be validated in the absence of a certified analytical standard.

Table 15 reports quantitative and semi-quantitative results on GTXs in honey from intoxication cases.

3.2.1.3. Grayanane occurrence in other foods

For other food categories, only limited data are available. A single study was reported on the presence of GTXs I and III in dietary supplements and homemade wine from local markets in the Republic of Korea (LOQ: 0.008 mg/kg). Six out of 39 supplements were found to contain relatively low levels of GTXs (Table 14). This was also the case for eight out of 12 homemade wine samples: total concentrations were below 0.5 mg/L. However, four wine samples contained high levels of GTX I and III, in the range of 64 and 154 mg/L for the sum (Hwang, 2018). The authors correlated the high levels of GTXs in wine to the use of *R. brachycarpum*, which is collected locally.



18314732

10.2903

Country of sampling	Year of survey	Matrix	N	No of positives	LOD or LOQ (mg/ kg)	GTX I (mg/kg)	GTX II (mg/kg)	GTX III (mg/kg)	GTX total ^(a) (mg/kg)	GTX average ^(b) (mg/kg)	Remarks	Reference
Grayanan	e occurre	nce in honeys o	of EU (origin								
Italy	2017	Italian <i>Rhododendron</i> honey	45	16	0.01	< LOQ- 0.07	n.a.	n.a.	< LOQ-0.07 (< LOQ-0.15) ^(c)	0.04 (pos) 0.01 (all)	Samples from alpine region, where <i>R</i> .	Lucatello et al., 2022
	2018	Italian <i>Rhododendron</i> honey	45	13	0.01	< LOQ- 0.07	n.a.	n.a.	< LOQ-0.07 (< LOQ-0.14) ^(c)	0.03 (pos) 0.01 (all)	<i>ferrugineum</i> is abundant	
	2019	Italian <i>Rhododendron</i> honey	35	9	0.01	< LOQ- 0.10	n.a.	n.a.	< LOQ-0.10 (< LOQ-0.21) ^(c)	0.05 (pos) 0.01 (all)		
Germany	n.i	Retail honey (18), beekeeper honey (6)	24	0	< 0.17	< LOD	< LOD	< LOD	< LOD	< LOD	Samples from Germany (6), Hungary (2), Spain (2), Austria (1), Italy (1), Romania (1), other EU countries (12)	These et al., 2015
Nether- lands	2021	Ericaceae honey	4	0	0.1	< LOQ	< LOQ	< LOQ	< LOQ	< LOQ	Samples (4) from the Netherlands and France, Spain and Portugal	Mulder and de Vries, 2022 under Documentation as provided to EFSA
Grayanan	e occurre	nce in honeys o	of non	-EU origin								
Germany	n.i	Retail honey (25)	25	0	< 0.17	< LOD	< LOD	< LOD	< LOD	< LOD	Samples from South America (7), China (7), Turkey (4), not specified (7)	These et al., 2015

Table 14: Summary results of surveys conducted on GTXs in honey and other foodstuffs



18314732, 202

Country of sampling	Year of survey	Matrix	N	No of positives	LOD or LOQ (mg/ kg)	GTX I (mg/kg)	GTX II (mg/kg)	GTX III (mg/kg)	GTX total ^(a) (mg/kg)	GTX average ^(b) (mg/kg)	Remarks	Reference
Korea	n.i.	Korean honey	25	0	0.15	< LOD	< LOD	< LOD	< LOD	< LOD		Lee et al., 2008
		Korean wild honey	21	0	0.15	< LOD	< LOD	< LOD	< LOD	< LOD		
		Korean comb honey	13	0	0.15	< LOD	< LOD	< LOD	< LOD	< LOD		
		Imported honey	44	0	0.15	< LOD	< LOD	LOD < LOD < LOD < LOD < LOD Samples fr New Zealar (11), Italy Canada (8) (7), France Australia (3 others (2)	Samples from New Zealand (11), Italy (9), Canada (8), USA (7), France (4), Australia (3), others (2)			
		Imported wild honey	8	4	0.15	3.1-13	< LOD-1.1	0.25-3.3	3.4-17	11 (pos) 5.5 (all)	All samples imported from Nepal	
Korea	n.i.	'Mad honey' from Nepal	60	33	0.25	< LOQ-65	n.a.	< LOQ-64	< LOQ-100	42 (pos) 23 (all)	Samples were confiscated by customs at Korean airports	Ahn et al., 2022
Nether- lands	n.i.	Honey from Nepal	9	2	0.1	< LOQ-8.5	< LOQ- 0.50 ^(d)	< LOQ-2.4	< LOQ-11 (14) ^(e)	7.2 (pos) (9.9) ^(e) 1.7 (all) (2.2 (all)) ^(e)	Several other GTXs tentatively identified and quantified in the 2 honeys, including rhodojaponin III (up to 1.2 mg/kg)	Mulder and de Vries, 2022 under Documentation as provided to EFSA
	n.i.	Honey from Turkey	2	2	0.1	0.27–0.94	< LOQ- 0.22 ^(d)	< LOQ-1.53	0.27–2.7 (0.27–3.3) ^(e)	1.5 (1.8) ^(e)	Rhodojaponin III (0.34 mg/kg) was detected in one sample	



18314732, 202

Country of sampling	Year of survey	Matrix	N	No of positives	LOD or LOQ (mg/ kg)	GTX I (mg/kg)	GTX II (mg/kg)	GTX III (mg/kg)	GTX total ^(a) (mg/kg)	GTX average ^(b) (mg/kg)	Remarks	Reference
Turkey	2010- 2012	<i>R. ponticum</i> honey	187 (3) ^(f)	-	0.1	16–26	n.a.	2.5–12	19–38	30	Samples from western Black Sea region Only aggregated data are presented	Kurtoglu et al., 2014
Turkey	2012	<i>Rhododendron</i> honey	6	6	n.i.	n.a.	n.a.	2.1–11	2.1–11	6.2	Monofloral <i>Rhododendron</i> honeys from Black Sea region	Sibel et al., 2014; Silici et al., 2016
Turkey	n.i.	Honey, local production	10	10	0.01	< LOD-39	n.a.	< LOD-35	0.1–74	22	Samples from Black Sea region	Kaplan et al., 2014
Turkey	n.i.	'Suspect' honey	6	6	0.01	n.a.	n.a.	11–69	11–69	25	Samples from Black Sea region	Sahin, 2014 ^(g)
Turkey	2009- 2012	<i>Rhododendron</i> honey	16	16	0.01	n.a.	n.a.	0.63–39	0.63–39	7.5	Samples from Black Sea region	Sahin, 2014 ^(g)
Turkey	2013	Royal jelly/ bee milk	8	8	0.01	n.a.	n.a.	0.19–7.3	0.19–7.3	2.4	Samples from Artvin area, Turkey	Sahin, 2014
Turkey	2017	Honey, local production	81	52	0.001	n.a.	n.a.	< LOQ-0.067	< LOQ-0.067	0.014 (pos) 0.009 (all)	Samples from Duzce area, Turkey	Donmez and Kaya, 2020
Grayanane	e occurre	nce in other fo	ods									
Korea	n.i.	Dietary supplements	39	6	0.008	0.087– 0.347 ^(h)	n.a.	0.037– 0.127 ^(h)	0.124–0.474 ^(h)	0.301 (pos) ^(h) 0.046 (all) ^(h)	Obtained from local Korean markets. All positive samples were liquid supplements	Hwang et al., 2018
		Homemade wine	12	12	0.008	< LOQ- 101 ^(h)	n.a.	0.015–56 ^(h)	0.034–154 ^(h)	36.8 ^(h)	Obtained from local Korean markets	



n.i: not indicated; n.a: not analysed.

- (a): Calculated concentration range for the sum of GTX concentrations per sample; the calculations do not take into account that GTX II has a lower RPF than GTX I and III.
- (b): Average calculated from the GTX I, II and III total concentration divided by the number of samples.
- (c): Semi-quantitative sum of GTX I and GTX III; no analytical standard was available for GTX III in the study by Lucatello et al. (2022). Assuming an equal response between the isoforms of GTX I and GTX III, the authors estimated that 73.7% of samples with GTX I contain equal or lower levels of GTX III but never higher.
- (d): Semi-quantitative result.
- (e): Semi-quantitative sum of all (tentatively) identified grayananes.
- (f): Only aggregated results per year are reported.
- (g): Data shown are from Sahin (2014). The results for a selection of 10 samples are also reported in Sahin et al. (2015).

(h): Concentration in mg/L.

Country of sampling	Year of survey	Matrix	N	No of positives	GTX I (mg/ kg)	GTX II (mg/ kg)	GTX III (mg/ kg)	GTX total ^(a) (mg/ kg)	GTX average ^(b) (mg/kg)	Remarks	Reference
Netherlands	2010	Nepalese wild honey	1	1	n.a.	n.a.	30	30	n.a.		Mol et al., 2011
		(c)			2.7	< LOQ	13	16 (111) ^(d)	n.a.	Several other GTXs tentatively identified and quantified, including rhodojaponin III (60 mg/kg)	Mulder and de Vries, 2022 under Documentation as provided to EFSA
Germany	2010	Black Sea region honey	1	1	58 ^(d)	2.0 ^(e)	54	114 (358) ^(d)	n.a.	12 other GTXs tentatively identified and quantified	These et al., 2015
Turkey	n.i.	Turkish `mad honey'	4	4	0.45–9.9	n.a.	3.0–17	3.1–27	13		Aygun et al., 2015
Turkey	2013–2014	Turkish `mad honey'	25	25	0.17–20	n.a.	< LOD- 76	n.a.	36		Aygun et al., 2018
France	n.i.	Nepalese wild honey	2	2	26–28	n.a.	n.a.	26–28	27		Nassibou et al., 2020

Table 15: Summary results of analytical reports on GTXs in honey from intoxication cases

n.i.: not indicated; n.a.: not analysed/not applicable.

(a): Calculated concentration range for the sum of GTX I, II and III concentrations per sample.

(b): Average calculated from the GTX total concentration divided by the number of samples.

(c): The original sample from 2010 was reanalysed in 2022.

(d): Semi-quantitative sum of all (tentatively) identified grayananes.

(e): Semi-quantitative result.

3.2.2. Processing of honey

No studies were identified that have investigated the effect of processing (e.g. heat) of raw honey on GTX levels.

Storage of honey for up to 6 months at room temperature had only limited effects on the concentration of GTX analogues. The concentration of GTX I and III increased slightly in one sample but remained constant in two other samples that were analysed (Kurtoglu et al., 2014). Kaplan et al. (2014) reported GTX I and III to be stable in honey when stored at room temperature for 12 months.

Grayanotoxins were found to be relatively resistant to biodegradation when *R. ponticum* leaves were subjected to pilot-scale composting for up to 3 months (Hough et al., 2010). The concentration of GTX III remained constant during the duration of the study, but that of GTX I declined by approximately a factor 2.

3.3. Dietary exposure assessment for humans

In the absence of representative occurrence data for the sum of the GTX I and III and consumption data from Ericaceae honey, dietary exposure was estimated based on specific concentrations for GTX I and III reflecting the concentrations measured in 'certain honeys' (see Section 3.4).

In relation to other products based on *Rhododendron* spp. (e.g. dietary supplements, herbal teas originating from *Rhododendron* species and processed foods (wine) from third countries), no information was available regarding GTX occurrence in similar foods in the EU, nor on their consumption by European consumers. A search on the GNPD database²³ did not produce any results for food products having *Rhododendron* as ingredient in the EU. Such products seem to be niche products in the EU with unknown relevance for the internal market and were not considered in the current assessment.

3.4. Risk characterisation

In order to provide information that might be useful to all stakeholders including risk managers, two approaches were followed by the Panel to characterise the risk from consumption of 'certain honey': one assessing the acute effects of grayananes, and a second one addressing the risks of the chronic effects.

As no chronic toxicity or carcinogenicity studies could be identified for GTXs, no RP for chronic toxicity could be identified.

3.4.1. Risk characterisation of acute effects

The risk characterisation was based on acute consumption of honey for toddlers (the age group with the highest P95 acute consumption) and for adults. The Panel decided to not include other populations in the risk characterisation for the following reasons: Children and adolescents had acute consumption values within the range of those of toddlers; elderly and very elderly age groups had acute consumption values within the range of adults. For infants and the specific population groups such as pregnant and lactating women and vegetarians due to the limited number of surveys (less than 3) available with at least 60 subject days of honey consumption to ensure reliable statistics for the 95th percentile of the consumption (EFSA 2011), the risk could not be characterised separately.

3.4.1.1. Margin of exposure approach

The Panel considered it relevant to proceed with a margin of exposure (MOE) estimate for the sum of GTX I and III in order to characterise the risk from acute consumption of 'certain honey'. In the absence of representative occurrence data for the sum of GTX I and III and consumption data from Ericaceae honey, the Panel estimated MOEs based on the following data and assumptions:

²³ https://www.gnpd.com/sinatra/gnpd/search/#

- The RP for acute effects of 15.3 μ g/kg bw for the sum of GTX I and III;
- three concentrations of 0.1 mg/kg,²⁴ 1.0 mg/kg²⁵ and 10 mg/kg,²⁵ selected for illustrative purposes for the sum of GTX I and III;
- the maximum P95 for acute consumption of honey per kg bw among EU surveys and population subgroups.

The results from the MOE calculations are reported in Table 16.

Table 16:MOE estimates (rounded) for acute effects based on the selected concentrations and the
RP of 15.3 μ g/kg bw for the sum of GTX I and III

	Maximum of P95 consumption ^(a)	Exposure an using a sel concentrat 10 mg/	d MOE ected ion of kg	Exposure an using a sel concentrat 1.0 mg/	d MOE ected ion of kg	Exposure and MOE using a selected concentration of 0.1 mg/kg		
	(g/kg bw)	Exposure (µg/kg bw)	MOE	Exposure (µg/kg bw)	MOE	Exposure (µg/kg bw)	MOE	
Toddlers	3.00	30	0.5	3.0	5	0.3	50	
Adults	1.22	12	1.3	1.2	13	0.12	130	

(a): For the estimation of the 95th percentile, surveys with less than 60 observations were excluded since results may not be statistically robust (EFSA 2011).

These MOE calculations suggest that consumption of 'certain honeys' containing GTX I and/or III within the range of concentrations indicated in Table 16 raises a health concern for acute toxicity. The Panel is aware that 'mad' honeys in Tables 9 and 15 are very unlikely to be consumed by toddlers or other children. Furthermore, consumption by adults of 1.22 g honey/kg bw is at the top of the range of advised portion sizes for 'mad' honey.

3.4.2. Risk characterisation for genotoxic/chronic effects

Positive evidence of induction of chromosomal damage in an *in vivo* micronucleus (MN) test in male mice indicates genotoxicity of GTX III and honey containing GTX I and III (see Section 3.1.2.3). In the same honey sample, the analysis of chromosomal aberrations confirmed this result. There is no indication whether this is via a direct or indirect mode of action.

Due to the lack of information on the underlying mode of action of genotoxicity and the lack of data on chronic toxicity and carcinogenicity, the Panel is unable to assess the risk related to chronic/ repeated exposure.

3.4.3. Non-dietary sources of exposure

No non-dietary sources of exposure were identified.

3.5. Estimation of the highest concentrations of the sum of GTX I and III in 'certain honey' below which no acute effects are expected

The Panel estimated the *highest grayanane concentration(s)* below which no acute effects in heart rate and blood pressure would be expected following consumption of 'certain honey' (See Section 3.1.7). These estimates for the sum of GTX I and III in the honey are based on the RP of 15.3 μ g/kg bw and an MOE higher than 100.

The Panel is aware that this methodology does not consider the impact of other co-occurring grayananes in 'certain honey' which cannot be analysed presently. When analytical standards for other grayananes become available, it may be possible to refine the assessment and include other substances.

The backward calculation for the *highest concentration* of the sum of GTX I and III in 'certain honey' was performed based on the following consumption scenarios:

18314732, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by Cochranettalia, Wiley Online Library on [02/03/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons

License

²⁴ Based on the highest value for only GTX I for a honey sample from the EU (Lucatello et al., 2022). This result does not include the contribution of GTX III and other grayananes in the honey sample.

²⁵ Levels of GTX I and/or III in 'certain honeys' (see Tables 9, 14 and 15).

3.5.1. Scenario A: Consumption of *Rhododendron* honey as 'normal' honey

Since acute intoxications with *Rhododendron* honeys have been reported from consumption of 'certain honey' as 'normal honey' (Kerkvliet, 1981; Nassibou et al., 2020), the Panel considered it appropriate to build a scenario where 'normal honey' consumption from the EFSA Comprehensive database was used (details in Section 2). In this scenario, it is assumed that *Rhododendron* honey is intentionally or accidentally consumed as normal food and not because of its use as traditional folk medicine.

Acute consumption statistics of honey per country and age class (consumers only), including specific population groups, are reported in Annex C.2 in g/kg bw per day and in g per day. Annex C.3 shows the range of consumption statistics across surveys per each population group.

Results for toddlers and adults are presented in Table 17. Under this scenario, the concentration for toddlers was estimated to be 0.05 mg GTX I and III/kg honey and that for adults to be 0.12 mg GTX I and III/kg honey.

Table 17:	Calculated highest concentrations of GTX I and III in honey below which no acute effects
	are expected at mean and P95 consumption of honey

	No of surveys	Honey consumption ^(a) (g/kg bw per day)	Highest concentration in honey (mg/kg)	
Results for mean consumption				
Toddlers	16	1.35	0.11	
Adults	24	0.44	0.34	
Results for consumption at P95				
Toddlers	5	3.00	0.05	
Adults	22	1.22	0.12	

(a): Maximum mean and P95 acute consumption across dietary surveys (consumers only).

3.5.2. Scenario B: Consumption of *Rhododendron* honey as traditional folk medicine in adults

In certain regions (e.g. Turkey and Nepal), *Rhododendron* honey rich in grayananes can be used as a traditional folk medicinal product. Human intoxications due to consumption of such honey are often associated with honeys originating from these two regions (see Section 3.1.3.1). The consumption of 'mad honey' in Europe is not recorded in the EFSA Consumption Database and, as suggested by acute intoxication statistics in the EU, it is likely a sporadic practice in Europe (see Section 3.1.3.1).

Portions of *Rhododendron* honey were derived from labels of 'mad honey' products sold on the internet and originating from Nepal or Turkey (as described in Section 2.3). The recommended portions on the labels ranged from a minimum portion of 8 g (1 teaspoon) up to 66 g (equal to 3 tablespoons) of honey. Results are reported in Table 18. Considering the highest recommended portion of 66 g of 'mad honey', this would result in a *highest concentration* of 0.16 mg GTX I and III/kg honey for the adult population.

Table 18:	Calculated highest concentration of GTX I and III in honey below which no acute effects
	are expected based on recommended portions ^(a) on product labelling

Portion ^(a)	Honey consumption ^(b) (g/kg bw per day)	Highest concentration in honey (mg/kg)	
8 g (1 teaspoon)	0.11	1.31	
16 g (2 teaspoons)	0.23	0.65	
66 g (3 tablespoons)	0.94	0.16	

(a): Portion values are indicative and subject to uncertainty, see Section 2.5. (b): estimated for a standard bw of 70 kg (EFSA, 2012).

(b): estimated for a standard bw of 70 kg (EFSA, 2012)

3.6. Uncertainty analysis

The purpose of the uncertainty analysis was to identify and quantify sources of uncertainty affecting the risk assessment and combine them to assess the overall certainty of the final conclusions, as recommended in EFSA's guidance on uncertainty analysis (EFSA, 2018a).

In a first step, sources of uncertainties related to hazard identification and characterisation of grayananes and consumption of honey containing grayananes in the EU were listed and discussed. A complete list is presented in Annex D. The most important uncertainties are listed in order of importance in Appendix E, together with 'low' and 'high' scenarios describing the Panel's qualitative evaluation of their potential impact on the assessment.

Hazard identification for genotoxic effects followed EFSA's guidance for genotoxicity assessment (EFSA Scientific Committee, 2011, 2017b), which can be considered as a standardised procedure for the purposes of uncertainty analysis (Section 3 in EFSA Scientific Committee, 2018a). As there was a positive result from a valid *in vivo* MN study with GTX III and *Rhododendron* honey, the Panel concluded that GTX III and honey containing GTX I and III should be considered as *in vivo* genotoxins and, therefore, an HBGV cannot be established. The Panel did not identify any non-standard uncertainties affecting these conclusions, so no further uncertainty analysis was required for them (Section 3.1 in EFSA Scientific Committee, 2018a).

The assessment of acute toxicity and risk were affected by major uncertainties. The Panel's analysis of these is described in detail in Appendix D. In summary, the Panel quantified by expert judgement the impact of the identified uncertainties on the RP for acute toxicity for the sum of GTX I and III, the RPFs of other grayananes and the P95 of honey consumption for EU toddlers and adults. These judgements were elicited by semi-formal structured methods of Expert Knowledge Elicitation (semi-formal EKE, Annex B.8 of EFSA (2018b)). The quantified uncertainties affecting the RP for acute toxicity and P95 consumption were then combined by probability bounds analysis (section 14.1 of EFSA, 2018a) to quantify the uncertainty of the MOEs calculated by the Panel for the exposure at selected concentrations for the sum of GTX I and GTX III and acute consumption from the EFSA database. The Panel also calculated the combined uncertainty associated with the *calculated highest concentrations*, based on the RP for acute toxicity and data on acute consumption (i) from the EFSA database, and (ii) from consumption of 'certain honey' as folk medicine.

Uncertainties affecting the RP for acute toxicity for the sum of GTX I and GTX III were quantified by eliciting expert judgements of the probability that this RP would be higher than the assessed value of 15.3 μ g/kg bw if the uncertainties were resolved. This provides a measure of the conservativism of the assessed value. The judgements were made by four members of the WG with specialist expertise on toxicology and for the interpretation of reported cases of human intoxication. The judgements of the four experts were all in the range 70–90% probability. After discussion, the experts agreed on a consensus of 85% probability, considering the balance between the most important sources of uncertainty: the potential for a lower value of the RP for acute toxicity in females than males, higher values for oral compared to i.p. exposure, and the bioavailability of GTX I and III when administered via honey instead of as pure substances. On this basis, the Panel is 85% certain²⁶ that the RP for acute toxicity of 15 μ g/kg bw for the sum of GTX I and III is protective (i.e. would not be lower if the uncertainties were resolved).

RPFs for the acute toxicity of grayananes classified into four groups are presented in Table 11 (Section 3.1.5). The combined impact of uncertainties affecting these was quantified by eliciting expert judgements of the probability that the RPF for each group is conservative (i.e. would not be higher if the uncertainties were resolved) for all grayananes in Table 1 of the Opinion that would fall in that group. Expert judgements were made by four experts: three toxicologists in the WG and a fourth WG member with expertise on assessment of relative potencies. Considering the range of judgements of the four experts, the Panel is 70–80% certain for Group 1A, 90–95% for Group 1B and 80–90% certain for Groups 2 and 3 (Table 11) that the RPFs are conservative. The difference in certainty between groups reflects the different amounts of data on which their RPFs are based.

Maximum values for the P95 of acute consumption of honey for toddlers and adults who are consumers of honey are presented in Table 17. The value for each age group is the maximum of P95 values from surveys in the EFSA Comprehensive Database from a subset of EU countries. Taking the maximum of the available values is intended to provide a conservative estimate of the P95 for each age group in the whole of the EU. The combined impact of uncertainties affecting estimation of the P95 for each age group in the whole of the EU was quantified by eliciting expert judgements from three members of the WG with expertise in exposure assessment. Considering the range of

²⁶ The impact of uncertainties on the conclusion of each part of the risk assessment was quantified using % probabilities assessed by expert judgement, following EFSA's guidance on uncertainty analysis (EFSA, 2018a). The resulting probabilities are reported as % certainty for the more probable outcome for each conclusion, following EFSA's guidance on communication of uncertainty (EFSA, 2019).

judgements of the four experts for each of the two age groups, the Panel is at least 90% certain that the 95th percentile of acute consumption of honey by EU consumers of honey, is not higher than the assessed value, i.e. 3 g/kg bw per day for toddlers and 1.22 g/kg bw per day for adults. The Panel did not include estimates of P95 consumption for other age groups in Table 17. However, based on the data that were available for other age groups (Annex C), the experts judged with 90–95% probability that the P95 of acute honey consumption for infants, adolescents and other children in the EU would be lower than 3 g/kg bw (the value for toddlers), and 95–99% probability for the elderly and very elderly. These judgements took account of all identified sources of uncertainty but excluded consumption of honey as folk medicine.

The MOEs in Table 16 were calculated as the ratio of the RP of 15.3 μ g/kg bw to the estimated exposure, based on the selected concentrations and the P95 of acute honey consumption for each age group from dietary surveys (see Section 3.4.1.1). Therefore, the uncertainty of the RP must be combined with the uncertainty of the P95 consumption for the EU population in each age group to assess the uncertainty of each MOE. The selected concentrations used by the Panel when calculating the MOEs are fixed values (not subject to uncertainty), so the calculated MOEs are conditional on them. The combination of the uncertainties relating to the RP and acute consumption was performed by probability bounds analysis (Section 14.1 and Annex B.13 of EFSA (2018b)) and is described in Appendix D.

Based on this assessment of the combined uncertainties, the Panel is 75% or more certain for toddlers and for adults that the MOEs shown in Table 16 are conservative, i.e. would not be lower if the uncertainties affecting the RP and P95 consumption estimates were resolved. The Panel noted that this is a minimum probability (see Appendix D).

The Panel did not calculate MOEs for age groups other than toddlers and adults. However, substituting the experts' judgements about P95 consumption for other age groups into the probability bounds analysis conducted for toddlers and adults, the Panel was 75% or more certain that the MOE for toddlers is conservative (i.e. would not be lower) for infants, adolescents and other children, and 80% or more certain that it is also conservative for the elderly and very elderly.

The Panel calculated *highest concentrations* of GTX I and III in 'certain honey' at which there is no concern *for acute toxicity* based on the P95 of acute honey consumption (Table 17). These were calculated as the ratio of the RP for acute toxicity to the estimated P95 of acute honey consumption for each age group, based on dietary surveys and taking into account that an MOE above 100 implies a low health concern (see Section 3.5.1). Therefore, the uncertainty of the RP for acute toxicity must be combined with the uncertainty of the P95 consumption to assess the uncertainty of these *highest concentrations*.

The combination was performed by probability bounds analysis, as was done for the MOE. Based on the result, the Panel is 75% or more certain for toddlers and for adults that the *highest concentrations* shown in Table 17 are protective, i.e. would not lead to a concern for *acute toxicity* at the P95 of normal honey consumption by that age group in the EU. The Panel noted again that the result of the probability bounds analysis is a minimum probability.

The Panel did not calculate *highest concentrations* for age groups other than toddlers and adults. However, substituting the experts' judgements about P95 consumption for other age groups into the probability bounds analysis conducted for toddlers and adults, the Panel was 75% or more certain that the *highest concentration* of 0.05 mg/kg for toddlers is also protective for infants, adolescents and other children, and at least 80% certain that it is also protective for the elderly and very elderly.

The Panel also calculated *highest concentrations* for indicative portion sizes of 'certain honey' used as folk medicine (Table 18). There were insufficient data for the Panel to quantify uncertainties regarding variation of recommended portion sizes on product labels or the actual portion sizes consumed by users of those products. Instead, the Panel quantified the uncertainty of the *highest concentrations* for folk medicine use conditional on the specific portion sizes considered. Therefore, the specified portion size was considered as fixed and the Panel's certainty that the corresponding *highest concentration* is protective is determined solely by the Panel's certainty of 85% that the RP for acute toxicity is conservative. Consequently, the same level of certainty applies to each of the indicative portion sizes shown in Table 18. For example, for a portion size of 66 g, the Panel is 85% certain that concentrations up to the calculated *highest concentration* of 0.16 mg/kg would not lead to a concern *for acute toxicity*, taking into account the uncertainty of that level. For portion sizes larger than 66 g, proportionately lower *highest concentrations* would be needed.

The Panel noted that the calculated MOEs and *highest concentrations* are based on acute toxicity of GTX I and III and do not consider the presence of other grayananes in `certain honey'.

4. Conclusions

4.1. General

Grayananes are polyhydroxylated diterpenes. There are approximately 180 natural grayananes that have been isolated from Ericaceae plants and identified of which approximately 20 grayananes have been found in or associated with 'certain honeys'.²⁷ The majority of the available analytical and toxicological data concern GTX I and III.

Certain Ericaceae honeys, mostly from *Rhododendron* species from the Turkish Black Sea region and Nepal are often described as 'mad honeys' or 'toxic honeys' and are associated with the presence of grayananes and acute intoxication in humans. Non-*Rhododendron* honey from *Agarista salicifolia* is associated with grayanane intoxications in humans on the French island of La Réunion.

For this Opinion, the most important *Rhododendron* species used for honey production are *R. ponticum*, *R. luteum* and *R. ferrugineum*.

The first two species are particularly abundant in Turkey and are associated with intoxication in humans due to honey consumption. The latter species occurs in the Alps and Pyrenees and the honey produced is not reported to induce intoxication.

The yearly production volume of Ericaceae honey including *Rhododendron* honey in Europe is unknown but estimated to be small. The import volumes of these honeys from Turkey and Nepal in the EU are also limited.

4.2. Hazard identification and characterisation

4.2.1. Toxicokinetics

- Few toxicokinetic studies are available and data are limited.
- Oral toxicokinetic studies with GTX I and III in experimental animals do not allow conclusions on the bioavailability of GTX I and III following exposure to 'certain honey'.
- In case reports on intoxications related to consumption of 'mad honey', GTX I and GTX III were detected in human blood and urine, showing that they are absorbed from the GI-tract.
- No information on the biotransformation of grayananes in mammals is available to EFSA.

4.2.2. Single-dose toxicity studies in experimental animals

- Few oral studies with grayananes or preparations containing grayananes (e.g. 'mad honey') in experimental animals have been undertaken. Most toxicodynamic studies with grayananes use i.p. or i.v. injection and are LD₅₀ studies, mainly in male rodents. Cardiovascular effects and depression of respiration are the main causes of death.
- The acute effects arising from the animal models most relevant for this opinion are impairment of the nervous system, bradycardia and reduction of blood pressure.
- Studies indicating differences in blood chemistry parameters were inconsistent and not supported by histological changes.
- Characterisation of the acute toxicity of grayananes other than GTX I and GTX III is limited.
- The lack of appropriate studies did not allow to extrapolate data from i.p. injection to an oral dose.
- Based on their structure–activity relationships, grayananes can be divided into four groups. The two most potent groups showed three structural alerts, and the number of alerts was inversely proportional to the LD₅₀ values. Relative potency factors (RPFs) were derived for each group: 2 for Group 1A, 1 for Group 1B, 0.1 for Group 2 and < 0.01 (negligible potency) for Group 3. An RPF of 1 was assigned to GTX I and III which have similar toxicity.
- Grayananes with high potency share three structural features: They all contain a 3β -hydroxy or a 2β , 3β -epoxy group, a 6β -hydroxy and a 10β -methyl group. When one of these structural features is modified or missing, toxicity is greatly reduced.

²⁷ The term 'certain honey' shall refer to honey and honey products originating from Ericaceae family plants and that contain grayananes.

4.2.3. Repeated-dose toxicity studies in experimental animals

- Only subacute and subchronic studies in rodents with grayananes or 'mad honey' were available. Changes were mostly observed in biochemical parameters and were considered not toxicologically relevant.
- One study suggests potential effects on the testes but the available data are limited.
- No chronic/carcinogenicity studies have been carried out.

4.2.4. Genotoxicity

• There is evidence of chromosomal damage for GTX III and honey containing GTX I and III from an *in vivo* MN test in mice. In the same honey sample, the analysis of chromosomal aberrations confirmed this result. There is no indication whether this is via a direct or indirect mode of action.

4.2.5. Observations in humans

- Acute symptoms from human intoxications with honey containing grayananes affect the muscles and the nervous and cardiovascular systems with bradycardia and hypotension being most prominent. Other common symptoms are dizziness, nausea, paraesthesia, blurred vision, weakness and change of consciousness. Higher doses may lead to complete atrioventricular block, convulsions, mental confusion, agitation, syncope and respiratory depression.
- Severe cases of poisoning are in general effectively treated in emergency medical care with atropine administered intravenously, fatal cases not having been reported in the last decades.
- The doses associated with acute poisoning cases on a single occasion observed with *Rhododendron* honeys from the Turkish Black Sea region range from 0.6 to 17 μ g/kg bw or 3.9 to 24 μ g/kg bw for GTX I and GTX III, respectively, and from 4.8 to 40 μ g/kg bw for the sum of GTX I and III.
- From cases of acute intoxications following the ingestion of a Nepalese *Rhododendron* honey, where GTX I and III were not detected, it can be concluded that other grayananes present in *Rhododendron* species cause adverse effects on the muscles and the nervous and cardiovascular systems.

4.2.6. Mode of action

- Grayananes bind specifically to the voltage operated sodium channels preventing their inactivation and resulting in long-lasting depolarisation of excitable cells like neurons, muscle cells and cardiomyocytes.
- Reversal with atropine, a muscarinic antagonist, in human intoxication is consistent with a facilitated release of acetylcholine due to overactivation of sodium channels on cholinergic neurons (e.g. the vagus nerve).

No mechanistic data on genotoxicity and repeated dose toxicity were identified.

4.2.7. Critical effects and dose–response analysis

- The acute critical effects in experimental animals are on heart rate and blood pressure which are consistent with those observed in humans.
- An RP for GTX I, GTX III or other grayananes could not be identified from the data available for acute intoxications in humans.
- The RP associated with the acute critical effects was derived from BMD analysis of the reduction of heart rate after i.p. injection of GTX III in rats. A BMDL₁₀ of 15.3 μ g/kg bw was determined and considered as the RP for the sum of GTX I and III, considering equal potency of both GTXs.
- The RP for the sum of GTX I and III may be applied to other grayananes, such as GTX II, by making use of their RPFs.
- It was not possible to take into account potential differences in grayanane bioavailability after oral and i.p. exposure to 'certain honey' or the pure substance(s) because of the limited toxicokinetic studies.
- No RP could be identified from repeated dose toxicity studies on grayananes.
- A health-based guidance value (HBGV) could not be derived for grayananes due to the indicated genotoxicity *in vivo*.

4.2.8. Occurrence

- Occurrence data on grayananes are only available from literature and limited to GTX I and/or GTX III. Most of the data are on *Rhododendron* honey from Turkish or Nepalese origin. Levels of GTX I and/or GTX III in these honeys range from 0.1 to 54 mg/kg.
- One study is available with quantifiable levels of GTX I in *Rhododendron* honey of EU origin (Italian Alps), showing a highest level for GTX I of 0.1 mg/kg. GTX III was also detected in some samples but not quantified.
- Grayananes other than GTX I and III have been detected in 'certain honey' originating from the Ericaceae family using modern methods of analysis.
- The lack of reference standards and the lack of analytical methods with sensitivity as low as 0.01 mg GTX per kg honey were identified as important factors hampering the collection of representative occurrence data.
- For other food categories, only limited data are available.
- No conclusions on the effect of processing can be drawn as no studies were identified. The limited available information suggests that GTX I and III are stable in honey when stored at room temperature for 12 months.

4.2.9. Dietary exposure assessment

In the absence of representative occurrence data for the sum of the GTX I and III and consumption data from Ericaceae honey, dietary exposure was estimated based on specific concentrations for GTX I and III in 'certain honey'. These reflect the highest measured concentration for GTX I in a honey sample of EU origin (0.1 mg/kg) and concentrations measured for the sum of GTX I and III in 'certain honeys' (1 and 10 mg/kg).

4.2.10. Risk characterisation

Risk characterisation for acute effects

- In the absence of an HBGV, the Panel decided to apply an MOE approach.
- MOEs were estimated from the RP of 15.3 μ g/kg bw for GTX I and III, and three exposure estimates based on selected concentrations for the sum of GTX I and III reflecting the concentrations measured in `certain honey' and the maximum P95 for acute consumption of honey per kg bw among EU surveys and population subgroups.
- The estimated MOEs raised health concerns for acute toxicity.

Risk characterisation for genotoxic/chronic effects

• Induction of chromosomal damage as shown in an *in vivo* micronucleus (MN) test in male mice indicates genotoxicity for GTX III and honey containing GTX I and III. Due to the lack of information on the underlying mode of action of genotoxicity and the lack of data on chronic toxicity and carcinogenicity, the Panel is unable to assess the risk related to chronic/repeated exposure.

4.2.11. Derivation of the highest levels in honey below which no acute effects are expected

- *Highest concentrations* for the sum of GTX I and III in honey below which no acute effects are expected were calculated based on the RP and an MOE of at least 100.
- When *Rhododendron* honey is consumed as 'normal' honey, a *highest concentration* for the sum of GTX I and III was estimated for toddlers being 0.05 mg GTX I and III/kg honey, based on the P95 acute consumption of honey from the EFSA database. For other age groups, the *highest concentrations* in 'normal' honey were higher.
- When *Rhododendron* honey is consumed by adults as traditional folk medicine, the *highest concentration* was estimated to be 0.16 mg GTX I and III/kg honey for consumption of 3 tablespoons (66 g) of honey. For larger portion sizes, the *highest concentration* would be lower.
- The estimated MOE and *highest concentrations* do not consider the impact of other cooccurring grayananes in 'certain honey'.

4.2.12. Uncertainty analysis

- The risk assessment on genotoxicity was not subject to any non-standard uncertainties, so no further uncertainty analysis was required for this.
- In the assessment of acute toxicity, taking account of all identified sources of uncertainty:
 - The Panel is 85% certain²⁸ that the RP for the sum of GTX I and III is protective (i.e. would not be lower if the uncertainties were resolved).
 - The Panel's certainty that the RPFs presented in the opinion are conservative is 70–80% for Group 1A, 90–95% for Group 1B and 80–90% for Groups 2 and 3.

When Rhododendron honey is consumed as 'normal' honey,

- The Panel is at least 90% certain that the 95th percentile of acute consumption of honey by EU consumers of honey, is not higher than 3 g/kg bw per day for toddlers and 1.22 g/kg bw per day for adults (excluding consumption as folk medicine).
- The Panel is 75% or more certain for toddlers and adults that the MOEs presented in the opinion are conservative, i.e. would not be lower if the uncertainties affecting the RP and P95 consumption estimates were resolved. The same conclusion applies to all other age groups.
- The Panel is 75% or more certain that a highest concentration of 0.05 mg/kg below which no acute effects are expected is protective for toddlers, i.e. will not lead to a concern for acute toxicity at the 95th percentile of acute consumption. The same conclusion applies to all other age groups.

When Rhododendron honey is consumed as folk medicine,

- there were insufficient data for the Panel to quantify uncertainties regarding the portion sizes consumed. Instead, uncertainty was quantified for a selection of indicative portion sizes. For a portion size of 66 g, the Panel is 85% certain that a *concentration* of 0.16 mg/ kg or lower would not lead to a *concern for acute toxicity*. For larger portion sizes, proportionally lower concentrations would apply.
- The Panel noted that the calculated MOEs and *highest concentrations* are based on acute toxicity of GTX I and III and do not consider the presence of other grayananes in 'certain honey'. In addition, they may not be protective for genotoxic effects.

5. Recommendations

The following needs have been identified to improve the risk assessment for humans and reduce the uncertainties:

- Analytical standards are required for grayananes other than GTX I and III to enable accurate measurement of these substances in 'certain honey' and other foods, in particular for rhodojaponin III.
- Validated analytical methods should be developed for the quantification of total grayananes in 'certain honey' with LOQs per grayanane at 0.01 mg/kg or lower.
- There is a need for occurrence data on grayananes in honey and commodities thereof from Ericaceae species. Quantitative data should be collected for at least GTX I and GTX III.
- Integration of the monitoring of grayananes in *Rhododendron* honeys on the EU market in the national control activities is recommended, especially from specific regions of production (e.g. Alps and Pyrenees).
- Investigations of the market share of *Rhododendron* honeys in the EU market are needed.
- Mechanistic studies are required to understand the mode of action of genotoxicity.
- The short- and long-term effects of orally administered grayananes on the reproductive system need to be investigated with single and repeated-dose toxicity studies across a broader dose range (> $0.8 \mu g/kg$ bw).
- Experimental studies via the oral route on the effects of acute and chronic exposure/ carcinogenicity to grayananes are required.
- Observations/surveys in humans on the acute effects of grayananes in the low-dose range and on the NOAEL are desirable.

²⁸ i.e. judges with 85% probability (EFSA, 2019).

Documentation as provided to EFSA

- 1) France, 2022. Description of poisoning cases related to consumption of honey contaminated with grayanatoxins registered by the French Poison Control Centres. Information submitted from ANSES to EFSA on 1 March 2022.
- 2) Germany, 2022. Case reports on Poisonings assigned to grayanotoxins from 1990 to 2021, retrieved for BfR National Case Database of Poisonings. Information submitted from BfR to EFSA on 2 February 2022.
- Mulder P.P.J. and de Vries V.E, 2022. Preliminary results on the presence of grayanotoxins in 16 Ericaceae honey samples by LC-Orbitrap-MS analysis. Information provided to EFSA, 26 August 2022.
- 4) Stakeholder and focal points inputs on grayanotoxins, 2022. Information submitted to EFSA in reply to the call for stakeholder inputs in relation to the EFSA mandate on grayanotoxins in 'certain honey', from 17 January-28 February 2022.
- 5) Stout J. March 2022. Data submitted by Dr Jane Stout about the characteristics of *Rhododendron* species and honey and used in Section 1.2.2 Sources.
- 6) Turkey, 2022. Grayanane-containing Ericaceae and grayanane-containing honey. Information submitted to EFSA from the Turkish Ministry of Agriculture and Forestry on 11 February 2022.

References

- Ahn SY, Kim S and Cho H, 2022. Determination of Grayanotoxin I and Grayanotoxin III in mad honey from Nepal using liquid chromatography-tandem mass spectrometry. Analytical Science and Technology, 35, 82–91.
- Akgün N, Çelik ÖF and Kelebekli L, 2021. Physicochemical properties, total phenolic content, and antioxidant activity of chestnut, rhododendron, acacia and multifloral honey. Journal of Food Measurement and Characterization, 15, 3501–3508.
- Akinci S, Arslan U, Karakurt K and Çengel A, 2008. An unusual presentation of mad honey poisoning: acute myocardial infarction. International Journal of Cardiology, 129, e56–e58.
- Aleguas A, Vitale C, Sheroff A and Burns-Ewald M, 2008. Grayanotoxin poisoning from Pieris japonica. In Clinical Toxicology, (Vol. 46, No. 5, pp. 410-410). New York, NY, USA: Informa Healthcare.
- Aliyev F, Türkoğlu C and Çeliker C, 2009a. Nodal Rhythm and Ventricular Parasystole: an Unusual Electrocardiographic Presentation of Mad Honey Poisoning. Clinical Cardiology: An International Indexed and Peer-Reviewed Journal for Advances in the Treatment of Cardiovascular Disease, 32(11), E52–E54.
- Aliyev F, Türkoğlu C, Çeliker C, Firatli I, Alici G and Uzunhasan I, 2009b. Chronic mad honey intoxication syndrome: a new form of an old disease? Europace, 11(7), 954–956.
- Alkan S, Akgün M, Ertürk Ö, Çol Ayvaz M and Başkan C, 2020. Properties of Honey and Pollen Samples Obtained from Different Rhododendron Species Collected from Black Sea Region of Turkey. Journal of Apicultural Science, 64, 321–334.
- Al Shoyaib A, Archie SR and Karamyan VT, 2020. Intraperitoneal route of drug administration: should it be used in experimental animal studies? Pharmaceutical Research, 37, 1–7.
- Ascioglu M, Ozesmi C, Dogan P and Ozturk F, 1996. Effects of chronic grayanotoxin-I administration on hepatic and renal functions in rats. Tohoku Journal of Experimental Medicine, 179, 47–53. https://doi.org/10.1620/ tjem.179.47
- Ascioglu M, Ozesmi C, Dogan P and Ozturk F, 2000. Effects of acute grayanotoxin I administration on hepatic and renal functions in rats. Turkish Journal of Medical Science, 30, 23–27.
- Aygun A, Gunduz A, Turedi S, Turkmen S, Karaca Y, Ayaz FA, Ahn SY and Kim S, 2015. Examination using LC-MS/ MS determination of grayanotoxin levels in blood, urine, and honey consumed by patients presenting to the emergency department with mad honey intoxication and relations with clinical data: a preliminary study. Annals of Saudi Medicine, 35, 161–164.
- Aygun A, Vuran HS, Aksut N, Karaca Y, Gunduz A and Turedi S, 2016. Mad honey poisoning-related hypothermia: a case series. The Journal of Emergency Medicine, 50, 51–54.
- Aygun A, Sahin A, Karaca Y, Turkmen S, Turedi S, Ahn SY, Kim S and Gunduz A, 2018. Grayanotoxin levels in blood, urine and honey and their association with clinical status in patients with mad honey intoxication. Turkish Journal of Emergency Medicine, 18, 29–33.
- Bayram NA, Keles T, Durmaz T, Dogan S and Bozkurt E, 2012. A rare cause of atrial fibrillation: mad honey intoxication. The Journal of Emergency Medicine, 43, e389–e391.
- BfR (Bundesinstitut fűr Risikobewertung), 2010a. BfR Opinion No 043/2010 of 3 September 2010. Cases of poisoning through grayanotoxins in *Rhododendron* honey originating from the Turkish Black Sea Region (Abstract in English). Available online: https://www.google.com/search?client=firefox-b-d&q=Cases+ of+poisoning+through+grayanotoxins+in+rhododendron+honey+originating+from+the+Turkish+Black+Sea+ Region

- BfR (Bundesinstitut fűr Risikobewertung), 2010b. Stellungnahme Nr. 043/2010 des BfR vom 3. September 2010. Vergiftungsfälle durch Grayanotoxine in Rhododendron-Honigen aus der türkischen Schwarzmeerregion. Biberoglu S, Biberoglu K and Komsuoglu B, 1988. Mad honey. JAMA, 259, 1943.
- Biberoglu K and Komsuoglu B, 1988. Transient Wolff-Parkinson-White syndrome during honey intoxication. Israel Journal of Medical Sciences, 24, 253–254.
- Bilir O, Ersunan G, Yavasi O and Kalkan A, 2016. Mad honey-related intoxication in an infant: a case report. Journal of Emergency Medicine Case Reports, 7, 80–81. https://doi.org/10.5152/jemcr.2016.1476
- Bilir O, Ersunan G, Yavasi O, Kayayurt K, Giakoup B and Bostan M, 2017. How much should we observe patients with mad honey poisoning?
- Booth NL, Kruger CL, Hayes AW and Clemens R, 2012. An innovative approach to the safety evaluation of natural products: Cranberry (Vaccinium macrocarpon Aiton) leaf aqueous extract as a case study. Food and Chemical Toxicology, 50, 3150–3165.
- Bostan M, Bostan H, Kaya AO, Bilir O, Satiroglu O, Kazdal H and Bozkurt E, 2010. Clinical events in mad honey poisoning: a single centre experience. Bulletin of the Environmental Contamination and Toxicology, 84, 19–22. https://doi.org/10.1007/s00128-009-9906-2
- Brackenbury WJ and Isom LL, 2011. Na+ channel β subunits: overachievers of the ion channel family. Frontiers in Pharmacology, 2, 53.
- Bruni I, De Mattia F, Fluch S, Ferrari C, Corazza M, Dinelli E and Labra M, 2016. Genetic introgression of hybrid Rhododendron x intermedium Tausch is habitat mediated: Evidences from south-eastern Alps (Italy). Plant Biosystems An International Journal Dealing with all Aspects of Plant Biology, 150, 449–458.
- Cagli KE, Tufekcioglu O, Sen N, Aras D, Topaloglu S, Basar N and Pehlivan S, 2009. Atrioventricular block induced by mad-honey intoxication: confirmation of diagnosis by pollen analysis. Texas Heart Institute Journal, 36, 342.
- Cakar MA, Can Y, Vatan MB, Demirtas S, Gunduz H and Akdemir R, 2011. Atrial fibrillation induced by mad honey intoxication in a patient with Wolf–Parkinson–White syndrome. Clinical Toxicology, 49, 438–439.
- Cakmak-Arslan G, Haksoy H, Goc-Rasgele P and Kekecoglu M, 2020a. Determination of the dose-dependent toxic effects of mad honey on mouse liver using ATR-FTIR spectroscopy. Spectrochimica Acta Part A: Molecular and Biomolecular Spectroscopy, 228, 117719. https://doi.org/10.1016/j.saa.2019.117719
- Cakmak-Arslan G, Emir S, Goc-Rasgele P and Kekecoglu M, 2020b. Investigation of the toxic effects of rhododendron honey on mouse cardiac muscle tissue lipids at molecular level. Kafkas Universitesi Veteriner Fakultesi Dergisi, 26, 287–294. https://doi.org/10.9775/kvfd.2019.22996
- Catterall WA, 2000. From ionic currents to molecular mechanisms: the structure and function of voltage-gated sodium channels. Neuron, 26, 13–25.
- Cestèle S and Catterall WA, 2000. Molecular mechanisms of neurotoxin action on voltage-gated sodium channels. Biochimie, 82, 883–892.
- Chai B, Li Y and Yu SS, 2020. Three new antinociceptive diterpenoids from the roots of Rhododendron micranthum. Journal of Asian Natural Products Research, 22, 895–904.
- Chen G, Gao Y, Mao HY and Li YS, 1985. Hemodynamic effects of rhomotoxin assessed by impedance cardiography in hypertensive patients. Chinese Medical Journal, 98, 325–328.
- Cho HE, Ahn SY, Kim DW, Woo SH, Park SH, Hwang K, Moon DC and Kim S, 2014. Development of a liquid chromatography–tandem mass spectrometry method for the determination of grayanotoxins in rat blood and its application to toxicokinetic study. Biomedical Chromatography, 28, 1624–1632.
- Choi HL, Park KH, Park JS, Choi HY, Kim H and Kim SM, 2017. Relationship between blood toxin level and clinical features in patients with grayanotoxin poisoning–six clinical cases. Clinical Toxicology, 55, 991–995.
- Cotten MD, Maling HM and Moran NC, 1956. Comparison of the direct and indirect effects of andromedotoxin on the contractile force of the heart. The Journal of Pharmacology and Experimental Therapeutics, 118, 55–62.
- Creveling CR, McNeal ET, McCulloh DH and Daly JW, 1980. Membrane potentials in cell-free preparations from guinea pig cerebral cortex: Effect of depolarizing agents and cyclic nucleotides. Journal of Neurochemistry, 35, 922–932.
- Cucer N and Eroz R, 2010. Investigation of mutagenic effects of grayanotoxins II and III on cultured human lymphocytes. Al Ameen Journal of Medical Sciences, 4, 293–299.
- Dampc A and Luczkiewicz M, 2015. Labrador tea–the aromatic beverage and spice: a review of origin, processing and safety. Journal of the Science of Food and Agriculture, 95, 1577–1583.
- De Sousa IP, Sousa Teixeira MV and Jacometti Cardoso Furtado NA, 2018. An overview of biotransformation and toxicity of diterpenes. Molecules, 23, 1387.
- Demir AAS and Kahveci FO, 2012. An indispensable toxin known for 2500 years: victims of mad honey. Turkish Journal of Medical Sciences, 42, 1499–1504.
- Demir H, Denizbasi A and Onur O, 2011. Mad honey intoxication: a case series of 21 patients. International Scholarly Research Notices, 2011, 526426. https://doi.org/10.5402/2011/526426
- Doğanyiğit Z, Silici S, Demirtaş A, Kaya E and Kaymak E, 2019. Determination of histological, immunohistochemical and biochemical effects of acute and chronic grayanotoxin III administration in different doses in rats. Environmental Science and Pollution Research, 26, 1323–1335.

64

- Doğanyiğit Z, Kaymak EM and Silici Sİ, 2020. The cardiotoxic effects of acute and chronic grayanotoxin-III in rats. Human & Experimental Toxicology, 39, 374–383.
- Dong LC, Zhang XH, Ma J, Luo N, Song W, Li P and Li HJ, 2014. The integrated pharmacokinetics of major rhodojaponins correlates with the cardiotoxicity after oral administration of Rhododendri Mollis Flos extract in rats. Journal of Ethnopharmacology, 157, 69–78.
- Dönmez M and Kaya E, 2020. Separating mad Honey from other honeys with grayanotoxin analysis in LC-MS/MS. International Journal of Traditional and Complementary Medicine Research, 1, 48–53.
- Dusemund B, Soffers AE and Rietjens IM, 2012. Plant-derived contaminants in food. Chemical Contaminants and Residues in Food. Woodhead Publishing. pp. 394–417.
- EFSA (European Food Safety Authority), 2009. Compendium of botanicals that have been reported to contain toxic, addictive, psychotropic or other substances of concern on request of EFSA. EFSA Journal 2009;7(9):281, 100 pp. https://doi.org/10.2903/j.efsa.2009.281
- EFSA (European Food Safety Authority), 2011. Use of the EFSA comprehensive european food consumption database in exposure assessment. EFSA Journal 2011;9(3):2097, 34 pp. https://doi.org/10.2903/j.efsa.2011.2097
- EFSA (European Food Safety Authority), 2012. Compendium of botanicals reported to contain naturally occuring substances of possible concern for human health when used in food and food supplements. EFSA Journal 2012;10(5):2663, 60 pp. https://doi.org/10.2903/j.efsa.2012.2663
- EFSA (European Food Safety Authority), 2015. The food classification and description system FoodEx2 (revision 2). EFSA supporting publication 2015:EN-804, 90 pp. https://doi.org/10.2903/sp.efsa.2015.en-804
- EFSA (European Food Safety Authority), 2017. Technical report on the outcome of the consultation with Member States and EFSA on the basic substance application for honey from Rhododendron for use in plant protection as rodenticide. EFSA supporting publication 2017:EN-1155, 54 pp. https://doi.org/10.2903/sp.efsa.2017.en-1155
- EFSA (European Food Safety Authority), Hart A, Maxim L, Siegrist M, Von Goetz N, da Cruz C, Merten C, Mosbach-Schulz O, Lahaniatis M, Smith A and Hardy A, 2019. Guidance on communication of uncertainty in scientific assessments. EFSA Journal 2019;17(1):5520, 73 pp. https://doi.org/10.2903/j.efsa.2019.5520
- EFSA Scientific Committee, 2011. Scientific Opinion on genotoxicity testing strategies applicable to food and feed safety assessment. EFSA Journal 2011;9(9):2379, 69 pp. https://doi.org/10.2903/j.efsa.2011.2379
- EFSA Scientific Committee, 2012. Guidance on selected default values to be used by the EFSA Scientific Committee, Scientific Panels and Units in the absence of actual measured data. EFSA Journal 2012;10(3):2579, 32 pp. https://doi.org/10.2903/j.efsa.2012.2579
- EFSA Scientific Committee, Hardy A, Benford D, Halldorsson T, Jeger MJ, Knutsen KH, More S, Mortensen A, Naegeli H, Noteborn H, Ockleford C, Ricci A, Rychen G, Silano V, Solecki R, Turck D, Aerts M, Bodin L, Davis A, Edler L, Gundert-Remy U, Sand S, Slob W, Bottex B, Abrahantes JC, Marques DC, Kass G and Schlatter JR, 2017a. Update: Guidance on the use of the benchmark dose approach in risk assessment. EFSA Journal 2017;15(1):4658, 41 pp. https://doi.org/10.2903/j.efsa.2017.4658
- EFSA Scientific Committee, Hardy A, Benford D, Halldorsson T, Jeger M, Knutsen HK, More S, Naegeli H, Noteborn H, Ockleford C, Ricci A, Rychen G, Silano V, Solecki R, Turck D, Younes M, Aquilina G, Crebelli R, Gurtler R, Hirsch-Ernst KI, Mosesso P, Nielsen E, van Benthem J, Carfi M, Georgiadis N, Maurici D, Parra Morte J and Schlatter J, 2017b. Scientific Opinion on the clarification of some aspects related to genotoxicity assessment. EFSA Journal 2017;15(12):5113, 25 pp. https://doi.org/10.2903/j.efsa.2017.5113
- EFSA Scientific Committee, Benford, D, Halldorsson, T, Jeger, MJ, Knutsen, HK, More, S, Naegeli, H, Noteborn, H, Ockleford, C, Ricci, A, Rychen, G, Schlatter, JR, Silano, V, Solecki, R, Turck, D, Younes, M, Craig, P, Hart, A, Von Goetz, N, Koutsoumanis, K, Mortensen, A, Ossendorp, B, Martino, L, Merten, C, Mosbach-Schulz, O and Hardy, A, 2018a. Guidance on Uncertainty Analysis in Scientific Assessments. EFSA Journal 2018;16(1):5123, 39 pp. https://doi.org/10.2903/j.efsa.2018.5123
- EFSA Scientific Committee, Benford, D, Halldorsson, T, Jeger, MJ, Knutsen, HK, More, S, Naegeli, H, Noteborn, H, Ockleford, C, Ricci, A, Rychen, G, Schlatter, JR, Silano, V, Solecki, R, Turck, D, Younes, M, Craig, P, Hart, A, Von Goetz, N, Koutsoumanis, K, Mortensen, A, Ossendorp, B, Germini, A, Martino, L, Merten, C, Mosbach-Schulz, O, Smith, A and Hardy, A, 2018b. Scientific Opinion on the principles and methods behind EFSA's Guidance on Uncertainty Analysis in Scientific Assessment. EFSA Journal 2018;16(1):5122, 235 pp. https://doi.org/10.2903/j.efsa.2018.5122
- EFSA Scientific Committee, More SJ, Bampidis V, Benford D, Bragard C, Halldorsson TI, Hernandez-Jerez AF, Bennekou SH, Koutsoumanis K, Lambre C, Machera K, Mennes W, ´ Mullins E, Nielsen SS, Schrenk D, Turck D, Younes M, Aerts M, Edler L, Sand S, Wright M, Binaglia M, Bottex B, Cortinas Abrahantes J and Schlatter J, 2022. Guidance on the use of the benchmark dose ~ approach in risk assessment. EFSA Journal 2022;20 (10):7584, 67 pp. https://doi.org/10.2903/j.efsa.2022.7584
- Egan PA, Stevenson PC, Tiedeken EJ, Wright GA, Boylan F and Stout JC, 2016. Plant toxin levels in nectar vary spatially across native and introduced populations. Journal of Ecology, 104, 1106–1115.
- Egan PA, 2015. Chemical Ecology and Conservation Biogeography of Rhododendron Ponticum L (Doctoral dissertation, Trinity College Dublin).

- Eraslan GÖ, Kanbur MU, Karabacak M, Arslan KO, Siliğ Y, Soyer Sarica ZE, Tekeli MY and Taş A, 2018. Effect on oxidative stress, hepatic chemical metabolizing parameters, and genotoxic damage of mad honey intake in rats. Human and Experimental Toxicology, 37, 991–1004.
- Erenler AK, 2016. Cardiac effects of mad honey poisoning and its management in emergency department: a review from Turkey. Cardiovascular Toxicology, 16, 1–4.
- European Commission, 2022. Honey Market presentation, Expert Group for Agricultural Markets, 20 October 2022. Available online: https://agriculture.ec.europa.eu/system/files/2022-10/market-presentation-honey_autumn2022_ en.pdf

Frelin C, Lombet A, Vigne P, Romey G and Lazdunski M, 1982. Properties of Na+ channels in fibroblasts. Biochemical and Biophysical Research Communications, 107, 202–208.

- Fukuda H, Kudo Y, Ono H, Yasue M, Sakakibara J and Kato T, 1974. Structure-activity relationship of lyoniol-A and related compounds in association with the excitatory effect on muscle spindle afferents. Chemical and Pharmaceutical Bulletin, 22, 884–888.
- Fukumoto K, Tetsumi T, Nishikawa Y, Katakawa JI, Meguri H, Katai M and Kawaraya T, 1993. Acute toxicity of grayanotoxin III on total protein and electrophoresis fractions in rat serum. Seikatsu Eisei (Journal of Urban Living and Health Association), 37, 73–77.
- Furbee B and Wermuth M, 1997. Life-threatening plant poisoning. Critical Care Clinics, 13, 849–888. https://doi. org/10.1016/s0749-0704(05)70372-9
- Gossinger H, Hruby K, Haubenstock A, Pohl A and Davogg S, 1983. Cardiac arrhythmias in a patient with grayanotoxin-honey poisoning. Veterinary and Human Toxicology, 25, 328–329.
- Gunduz A, Turedi S, Uzun H and Topbas M, 2006. Mad honey poisoning. The American Journal of Emergency Medicine, 24, 595–598.
- Gunduz A, Bostan H, Turedi S, Nuhoğlu İ and Patan T, 2007. Wild flowers and mad honey. Wilderness and Environmental Medicine, 18, 69–71.
- Gunduz A, Turedi S, Russell RM and Ayaz FA, 2008. Clinical review of grayanotoxin/mad honey poisoning past and present. Clinical Toxicology (Phila), 46, 437–442. https://doi.org/10.1080/15563650701666306
- Gunduz A, Aydin M, Akca M, Türkmen S, Türedi S, Eryiğit U, Cansu A and Yildirim M, 2012a. Is grayanotoxin directly responsible for mad honey poisoning-associated seizures. Turkish Journal of Medical Sciences, 42, 1086–1092.
- Gunduz A, Kalkan A, Turedi S, Durmus I, Turkmen S, Ayaz FA and Ayar A, 2012b. Pseudocholinesterase levels are not decreased in grayanotoxin (mad honey) poisoning in most patients. The Journal of Emergency Medicine, 43, 1008–1013.
- Hanson JR, 2016. From 'mad honey' to hypotensive agents, the grayanoid diterpenes. Science Progress, 99, 327–334. https://doi.org/10.3184/003685016x14720691270831
- Hikino H, Ohta T, Ogura M, Ohizumi Y, Konno C and Takemoto T, 1976. Structure-activity relationship of ericaceous. Toxins on acute toxicity in mice. Toxicology and Applied Pharmacology, 35, 303–310.
- Hikino H, Ohizumi Y, Konno C, And HK and Wakasa H, 1979. Subchronic toxicity of ericaceous toxins and Rhododendron leaves. Chemical and Pharmaceutical Bulletin, 27, 874–879.
- Holstege DM, Puschner B and Le T, 2001. Determination of grayanotoxins in biological samples by LC-MS/MS. Journal of Agricultural and Food Chemistry, 49, 1648–1651.
- Hong SJ and Chang CC, 1983. Potentiation by crotamine of the depolarizing effects of batrachotoxin, protoveratrine A and grayanotoxin I on the rat diaphragm. Toxicon, 21, 503–514.
- Hotta Y, Takeya K, Kobayashi S, Harada N, Sakakibara J and Shirai N, 1980. Relationship between structure, positive inotropic potency and lethal dose of grayanotoxins in guinea pig. Archives of Toxicology, 44, 259–267.
- Hotta Y, Ando H, Takeya K and Sakakibara J, 1994. Direct measurement of increased myocardial cellular 23 Na NMR signals in perfused guinea-pig heart induced by dihydroouabain and grayanotoxin-I. Molecular and Cellular Biochemistry, 139, 59–70.
- Hough RL, Crews C, White D, Driffield M, Campbell CD and Maltin C, 2010. Degradation of yew, ragwort and rhododendron toxins during composting. Science of the Total Environment, 408, 4128–4137.
- Huybrechts I, Sioen I, Boon PE, Ruprich J, Lafay L, Turrini A, Amiano P, Hirvonen T, De Neve M, Arcella D and Moschandreas J, 2011. Dietary exposure assessments for children in Europe (the EXPOCHI project): rationale, methods and design. Archives of Public Health, 69, 1–2.
- Hwang T, Noh E, Jeong JH, Park SK, Shin D and Kang H, 2018. Determination of grayanotoxins from Rhododendron brachycarpum in dietary supplements and homemade wine by Liquid Chromatography– Quadrupole Time-of-Flight–Mass Spectrometry and Liquid Chromatography–Tandem Mass Spectrometry. Journal of Agricultural and Food Chemistry, 66, 1935–1940.
- Ishii H, Kinoshita E, Kimura T, Yakehiro M, Yamaoka K, Imoto K, Mori Y and Seyama I, 1999. Point-mutations related to the loss of batrachotoxin binding abolish the grayanotoxin effect in Na+ channel isoforms. The Japanese Journal of Physiology, 49, 457–461.

Isom LL, 2001. Sodium channel β subunits: anything but auxiliary. The Neuroscientist, 7, 42–54.

Ito K, Saruwatari N, Mitani K and Enomoto Y, 1985. Characterization of depolarization induced by palytoxin and grayanotoxin-I in isolated cardiac tissues from dogs and guinea pigs. Naunyn-Schmiedeberg's Archives of Pharmacology, 330, 67–73.

- Jansen SA, Kleerekooper I, Hofman ZL, Kappen IF, Stary-Weinzinger A and van der Heyden MA, 2012. Grayanotoxin poisoning:'mad honey disease' and beyond. Cardiovascular Toxicology, 12, 208–215.
- Jauhari AC, Johorey AC, Banerjee I, Shrestha P and Singhal KC, 2009. Nearly fatal wild honey intoxication: a case report of seven cases. Journal of Clinical and Diagnosis Research, 3, 1685–1689.
- Kalkan A, Gökçe M and Memetoglu ME, 2012. An unusual clinical state: atrial fibrillation due to mad-honey intoxication/Nadir bir klinik durum: deli bal zehirlenmesine bagli gelisen atriyal fibrilasyon. The Anatolian Journal of Cardiology, 12, 365.
- Kaplan M, Olgun EO and Karaoglu O, 2014. Determination of grayanotoxins in honey by liquid chromatography tandem mass spectrometry using dilute-and-shoot sample preparation approach. Journal of Agricultural and Food Chemistry, 62, 5485–5491.
- Kerkvliet JD, 1981. Analysis of a toxic rhododendron honey. Journal of Apicultural Research, 20, 249-253.
- Kim SE, Shin MC, Akaike N and Kim CJ, 2010. Presynaptic effects of grayanotoxin III on excitatory and inhibitory nerve terminals in rat ventromedial hypothalamic neurons. Neurotoxicology, 31, 230–238.
- Kimura T, Yamaoka K, Kinoshita E, Maejima H, Yuki T, Yakehiro M and Seyama I, 2001. Novel site on sodium channel α -subunit responsible for the differential sensitivity of grayanotoxin in skeletal and cardiac muscle. Molecular Pharmacology, 60, 865–872.
- Kobayashi T, Nakamura H and Yasuda M, 1990. Disturbance of refinement of retinotectal projection in chick embryos by tetrodotoxin and grayanotoxin. Developmental Brain Research, 57, 29–35.
- Koca I and Koca AF, 2007. Poisoning by mad honey: a brief review. Food and Chemical Toxicology, 45, 1315–1318. https://doi.org/10.1016/j.fct.2007.04.006
- Koda R, Honma M, Suzuki K, Kasai A, Takeda T, Narita I and Yoshida K, 2016. Hypotension and bradycardia caused by the inadvertent ingestion of Rhododendron japonicum. Internal Medicine, 55, 839–842.
- Kohanawa M, 1956. Pharmacological studies on 'Rhodotoxin' isolated from Rhododendron Hymenanthes 'andro medotoxin' isolated from Andromeda Japonica Thunberg. The Japanese Journal of Pharmacology, 6, 46–56.
- Kükner A, Ilter G, Söyler G, Rasgele PG, Kekeçoglu M and Kambur M, 2016. The effect of rhododendron honey on mice liver tissue. International Journal of Morphology, 34, 842–847.
- Kurtoglu AB, Yavuz R and Evrendilek GA, 2014. Characterisation and fate of grayanatoxins in mad honey produced from Rhododendron ponticum nectar. Food Chemistry, 161, 47–52.
- Kuru P, Torun M, Halac HM, Temiz G, Iskender E, Karamahmutoglu T, Idrizoglu MG and Onat FY, 2014. Electroencephalographic and behavioral effects of intracerebroventricular or intraperitoneal injections of toxic honey extract in adult Wistar rats and GAERS. Neurological Sciences, 35, 1903–1908.
- Lee SW, Choi SH, Hong YS and Lim SI, 2007. Grayanotoxin poisoning from flower of Rhododendron mucronulatum in humans. Bulletin of Environmental Contamination and Toxicology, 78, 11–12.
- Lee SY, Choi YJ, Lee KB, Cho TY, Kim JS, Son YW, Park JS, Im SI, Choi HJ and Lee DH, 2008. Determination and monitoring of grayanotoxins in honey using LC-MS/MS. Korean Journal of Food Science and Technology, 40, 8–14.
- Lechtenberg M, Dierks F, Sendker J, Louis A, Schepker H and Hensel A, 2014. Extracts from Rhododendron ferrugineum do not exhibit grayanotoxin I: an analytical survey on grayanotoxin I within the genus Rhododendron. Planta Medica, 80, 1321–1328. https://doi.org/10.1055/s-0034-1383039
- Li C, Jinjing C and Mao H, 1982. The effect of rhomotoxin on the carotid pressor reflex. Acta Academiae Medicinae Wuhan, 2, 149–153.
- Li Y, Liu YB and Yu SS, 2013. Grayanoids from the Ericaceae family: structures, biological activities and mechanism of action. Phytochemistry Reviews, 12, 305–325. https://doi.org/10.1007/s11101-013-9299-z
- Li Y, Liu YB, Zhang JJ, Liu Y, Ma SG, Qu J, Lv HN and Yu SS, 2015. Antinociceptive grayanoids from the roots of Rhododendron molle. Journal of Natural Products, 78, 2887–2895.
- Li Y, Zhu Y, Zhang Z, Li L, Liu Y, Qu J, Ma S and Yu S, 2020. Antinociceptive grayanane-derived diterpenoids from flowers of Rhododendron molle. Acta Pharmaceutica Sinica B, 10, 1073–1082.
- Li CH, Yan XT, Zhang AL and Gao JM, 2017. Structural diversity and biological activity of the genus Pieris terpenoids. Journal of Agricultural and Food Chemistry, 65, 9934–9949.
- Li CH, Zhang JY, Zhang XY, Li SH and Gao JM, 2019. An overview of grayanane diterpenoids and their biological activities from the Ericaceae family in the last seven years. European Journal of Medical Chemistry, 166, 400–416. https://doi.org/10.1016/j.ejmech.2019.01.079
- Li Z and Mao H, 1982. Clinical observation of haemodynamic effects of rhomotoxin in hypertensive patients. Acta Academiae Medicinae Wuhan, 2, 138–143.
- Lim MY, Phua DH and Ooi CK, 2016. Heart stopping honey—not just Turkish honey. The American Journal of Emergency Medicine, 34, 1915–e1.
- Liu YJ, Ya-Zhen W, Huan-Yuan M, He-Ping G, Shu-Mao L, Xi-Lin C, Yi-Hua C, De-Cun S, Jia-Xiu S and Zhi-Wei S, 1986. Distribution and excretion of 3H-rhomotoxin in mice. Journal of Tongji Medical University, 6, 172–175.
- Lucatello L, Piana L, Fasolato L and Capolongo F, 2022. A multivariate statistical approach to identify the factors influencing the grayanotoxin content of Italian Rhododendron honey. Food Control, 136, 108881.
- Machado AM, Miguel MG, Vilas-Boas M and Figueiredo AC, 2020. Honey volatiles as a fingerprint for botanical origin-a review on their occurrence on monofloral honeys. Molecules, 25, 374.

- Maejima H, Kinoshita E, Yuki T, Yakehiro M, Seyama I and Yamaoka K, 2002. Structural determinants for the action of grayanotoxin in D1 S4–S5 and D4 S4–S5 intracellular linkers of sodium channel α -subunits. Biochemical and Biophysical Research Communications, 295, 452–457.
- Mao HY, Tu Y, Nei F, Liang G and Feng Y, 1981. Rapid antihypertensive effect of rhomotoxin in 105 hypertension cases. Chinese Medical Journal, 94, 733–736.

Mao HY, Li CY, Cui JJ, Feng YB, Hu WS, Guo QG and Jiang MX, 1982. Rhomotoxin pharmacologic action in lowering blood pressure and slowing heart rate. Chinese Medical Journal, 95, 311–318.

Manceau, 2015. Rhododendron Ponticum L. Code de conduite, Plantes envahissantes.

- Masom CP, Kane KE and Katz KD, 2008. A 2-year-old girl with bradycardia and lethargy: is perseus to the rescue? Pediatric Emergency Care, 34, e60–e63.
- Masutani TE, Seyama IS, Narahashi TO and Iwasa JU, 1981. Structure-activity relationship for grayanotoxin derivatives in frog skeletal muscle. Journal of Pharmacology and Experimental Therapeutics, 217, 812–819.
- Martinet O, Pommier P, Sclossmacher P, Develay A and De Haro L, 2005. Poisoning by scabies wood (Agauria salicifolia). The Medical Press, 34, 797–798.
- Matthews JC, Warnick JE, Albuquerque EX and Eldefrawi ME, 1981. Characterization of the electrogenic sodium channel from rat brain membranes using neurotoxin-dependent 22Na uptake. Membrane Biochemistry, 4, 71–104.
- Mayda N, Özkök A and Sorkun K, 2018. Some characteristic properties of chestnut and rhododendron honeys in Turkey. Hacettepe Journal of Biology and Chemistry, 46, 135–145.
- Meguri H, 1959. Isolation of Pieristoxins A, B and C from the flowers of Pieris. Japonica Journal of the Pharmaceutical of Japan, 79, 1052.
- Mei H, Korfmacher W and Morrison R, 2006. Rapid in vivo oral screening in rats: reliability, acceptance criteria, and filtering efficiency. The AAPS Journal, 8, E493–E500.
- Merten C, Ferrari P, Bakker M, Boss A, Hearty A, Leclercq C, Lindtner O, Tlustos C, Verger P, Volatier JL and Arcella D, 2011. Methodological characteristics of the national dietary surveys carried out in the European Union as included in the European Food Safety Authority (EFSA) Comprehensive European Food Consumption Database. Food Additives & Contaminants: Part A, 28, 975–995.
- Mol HG, Van Dam RC, Zomer P and Mulder PP, 2011. Screening of plant toxins in food, feed and botanicals using full-scan high-resolution (Orbitrap) mass spectrometry. Food Additives & Contaminants: Part A, 28, 1405–1423.
- Moran NC, Perkins ME and Richardson AP, 1954. The action of andromedotoxin on the carotid sinus in dogs. Journal of Pharmacology and Experimental Therapeutics, 111, 454–458.
- Narahashi T and Seyama I, 1974. Mechanism of nerve membrane depolarization caused by grayanotoxin I. The Journal of Physiology, 242, 471.
- Narayanan P, Röhrl M, Zechmeister K and Hoppe W, 1970. Crystal and molecular structure of grayanotoxin-I, C22H36O7. Tetrahedron Letters, 45, 3943–3944. https://doi.org/10.1016/s0040-4039(01)98631-9
- Nassibou S, Paradis C, Frete F, Allorge D, Gaulier JM and Labadie M, 2020. Mad honey of Nepalese origin-related grayanatoxin intoxication. Toxicon: Official Journal of the International Society on Toxinology, 184, 202–203.
- National Food Institute, Technical University of Denmark, Bredsdorff L, Olesen PL, Sharma AK, Hansen M, Jørgensen K, Ekstrøm J, Ravn-Haren G, Skjolding LM, Baun A and Beltoft VM, 2020. Extensive literature search on grayanotoxins and 5-hydroxymethylfurfural. EFSA supporting publication 2020:EN-1920, 52 pp. https://doi. org/10.2903/sp.efsa.2020.EN-1920
- Niu CS, Li Y, Liu YB, Ma SG, Li L, Qu J and Yu SS, 2016. Analgesic diterpenoids from the twigs of Pieris formosa. Tetrahedron, 72, 44–49.
- Niu CS, Li Y, Liu YB, Ma SG, Liu F, Cui L, Yu HB, Wang XJ, Qu J and Yu SS, 2018. Grayanane diterpenoids with diverse bioactivities from the roots of Pieris formosa. Tetrahedron, 74, 375–382.
- Ocak T, Duran A, Basturk M and Sahin I, 2013. Mad honey poisoning presenting with ventricular tachycardia and acute myocardial infarction: a case report. Hong Kong Journal of Emergency Medicine, 20, 299–301.
- Oddo LP, Piro R, Bruneau É, Guyot-Declerck C, Ivanov T, Piskulová J, Flamini C, Lheritier J, Morlot M, Russmann H and Von der Ohe W, 2004. Main European unifloral honeys: descriptive sheets. Apidologie, 35, S38–S81.
- OECD Test No. 420: Acute Oral Toxicity: Fixed Dose Procedure, 2001. OECD Guidelines for the Testing of Chemicals, OECD Publishing, Paris. https://doi.org/10.1787/9789264070943-en
- OECD Test No. 475: Mammalian Bone Marrow Chromosomal Aberration Test, 2016. OECD Guidelines for the Testing of Chemicals, OECD Publishing, Paris. https://doi.org/10.1787/9789264264786-en
- Ohgaki T, Meguri H, Ogita K and Yoneda Y, 1987. Tetrodotoxin-insensitive central depression by grayanotoxin-III in mice. Brain Research, 425, 364–368.
- Ohgaki T, Uchida S, Meguri H, Ogita K and Yoneda Y, 1988. Preventive action of quisqualic acid against grayanotoxin-induced suppression of locomotor activity in mice. Neuropharmacology, 27, 1045–1053.
- Okuyan E, Uslu A and Ozan Levent M, 2010. Cardiac effects of "mad honey": a case series. Clinical Toxicology, 48, 528–532.
- Onat F, Yegen BC, Lawrence R, Oktay A and Oktay S, 1991a. Site of action of grayanotoxins in mad honey in rats. Journal of Applied Toxicology, 11, 199–201. https://doi.org/10.1002/jat.2550110308
- Onat F, Kurtel H, Yegen BC, Lawrence R, Oktay A and Oktay S, 1991b. Mad honey poisoning in man and rat. Marmara Medical Journal, 183, 575. Conference abstract

- Ösken A, Aydın E, Kocayigit İ, Cakar MA, Gündüz H, Yaylacı S and Tamer A, 2012. Slow ventricular response atrial fibrillation related to mad honey poisoning. Journal of Cardiovascular Disease Research, 3, 245–247.
- Ösken A, Aydın E, Özcan KS and Yaylacı S, 2021. Evaluation of Electrocardiographic Parameters and the Presence of Interatrial Block in Patients with Mad Honey Intoxication. Cardiovascular Toxicology, 21, 772–780.
- Özhan H, Akdemir R, Yazici M, Gündüz H, Duran S and Uyan C, 2004. Cardiac emergencies caused by honey ingestion: a single centre experience. Emergency Medicine Journal, 21, 742–744.
- Panseri S, Manzo A, Chiesa LM and Giorgi A, 2013. Melissopalynological and volatile compounds analysis of buckwheat honey from different geographical origins and their role in botanical determination. Journal of Chemistry.
- Peng X, Zhang H, Yuan X, Chen Z, Gao J, Teng Y and Yao G, 2020. Grayanane diterpenoids from the leaves of Rhododendron dauricum. Biochemical Systematics and Ecology, 89, 104009.
- Pischon H, Petrick A, Müller M, Köster N, Pietsch J and Mundhenk L, 2018. Grayanotoxin I intoxication in pet pigs. Veterinary Pathology, 55, 896–899.
- Poon WT, Ho CH, Yip KL, Lai CK, Cheung KL, Sung RY, Chan AY and Mak TW, 2008. Grayanotoxin poisoning from Rhododendron simsii in an infant. Hong Kong Medicinal Journal, 14, 405–407.
- Popescu R and Kopp B, 2013. The genus Rhododendron: an ethnopharmacological and toxicological review. Journal of Ethnopharmacology, 147, 42–62.
- Puschner B, Holstege DM and Lamberski N, 2001. Grayanotoxin poisoning in three goats. Journal of the American Veterinary Medical Association, 218, 573–575.
- Qiang Y, Zhou B and Gao K, 2011. Chemical constituents of plants from the genus Rhododendron. Chemistry & Biodiversity, 8, 792–815.
- Rasgele PG, Gokalp FD, Kaya ST, Kekecoglu M and Acar MK, 2021. Investigation of genotoxic effects of rhododendron honey using three mammalian bioassays in vivo. Drug and Chemical Toxicology, 45, 2301–2310.
- Sahin A, 2014. PhD thesis. Orman gülü bali ve bitkisindeki grayanotoksin-iii (gtx-iii) izoformunun lc-ms/ms ile analizi. Karadeniz teknik üniversitesi fen bilimleri enstitüsü.
- Sahin H, Turumtay EA, Yildiz O and Kolayli S, 2015. Grayanotoxin-III detection and antioxidant activity of mad honey. International Journal of Food Properties, 18, 2665–2674.
- Sahin H, Yildiz O and Kolayli S, 2016. Effects of Mad Honey on Some Biochemical Parameters in Rats. Journal of Evidence-Based Complementary & Alternative Medicine, 21, 255–259. https://doi.org/10.1177/ 2156587215596430
- Sahin A, Turkmen S, Guzel N, Mentese A, Turedi S, Karahan SC, Yulug E, Demir S, Aynaci O, Deger O and Gunduz A, 2018. A comparison of the effects of grayanotoxin-containing honey (mad honey), normal honey, and propolis on fracture healing. Medical Principles and Practice, 27, 99–106.
- Sarwar G and Takahashi H, 2013. Pollen morphology of *Rhododendron* L. and related genera and its taxonomic significance. Bangladesh Journal of Plant Taxonomy, 20, 185–199.
- Sayin MR, Dogan SM, Aydin M and Karabag T, 2011. Extreme QT interval prolongation caused by mad honey consumption. Canadian Journal of Cardiology, 27, 870–e17-9.
- Sayin MR, Karabag T, Dogan SM, Akpinar I and Aydin M, 2014. Transient ST segment elevation and left bundle branch block caused by mad-honey poisoning. Wiener Klinische Wochenschrift, 124, 278–281.
- Scott PM, Coldwell BB and Wiberg GS, 1971. Grayanotoxins. Occurrence and analysis in honey and a comparison of toxicities in mice. Food and Cosmetic. Toxicology, 9, 179–184. https://doi.org/10.1016/0015-6264(71)90303-8
- Seephonkai P, Popescu R, Zehl M, Krupitza G, Urban E and Kopp B, 2021. Ferruginenes A-C from Rhododendron ferrugineum and their cytotoxic evaluation. Journal of Natural Products, 74, 712–717.
- Seyama I and Narahashi T, 1973. Increase in sodium permeability of squid axon membranes by αdihydrograyanotoxin II. Journal of Pharmacology and Experimental Therapeutics, 184, 299–307.
- Seyama I, 1978. Effect of grayanotoxin I on SA node and right atrial myocardia of the rabbit. American Journal of Physiology-Cell Physiology, 235, C136–C142.
- Shi Y, Zhou M, Zhang Y, Fu Y, Li J and Yang X, 2021. Poisonous delicacy: Market-oriented surveys of the consumption of Rhododendron flowers in Yunnan, China. Journal of Ethnopharmacology, 265, 113320.
- Shrestha P, Vaidya R and Sherpa K, 2009. Mad honey poisoning: a rare case report of seven cases. Nepal Medical College Journal, 11, 212–213.
- Shrestha N, Wang Z, Su X, Xu X, Lyu L, Liu Y, Dimitrov D, Kennedy JD, Wang Q, Tang Z and Feng X, 2018. Global patterns of Rhododendron diversity: the role of evolutionary time and diversification rates. Global Ecology and Biogeography, 27, 913–924. https://doi.org/10.1111/geb.12750
- Sibel S, Enis YM, Hüseyin S, Timucin AA and Duran O, 2014. Analysis of grayanatoxin in Rhododendron honey and effect on antioxidant parameters in rats. Journal of Ethnopharmacology, 156, 155–161.
- Silici S and Atayoglu AT, 2015. Mad honey intoxication: a systematic review on the 1199 cases. Food and Chemical Toxicology, 86, 282–290.
- Silici S, Doga Z, Sahin H, Atayoglu T and Yakan B, 2016. Acute effects of grayanotoxin in rhododendron honey on kidney functions in rats. Environmental Science and Pollution Research, 23, 3300–3309. https://doi.org/10. 1007/s11356-015-5534-z

- Sohn CH, Seo DW, Ryoo SM, Lee JH, Kim WY, Lim KS and Oh BJ, 2014. Clinical characteristics and outcomes of patients with grayanotoxin poisoning after the ingestion of mad honey from Nepal. Internal and Emergency Medicine, 9, 207–211.
- Sosnovsky Y, Nachychko V, Prokopiv A and Honcharenko V, 2017. Leaf architecture in Rhododendron subsection Rhododendron (Ericaceae) from the Alps and Carpathian Mountains: taxonomic and evolutionary implications. Flora, 230, 26–38.
- Scott PM, Coldwell BB and Wiberg GS, 1971. Grayanotoxins. Occurrence and analysis in honey and a comparison of toxicities in mice. Food and Cosmetics Toxicology, 9, 179–184.
- Stout JC, Parnell JA, Arroyo J and Crowe TP, 2006. Pollination ecology and seed production of Rhododendron ponticum in native and exotic habitats. Biodiversity & Conservation, 15, 755–777.
- Stout JC, Duffy KJ, Egan PA, Harbourne M and Hodkinson TR, 2015. Genetic diversity and floral width variation in introduced and native populations of a long-lived woody perennial. AoB Plants, 7, plu087.
- Sumerkan MC, Simsek EE, Mayda AS and Agirbasli MA, 2013. OP-081 Epidemiological evaluation of mad-honey intoxication. International Journal of Cardiology, 3, S32–S33.
- Sun N, Zheng G, He M, Feng Y, Liu J, Wang M, Zhang H, Zhou J and Yao G, 2019. Grayanane diterpenoids from the leaves of Rhododendron auriculatum and their analgesic activities. Journal of Natural Products, 82, 1849–1860.
- Tallent WH, Riethof ML and Horning EC, 1957. Studies on the occurrence and structure of acetylandromedol (andromedotoxin). Journal of the American Chemical Society, 79, 4548–4554. https://doi.org/10.1021/ja01573a080
- Tekinsoy A, Yasli SO, Saritas A, Gunes H, Kaya E and Sonmez FT, 2017. Analysis of mad honey (grayanotoksin) cases admitted to Duzce University School of Medicine Emergency Department. Open Journal of Emergency Medicine, 5, 13–24.
- Terai T, Araho D, Osakabe K, Katai M, Narama I, Matsuura T, Katakawa JI, Tetsumi T and Sato M, 2000. Isolation of iso-grayanotoxin II from leaves of Leucothoe grayana Max. Its X-ray crystallographic analysis and acute toxicity in mice. Chemical and Pharmaceutical Bulletin, 48, 142–144.
- Terai T, Osakabe K, Katai M, Sakaguchi K, Narama I, Matsuura T, Katakawa J and Tetsumi T, 2003. Preparation of 9-hydroxy grayanotoxin derivatives and their acute toxicity in mice. Chemical and Pharmaceutical Bulletin (Tokyo), 51, 351–353. https://doi.org/10.1248/cpb.51.351
- These A, Bodi D, Uecker S, Reimers K, Ronczka S, Preiss-Weigert A and Lahrssen-Wiederholt M, 2015. A case of human poisoning by grayanotoxins following honey ingestion: elucidation of the toxin profile by mass spectrometry. Food Additives & Contaminants, 32, 1674–1684. https://doi.org/10.1080/19440049.2015. 1042410
- Tian T, Jin Y, Ma Y, Xie W, Xu H, Zhang K, Zhang L and Du Y, 2015. Identification of metabolites of oridonin in rats with a single run on UPLC-Triple-TOF-MS/MS system based on multiple mass defect filter data acquisition and multiple data processing techniques. Journal of Chromatography B, 1006, 80–92.
- Tiedeken EJ, Egan PA, Stevenson PC, Wright GA, Brown MJ, Power EF, Farrell I, Matthews SM and Stout JC, 2016. Nectar chemistry modulates the impact of an invasive plant on native pollinators. Functional Ecology, 30, 885– 893.
- Tschekunow H, Klug S and Marcus S, 1989. Symptomatic poisoning from ingestion of Pieris japonica: a case report. Veterinary and Human Toxicology, 31, 360.
- Turkmen S, Karagoz U, Gunduz A, Turedi S, Akca M and Yildirim M, 2013. The dose-dependent effect of grayanotoxin on the cardiovascular system. Turkish Journal of Medical Sciences, 43, 700–705. https://doi.org/ 10.3906/sag-1209-11
- Turner PV, Brabb T, Pekow C and Vasbinder MA, 2011. Administration of substances to laboratory animals: routes of administration and factors to consider. Journal of the American Association for Laboratory Animal Science, 50, 600–613.
- Ugur HG, Sıralı R, Tekgul AT and Efe B, 2019. Investigation of mad honey use as an alternative treatment in patients admitted to the pulmonary clinic: Ordu, Turkey example. Brazilian Archives of Biology and Technology, 62, e19180488.
- Ullah S, Khan SU, Saleh TA and Fahad S, 2018. Mad honey: uses, intoxicating/poisoning effects, diagnosis, and treatment. RSC Advances, 8, 18635–18646.
- Uzun H, Sari I, Gunes C and Kocabay K, 2013. A child with bradycardia and hypotension related to mad honey intoxication. Turkish Archives of Pediatrics, 48, 53–54.
- Wang X, Jiang R, Liu Z, Liu W, Xie M, Wei S and She G, 2014. Phytochemicals and biological activities of poisonous genera of Ericaceae in China. Natural Product Communications, 9, 1934578X1400900333.
- Weber M, Cadivel A, Chappel V, Abinaber F, Le Gallo A, Ragonneau S, Verdiere C, Lassalle C, Metas E and D'Ortenzio E, 2009. Collective intoxication with "mad honey" on Reunion Island. Bulletin de la Société de Pathologie Exotique (1990), 102, 7–8.
- Weiss TW, Smetana P, Nurnberg M and Huber K, 2010. The honey man—Second degree heart block after honey intoxication. International Journal of Cardiology, 142, e6–e7.

- WHO/ICPS (World Health Organization/International Programme on Chemical Safety), 2005. International Programme on Chemical Safety & IPCS Workshop on Incorporating Uncertainty and Variability into Risk Assessment (2000: Berlin, Germany). (2005). Chemical-specific adjustment factors for interspecies differences and human variability: guidance document for use of data in dose/concentration-response assessment. World Health Organization Available online: https://apps.who.int/iris/handle/10665/43294
- WHO/IPCS (World Health Organization/International Programme on Chemical Safety), 2009. Principles and methods for the risk assessment of chemicals in food (Environmental Health Criteria 240). Available online: https://apps.who.int/iris/bitstream/handle/10665/44065/WHO_EHC_240_eng.pdf

Yavuz H, Özel A, Akkus I and Erkul I, 1991. Honey poisoning in Turkey. The Lancet, 337, 789–790.

- Yaylaci S, Kocayigit I, Aydin E, Osken A, Genc AB, Cakar MA and Tamer A, 2014. Clinical and laboratory findings in mad honey poisoning: a single center experience. Nigerian Journal of Clinical Practice, 17, 589–593.
- Yaylaci S, Ayyildiz O, Aydin ER, Osken A, Karahalil F, Varim C, Demir MV, Genç AB, Sahinkus S, Can Y and Kocayigit I, 2015. Is there a difference in mad honey poisoning between geriatric and non-geriatric patient groups. European Review for Medical and Pharmacological Sciences, 19, 4647–4653.
- Yilmaz O, Eser M, Sahiner A, Altintop L and Yesildag O, 2006. Hypotension, bradycardia and syncope caused by honey poisoning. Resuscitation, 68, 405–408.
- Yilmaz İ, KaYa E, YaYKaşlI KO and TürKEr Y, 2017. Sıçanlarda intravenöz uygulanan deli bal toksini grayanotoksin-III'ün doza bağımlı kardiyak etkileri. Journal of Dr. Behcet Uz Children's Hospital, 7.
- Zhang HP, Wang HB, Wang LQ, Bao GH and Qin GW, 2005. Note: a new 1, 5-seco grayanotoxin from Rhododendron decorum. Journal of Asian Natural Products Research, 7, 87–90.
- Zhang M, Xie Y, Zhan G, Lei L, Shu P, Chen Y, Xue Y, Luo Z, Wan Q, Yao G and Zhang Y, 2015. Grayanane and leucothane diterpenoids from the leaves of Rhododendron micranthum. Phytochemistry, 117, 107–115.
- Zhang H, Zheng X, Zheng G, Teng Y, Zhou J and Yao G, 2020. Chemical constituents from the leaves of Lyonia ovalifolia var. hebecarpa. Biochemical Systematics and Ecology, 92, 104129.
- Zhang Q, Chen X, Chen S, Liu Z, Wan R and Li J, 2016. Fatal honey poisoning caused by Tripterygium wilfordii Hook F in Southwest China: a case series. Wilderness & Environmental Medicine, 27, 271–273.
- Zheng G, Zhou J, Huang L, Zhang H, Sun N, Zhang H, Jin P, Yue M, Meng L and Yao G, 2019. Antinociceptive grayanane diterpenoids from the leaves of Pieris japonica. Journal of Natural Products, 82, 3330–3339.
- Zhou J, Liu T, Zhang H, Zheng G, Qiu Y, Deng M, Zhang C and Yao G, 2018. Anti-inflammatory grayanane diterpenoids from the leaves of Rhododendron molle. Journal of Natural Products, 81, 151–161.
- Zou H-Y, Luo J, Xu D-R and Kong L-Y, 2014. Tandem solid-phase extraction followed by HPLC–ESI/QTOF/MS/MS for rapid screening and structural identification of trace diterpenoids in flowers of *Rhododendron molle*. Phytochemical Analysis, 25, 255–265.

Abbreviations

ADME	Absorption, Distribution, Metabolism, Elimination
BfR	German Federal Institute for Risk Assessment
BMD	Benchmark dose modelling
BMDL ₁₀	benchmark dose lower bound at 10% BMR
BMDU ₁₀	benchmark dose upper bound at 10% BMR
BMI	body mass index
BMR	benchmark response value
CONTAM	The EFSA Panel on Contaminants and in the Food Chain
ECHA	European Chemicals Agency
EMA	European Medicines Agency
EKE	expert knowledge elicitation
ESI	electrospray ionisation mode
GC	gas chromatography
GTX	grayanotoxin
HBGV	health-based guidance value
LC	liquid chromatography
LC-MS/MS	liquid chromatography coupled with tandem mass spectrometry
LD ₅₀	median lethal dose
LOAEL	lowest observed adverse effect level
LOD	limit of detection
LOQ	limit of quantification
MOE	margin of exposure
MS	mass spectroscopy
NOAEL	no-observed adverse effect level
RP	reference point

RPF	relative potency factor
SPE	solid phase extraction
ТК	toxicokinetic
UB	upper bound
UK	United Kingdom
USA	United States of America
WHO	World Health Organization


Appendix A – Grayananes considered in the opinion

Grayananes considered in this opinion

No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
1	 Grayanotoxin I [(1S,3R,4R,6S,8S,9R,10R,13R,14R,16R)- 3,4,6,9,14-pentahydroxy-5,5,9,14-tetramethyl-16- tetracyclo [11.2.1.01,10.04,8]hexadecanyl] acetate Andromedotoxin; acetylandromedol; andromedol acetate; rhodotoxin; asebotoxin 	$\begin{array}{c} C_{22}H_{36}O_7 \\ 4720\text{-}09\text{-}6 \\ 412.52 \\ 0.785 \pm 0.508 \\ 258\text{-}260^\circ\text{C} \end{array}$	5	HO HO OAc	14-acetate	L. grayana ^(b) , K. angustifolia, R. decorum (Li et al., 2013), P. formosa (Li et al., 2017), R. molle, R. decorum, R. yedoense, R. brachycarpum, R. przewalskii, R. oveodoxa (Qjang et al., 2011); R. micranthum, R. auriculatum, R. dauricum (Chai et al., 2020; Peng et al., 2020)
2	 Grayanotoxin III (1S,3R,4R,6S,8S,9R,10R,13R,14R,16R)- 5,5,9,14-tetramethyltetracyclo [11.2.1.01,10.04,8]hexadecane-3,4,6,9,14,16- hexol Andromedol; deacetylandromedotoxin; deacetylasebotoxin I 	$\begin{array}{l} C_{20}H_{34}O_6 \\ 4678\text{-}45\text{-}9 \\ 370.48 \\ -0.237 \pm 0.514 \\ 218^\circ\text{C} \text{ (decomp)} \end{array}$	6			L. grayana ^(b) , K. latifolia, R. molle (Li et al., 2013), P. formosa (Li et al., 2017), R. micranthum, R. auriculatum, R. dauricum (Chai et al., 2020; Peng et al, 2020)
	 1. 16-O-Acetylgrayanotoxin III 2. 3. 	C ₂₂ H ₃₆ O ₇ 412.52 No exp. value	5	HO H OH OAC	16-acetate	<i>R. micranthum</i> (Chai et al, 2020)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Grayanotoxin VI 2. (3R,4R,6S,8S,9R,10R,16S)-5,5,9,14- tetramethyltetracyclo[11.2.1.01,10.04,8]hexadec- 14-ene-3,4,6,9,16-pentol 3.	$\begin{array}{c} C_{20}H_{32}O_5 \\ 30460\text{-}36\text{-}7 \\ 352.47 \\ 0.717 \pm 0.493 \\ \text{No exp. value} \end{array}$	5		15,16-dehydro	<i>L. grayana</i> ^(b) (Li et al., 2013), <i>R. dauricum, R. micranthum</i> (Chai et al., 2020; Peng et al., 2020, Zhang et al., 2015)
	1. Asebotoxin I 2. [(1S,3R,4R,6S,8S,9R,10R,13R,14R)-3,4,6,9,14- pentahydroxy-5,5,9,14-tetramethyl-16-tetracyclo [11.2.1.01,10.04,8]hexadecanyl] propanoate 3. Pierisformosin B	$\begin{array}{c} C_{23H_{38}O_7} \\ \texttt{23984-17-0} \\ \texttt{426.54} \\ \texttt{1.295} \pm \texttt{0.508} \\ \texttt{196-198^{\circ}C} \end{array}$	5	HO HHO HHO OCOEt	14-propionate	<i>P. japonica</i> (Li et al., 2013), <i>P. formosa</i> (Li et al., 2017), <i>Rhododendron spp</i> . (Qjang et al., 2011)
	1. Asebotoxin III 2. (3,4,10,15-tetrahydroxy-5,5,10,15-tetramethyl- 7-oxapentacyclo [12.2.1.01,11.04,9.06,8] heptadecan-17-yl) 2-hydroxypropanoate 3.	$\begin{array}{c} C_{23H_{36}O_8} \\ 28894\text{-}73\text{-}7 \\ 440.53 \\ 0.877 \pm 0.594 \\ 258\text{-}260^\circ \text{C} \end{array}$	4 (5)	HO HO HO R = COCHOHMe	2,3-epoxide, 14- hydroxy- proprionate	<i>P. japonica</i> (Li et al., 2013)
	1. Asebotoxin VI 2. [(1R,2S,3R,4R,6S,8S,9R,10R,13R,14R,16R)- 3,4,6,9,14,16-hexahydroxy-5,5,9,14-tetramethyl- 2-tetracyclo[11.2.1.01,10.04,8]- hexadecanyl] propanoate 3.	$\begin{array}{l} C_{23}H_{38}O_8 \\ 63529\text{-}01\text{-}1 \\ 442.54 \\ 0.920 \pm 0.540 \\ \text{No exp. value} \end{array}$	6		7-propionate	<i>P. japonica</i> (Li et al., 2013)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Asebotoxin VII 2. [(1R,2S,3R,4R,6R,8S,9R,10S,11S,14R,15R, 17R)-2-acetyloxy-3,4,10,11,15-pentahydroxy- 5,5,10,15-tetramethyl-7-oxapentacyclo [12.2.1.01,11.04,9.06,8]heptadecan-17-yl] propanoate 3.	$C_{25}H_{38}O_{10}$ 34183-45-4 498.6 -0.7 No exp. value	5	HO H OH OH OCOEt OH	2,3-epoxide, 7- acetate, 14- proprionate	<i>P. japonica</i> (Li et al., 2013)
	1. Asebotoxin X 2. 3. Pieristoxin I	$\begin{array}{l} C_{23}H_{38}O_875796-\\ 14\text{-}4\\ 442.54\\ -0.039\pm0.539\\ \text{No exp. value} \end{array}$	5 (6)	HO HO HO HO HO R = COCHOHMe	14-hydroxy- propionate	<i>P. japonica</i> (Li et al., 2013)
	 Bisdeacetylkalmitoxin VI 3. 	$\begin{array}{l} C_{20}H_{34}O_8 \\ 75909\mathchar`-2.551 \pm 0.604 \\ \mbox{No exp. value} \end{array}$	8			<i>L. ovalifolia</i> (Zhang et al., 2020)
	 Craiobiotoxin II . . 	$\begin{array}{c} C_{20}H_{32}O_5 \\ 864679\text{-}90\text{-}3 \\ 352.47 \\ 1.822 \pm 0.551 \\ 115\text{-}117^\circ\text{C} \end{array}$	4	HO H H HO HO	2,3-epoxide	<i>C. yunnanense</i> (Li et al., 2013)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	 Craiobiotoxin III . . 	$\begin{array}{c} C_{25}H_{40}O_8\\ 864679\text{-}91\text{-}4\\ 468.58\\ 1.367 \pm 0.580\\ 135\text{-}137^\circ\text{C} \end{array}$	4 (5)	$R = \int_{0}^{HO} \int_{0}^{HO} H$	2,3-epoxide, 14- hydroxymethyl- butanoate	<i>C. henryi; C. yunnanense</i> (Li et al., 2013; Zhang et al., 2005)
	 Craiobiotoxin IV 3. 	$\begin{array}{l} C_{22}H_{34}O_8\\ 864679\text{-}92\text{-}5\\ 426.50\\ 1.502\pm0.624\\ 146\text{-}147^\circ\text{C} \end{array}$	6	HO OH OH OH	2,3-epoxide, 7-acetate	<i>C. yunnanens</i> e (Li et al., 2013)
	1. Craiobiotoxin VI 2. Grayanotoxin-2β,3β,5β,6β,10α,16α-hexol 3.	$\begin{array}{c} C_{20}H_{34}O_6\\ 864679\text{-}94\text{-}7\\ 370.48\\ -0.237\pm0.514\\ 162\text{-}163^\circ\text{C} \end{array}$	6			<i>C. henryi; C. yunnanense</i> (Li et al., 2013; Zhang et al., 2005)
	1. Craiobiotoxin IX 2. (1aS,1bS,2R,2aR,5R,6R,7aR,9R,9aR,10aR)- Dodecahydro-2,6,10,10-tetramethyl-5,7a- methano-7aH-cyclohept[6,7]azuleno[1,2-b] oxirene-2,5,6,9,9a-pentol 3.	$\begin{array}{c} C_{20}H_{32}O_6 \\ 1166135\text{-}98\text{-}3 \\ 368.46 \\ -0.103 \pm 0.588 \\ \text{No exp. value} \end{array}$	5	HO H OH	2,3-epoxide	<i>R. decorum</i> (Li et al., 2013; Qjang et al., 2011)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Dihydrolyoniol B 2. (2S,3R,4R,6S,8S,9R,10R,13S,14R)-5,5,9,14- tetramethyltetracyclo[11.2.1.01,10.04,8] hexadecane-2,3,4,6,9,14-hexol 3. Dihydrodeacetylyoniatoxin	$\begin{array}{c} C_{20}H_{34}O_6\\ 29708-82-5\\ 370.48\\ -0.128\pm0.533\\ \text{No exp. value} \end{array}$	6			
	 Kalmitoxin I (1R,2S,3R,4R,6S,8S,9R,10R,13R,14R,16R)- 5,5,9,14-tetramethyltetracyclo [11.2.1.01,10.04,8]hexadecane-2,3,4,6,9,14,16- heptol Desacetylpieristoxin C; desacetylpieristoxin F; bisdesacetylasebotoxin V 	$\begin{array}{c} C_{20}H_{34}O_7\\ 56663\text{-}60\text{-}6\\ 386\text{-}48\\ -1.237\pm0.561\\ 255\text{-}257^\circ\text{C} \end{array}$	7			<i>K. latifolia</i> (Li et al., 2013), <i>P. formosa</i> (Li et al., 2017)
	1. Kalmitoxin VI 2. [(1R,2S,3R,4R,6R,7R,8S,9R,10R,13R,14R,16R)- 2-acetyloxy-3,4,6,7,9,14-hexahydroxy-5,5,9,14- tetramethyl-16-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] acetate 3.	$\begin{array}{l} C_{24}H_{38}O_{10} \\ 72514\text{-}67\text{-}1 \\ 486.55 \\ 0.401 \pm 0.579 \\ \text{No exp. value} \end{array}$	6	HO, HO, H HO, HO HO, r>HO HO HO HO HO HO HO HO HO HO HO H	7,14-diacetate	<i>K. latifolia</i> (Li et al., 2013)
	1. Lyoniol A 2. [(1R,2S,3R,4R,6R,8S,9S,10R,11R,14R,15R)- 2,4,10,15-tetrahydroxy-5,5,10,15-tetramethyl-7- oxapentacyclo[12.2.1.01,11.04,9.06,8] heptadecan-3-yl] acetate 3. Lyoniatoxin	$\begin{array}{c} C_{22}H_{34}O_7 \\ 31136\text{-}61\text{-}5 \\ 410.50 \\ 2.098 \pm 0.576 \\ \text{No exp value} \end{array}$	4	HO H HO OH HO OAC	2,3-epoxide, 7-acetate	<i>L. ovalifolia, C. yunnanense, K. latifolia</i> (Li et al., 2013)
	1. Lyoniol B 2. (2S,3R,4R,6R,8S,9S,10R,11R,14S,15R)- 5,5,10,15-tetramethyl-7-oxapentacyclo [12.2.1.01,11.04,9.06,8]heptadecane- 2,3,4,10,15-pentol 3. Desacetyllyoniatoxin; desacetyllyoniol A	C ₂₀ H ₃₂ O ₆ 28894-74-8 368.46 -0.3 279-281°C	5	HO H HO OH HO OH	2,3-epoxide	<i>L. ovalifolia, C. yunnanense</i> (Li et al., 2013)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	 Mollfoliagein B 2β,3β-epoxy-5β,6β,10α,14β- tetrahydroxygrayan-16(17)-ene 3. 	C ₂₀ H ₃₀ O ₅ 350.45 209–211°C	4	HO HO HO HO	2,3-epoxide, 16- exomethylene	<i>R. molle</i> (Zhou et al., 2018)
	 Pierisformosin C Grayanotoxane-3,5,6,7,10,14,16-heptol, 14-propanoate, (3β,6β,7α,14R) 3. 	$\begin{array}{c} C_{23}H_{38}O_8 \\ 220796\text{-}29\text{-}2 \\ 442.54 \\ 0.728 \pm 0.549 \\ \text{No exp. value} \end{array}$	6		14-propionate	<i>P. formosa</i> (Li et al., 2013)
	 Pierisformosin D Grayanotoxane-3,5,6,7,10,14,16-heptol, 7-acetate 14-propanoate, (3β,6β,7α,14S) 3. 	$\begin{array}{c} C_{25}H_{40}O_9 \\ 217652\text{-}87\text{-}4 \\ 484.58 \\ 1.645 \pm 0.534 \\ \text{No exp. value} \end{array}$	5	HO H HO OH HO OACOEt	7-acetate, 14- propionate	<i>R. japonicum, R. molle</i> (Li et al., 2013), <i>P. formosa</i> (Li et al., 2017)
	 Pierisformosoid D . . 	$C_{20}H_{34}O_8$ 402.48 No exp. value	8			<i>P. formosa</i> (Niu et al., 2018)
	 Pierisformosoid F 3. 	$C_{25}H_{40}O_9$ 484.58 No exp. value	5	HO HO HO HO HO OAc	14-propionate, 15-acetate	<i>P. formosa</i> (Niu et al., 2018)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	 Pierisformosoid G 3. 	C ₂₀ H ₃₄ O ₉ 418.46 No exp. value	9			<i>P. formosa</i> (Niu et al., 2018)
	1. Pierisoid D	C ₂₃ H ₃₆ O ₈ 440.52 No exp. value	5	HO HHOOCOEt HO HHOOH	14-propionate, 15,16-epoxide	<i>P. formosa</i> (Li et al, 2017
	1. Pieristoxin G 2. (2S,3R,4R,6R,8S,9R,10S,11S,14S,15R,17S)- 5,5,10,15-tetramethyl-7-oxapentacyclo [12.2.1.01,11.04,9.06,8]heptadecane- 2,3,4,10,11,15,17-heptol 3. Bisdesacylasebotoxin VII	$\begin{array}{l} C_{20}H_{32}O_8 \\ 34206\text{-}60\text{-}5 \\ 400.46 \\ -0.570 \pm 0.669 \\ \text{No exp. value} \end{array}$	7	HO OH HO OH HO OH HO OH OH OH	2,3-epoxide	<i>P. japonica</i> (Li et al., 2013)
	 Pieristoxin H 2. 3. Desacetylkalmitoxin IV 	$\begin{array}{c} C_{20}H_{32}O_7 \\ 75899\text{-}53\text{-}5 \\ 384.46 \\ 0.465 \pm 0.615 \\ \text{No exp. value} \end{array}$	6	HO H HO OH HO OH OH	2,3-epoxide	<i>P. formosa</i> (Li et al., 2017), <i>P. japonica, C. yunnanense</i> (Li et al., 2013)
	1. Pieristoxin O 2. (1R,2S,3R,4R,6R,7R,8S,9R,10R,13R,14R, 16R)- 7-methoxy-5,5,9,14-tetramethyltetracyclo [11.2.1.01,10.04,8]hexadecane-2,3,4,6,9,14,16- heptol 3.	$\begin{array}{l} C_{20}H_{32}O_8 \\ 1836210\text{-}49\text{-}1 \\ 400.46 \\ -4.080 \pm 0.633 \\ \text{No exp. value} \end{array}$	7		7-ketone	<i>P. formosa</i> (Niu et al., 2016)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Pieristoxin Q 2. (1R,2S,3R,4R,6R,7R,8S,9R,10R,13R,14R, 16R)- 7-methoxy-5,5,9,14-tetramethyltetracyclo [11.2.1.01,10.04,8] hexadecane-2,3,4,6,9,14,16- heptol 3.	$\begin{array}{l} C_{21}H_{36}O_8 \\ 1836210\text{-}53\text{-}7 \\ 416.46 \\ -0.710 \pm 0.635 \\ \text{No exp. value} \end{array}$	7	MeQ HO H HO OH OH HO OH OH	2-methoxy	<i>Lyonia ovalifolia</i> (Zhang et al., 2020), <i>P. formosa</i> (Niu et al., 2016)
	 Rhododecorumin XII . . 	$\begin{array}{l} C_{20}H_{34}O_7 \\ 2230479\text{-}43\text{-}1 \\ 386.48 \\ -1.766 \pm 0.577 \\ \text{No exp. value} \end{array}$	7			<i>R. decorum</i> (Zhu et al, 2018)
	1. Rhodojaponin III 2. (1S,3R,4R,6R,8S,9S,10R,11R,14R,15R,17R)- 5,5,10,15-tetramethyl-7-oxapentacyclo [12.2.1.01,11.04,9.06,8]heptadecane- 3,4,10,15,17-pentol 3. Rhomotoxin, desacetylpieristoxin B	$\begin{array}{l} C_{20}H_{32}O_6\\ 26342\text{-}66\text{-}5\\ 368.46\\ 0.949\pm0.572\\ 285\text{-}287^\circ\text{C} \end{array}$	5	НО Н ОН ОН	2,3-epoxide	Lyonia ovalifolia (Zhang et al., 2020), <i>R. decorum, R. japonicum, R. molle, R. przewalski</i> (Li et al., 2013; Qjang et al., 2011; Wang et al., 2014; Zou et al., 2014)
	1. 14β-O-(2S,3R-nilyl)Rhodojaponin III 2. [(1S,3R,4R,6R,8S,9S,10R,11R,14R,15R,17R)- 3,4,10,15-tetrahydroxy-5,5,10,15-tetramethyl-7- oxapentacyclo[12.2.1.01,11.04,9.06,8] heptadecan-17-yl] (2S,3R)-3-hydroxy-2- methylbutanoate 3.	$\begin{array}{c} C_{25}H_{40}O_8 \\ 1118548\text{-}06\text{-}3 \\ 468.58 \\ 1.367 \pm 0.580 \\ \text{No exp. value} \end{array}$	4(5)	HO, H HO, H OH OR HO HO HO HO OR HO HO HO HO HO HO HO HO HO HO HO HO HO	14-nilylate	<i>C. henryi</i> (Li et al., 2013)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Rhodojaponin V 2. (3,4,10,15-tetrahydroxy-5,5,10,15-tetramethyl- 7-oxapentacyclo [12.2.1.01,11.04,9.06,8] heptadecan-17-yl) acetate 3.	$\begin{array}{l} C_{22}H_{34}O_7 \\ 37720\text{-}86\text{-}8 \\ 410.50 \\ 1.701 \pm 0.566 \\ \text{No exp. value} \end{array}$	4	HO H HO OH HO OAc OH	2,3-epoxide, 14- acetate	<i>R. japonicum</i> (Li et al., 2013), <i>R. molle</i> (Zhou et al., 2018)
	1. Rhodojaponin VI 2. (1S,3R,4R,6R,7R,8S,9R,10R,13R,14R,16R)- 5,5,9,14-tetramethyltetracyclo [11.2.1.01,10.04,8]hexadecane-3,4,6,7,9,14,16- heptol 3.	$\begin{array}{l} C_{20}H_{34}O_{7} \\ 37720\text{-}87\text{-}9 \\ 386.48 \\ -1.237 \pm 0.561 \\ \text{No exp. value} \end{array}$	7			<i>Lyonia ovalifolia</i> (Zhang et al., 2020), <i>P. japonica</i> (Zheng et al., 2019), <i>R. molle</i> (Li et al., 2013; Qjang et al., 2011; Zou et al., 2014)
	 2-Methoxyrhodojaponin VI 3. 	C ₂₁ H ₃₆ O ₇ 400.5 No exp. value	6	MeQ. HO H HO OH OH	2-methoxy	<i>P. japonica</i> (Zheng et al., 2019), <i>R. molle</i> (Li et al., 2020)
	1. Rhodomollein IX 2. (1R,3R,4R,6R,7R,8S,9R,10R,13S,16R)-5,5,9- trimethyl-14-methylidenetetracyclo [11.2.1.01,10.04,8]hexadecane-3,4,6,7,9,16-hexol 3.	$\begin{array}{l} C_{20}H_{32}O_6 \\ 798558\text{-}80\text{-}2 \\ 368.46 \\ -0.133 \pm 0.542 \\ 133\text{-}135^\circ\text{C} \end{array}$	6		16-exomethylene	<i>R. molle</i> (Li et al., 2013; Qjang et al., 2011)
	1. Rhodomollein X 2. (1S,3R,4R,6R,7R,8S,9R,10R,13S,16R)-5,5,9,14- tetramethyltetracyclo [11.2.1.01,10.04,8] hexadec-14-ene-3,4,6,7,9,16-hexol 3.	$\begin{array}{l} C_{20}H_{32}O_6 \\ 798558\text{-}81\text{-}3 \\ 368.46 \\ -0.133 \pm 0.542 \\ 223\text{-}224^\circ\text{C} \end{array}$	6		15,16-dehydro	<i>P. japonica</i> (Zheng et al., 2019), <i>R. molle</i> (Li et al., 2013; Qjang et al., 2011)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Rhodomollein XVI 2. [(1S,3R,4R,6R,7R,8S,9R,10R,13R,14R,16S) - 3,4,6,7,9,14-hexahydroxy-5,5,9,14-tetramethyl- 16-tetracyclo [11.2.1.01,10.04,8]hexadecanyl] acetate 3.	$\begin{array}{c} C_{22}H_{36}O_8\\ 292164\text{-}18\text{-}2\\ 428.5\\ 0.051\ \pm\ 0.555\\ \text{No\ exp.\ value} \end{array}$	6		14-acetate	<i>R. molle</i> (Li et al., 2013; Zou et al., 2014)
	1. Rhodomollein XVIII 2. (1S,3R,4R,6R,7S,8S,9R,10R,13R,14R,16S)- 5,5,9,14-tetramethyltetracyclo [11.2.1.01,10.04,8]hexadecane-3,4,6,7,9,14,16- heptol 3.	$\begin{array}{l} C_{20}H_{34}O_7 \\ 292164\text{-}20\text{-}6 \\ 386.48 \\ -1.237 \pm 0.561 \\ \text{No exp. value} \end{array}$	7			<i>R. molle</i> (Li et al., 2013; Qjang et al., 2011)
	1. 14β-O-(2S,3R-nilyl)Rhodomollein XVIII 2. [(1S,3R,4R,6R,7S,8S,9R,10R,13R,14R,16R) - 3,4,6,7,9,14-hexahydroxy-5,5,9,14-tetramethyl- 16-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] (2S,3R)-3-hydroxy-2-methylbutanoate 3.	$\begin{array}{l} C_{25}H_{42}O_9\\ 870785\text{-}29\text{-}8\\ 486.55\\ -0.283\pm0.569\\ 103\text{-}105^\circ\text{C} \end{array}$	6(7)	HO HO HO HO HO HO HO HO HO HO HO HO HO H	14-nilylate	<i>C. henryi</i> (Li et al., 2013)

(a): In this table all references are from Li et al., 2013, unless otherwise stated.

(b): Leucothoe grayana and not Lyonia grayana. Lyonia grayana is not an accepted name according to World Flora Online and probably a synonym of Leucothoe grayana (see http://www.worldfloraonline.org/search?query=Leucothoe+grayana and http://www.theplantlist.org/tpl1.1/record/kew-2349974).

82



Grayananes considered in this opinion

Table A.2: STRUCTURE ALERTS for relevant biological activity: (i) position 6β -OH and position 3β -OH or 2β , 3β -epoxide; or (ii) position 10β -CH₃ and position 3β -OH or 2β , 3β -epoxide; or (iii) position 6β -OH and position 10β -CH₃

No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Grayanotoxin II 2. (1S,3R,4R,6S,8S,10S,13R,14R,16R)-5,5,14- trimethyl-9-methylidenetetracyclo [11.2.1.01,10.04,8]hexadecane-3,4,6,14,16- pentol 3. $\Delta^{10(18)}$ -Andromedenol	$\begin{array}{c} C_{20}H_{32}O_5 \\ 4678\text{-}44\text{-}8 \\ 352.47 \\ 0.532 \ \pm \\ 0.484 \\ \text{No exp. value} \end{array}$	5	HO HO OH OH	10-exomethylene	L. grayana ^(b) (Iwasa et al., 1961; Kumazawa et al., 1961), Lyonia ovalifolia (Zhang et al., 2020), P. formosa (Li et al., 2017); P. japonica (Zheng et al., 2019), R. molle, R. micranthum, R. auriculatum, R. dauricum (Chai et al., 2020; Peng et al, 2020)
	1. Isograyanotoxin II 2. (1S,3R,4R,6S,8S,13R,14R,16R)-5,5,9,14- tetramethyltetracyclo[11.2.1.01,10.04,8] hexadec-9-ene-3,4,6,14,16-pentol 3.	$\begin{array}{c} C_{20}H_{32}O_5\\ 104459\text{-}10\text{-}1\\ 352.47\\ 0.426\ \pm\\ 0.488\\ 183\text{-}185^\circ\text{C} \end{array}$	5		9,10-dehydro	<i>L. grayana</i> ^(b) (Terai et al., 2000); <i>R. micranthum, R. dauricum</i> (Chai et al., 2020; Peng et al., 2020), <i>R. molle</i> (Li et al., 2019)
	 2α-O-methylgrayanotoxin II 3. 	$\begin{array}{l} C_{21}H_{34}O_6\\ 2171096787\\ 382.5\\ 0.594\ \pm\\ 0.547\\ \text{No\ exp.\ value} \end{array}$	5	MeO H H HO OH OH HO	2-methoxy, 10-exomethylene	<i>P. japonica</i> (Zheng et al., 2019)
	1. 16-O-Acetylgrayanotoxin II 2. [(15,3R,4R,6S,8S,10S,13R,14R,16R)- 3,4,6,16-tetrahydroxy-5,5,14-trimethyl-9- methylidene-14-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] acetate 3.	C ₂₂ H ₃₄ O ₆ 394.50 0.9 No exp. value	4	HO HO OH OH	10-exomethylene, 16-acetate	<i>R. micranthum</i> (Zhang et al., 2015)



No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	 1. 14-O-Acetylisograyanotoxin II 2. 3. 	C ₂₂ H ₃₄ O ₆ 394.50 No exp. value	4	HO OH OAC	9,10-dehydro, 14-acetate	<i>R. micranthum, R. dauricum</i> (Chai et al., 2020, Peng et al., 2020)
	1. 3-Epigrayanotoxin III 2. (3R,4R,6R,8S,9R,10R,14R,16S)-5,5,9,14- tetramethyltetracyclo[11.2.1.01,10.04,8] hexadecane-3,4,6,9,14,16-hexol 3.	$\begin{array}{c} C_{20}H_{34}O_6\\ 30460\text{-}38\text{-}9\\ 370.48\\ -0.237\pm\\ 0.514\\ \text{No exp. value} \end{array}$	6	HO HO HO HO HO HO HO HO HO HO HO HO HO H	3-epi-hydroxy	<i>L. grayana^(b)</i> (Hikino et al. 1976)
	1. Grayanotoxin IV 2. [(15,3R,4R,6S,10R,13R,14R,16R)-3,4,6,14- tetrahydroxy-5,5,14-trimethyl-9-methylidene- 16-tetracyclo [11.2.1.01,10.04,8]hexadecanyl] acetate 3. $\Delta^{10(18)}$ -Andromedenol, acetate; Acetylandromed-10(18)-enol; $\Delta^{10(18)}$ - acetylandromedenol	$\begin{array}{c} C_{22}H_{34}O_6\\ 30272\text{-}17\text{-}4\\ 394.50\\ 1.379\ \pm\\ 0.481\\ 177\text{-}179^\circ\text{C} \end{array}$	4	HO HO HO OAc	10-exomethylene, 14-acetate	<i>L. grayana</i> ^(b) (Li et al., 2013), <i>R. molle,</i> <i>R. micranthum, R. auriculatum,</i> <i>R. dauricum</i> (Chai et al., 2020; Peng et al, 2020)
	1. Grayanotoxin V 2. (3R,4R,8S,9R,10R,14R,16S)-3,4,9,14,16- pentahydroxy-5,5,9,14-tetramethyltetracyclo [11.2.1.01,10.04,8]hexadecan-6-one 3.	$\begin{array}{c} C_{20}H_{32}O_6\\ 30272\text{-}18\text{-}5\\ 368.46\\ -2.189\pm\\ 0.539\\ 230\text{-}232^\circ\text{C} \end{array}$	4		3-ketone	<i>L. grayana^(b)</i> (Li et al., 2013)
	1. Grayanotoxin VII 2. (1S,3R,4R,6S,8S,10S,13S,16R)-5,5,14- trimethyl-9-methylidenetetracyclo [11.2.1.01,10.04,8]hexadec-14-ene-3,4,6,16- tetrol 3. Principinol E	$\begin{array}{c} C_{20}H_{30}O_4\\ 30460\text{-}59\text{-}4\\ 334.45\\ 1.237\ \pm\\ 0.462\\ 185\text{-}187^\circ\text{C} \end{array}$	4	HO HO HO	10-exomethylene, 15,16-dehydro	<i>L. grayana</i> ^(b) (Li et al., 2013), <i>P. formosa</i> (Niu et al., 2018), <i>P. japonica</i> (Li et al., 2017), <i>R. molle</i> (Li et al., 2019), <i>R. micranthum</i> (Chai et al., 2020, Zhang et al., 2015)



No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)	
	1. Grayanotoxin VIII 2. (3R,4R,6S,8S,10S,16S)-5,5-dimethyl-9,14- dimethylidenetetracyclo[11.2.1.01,10.04,8] hexadecane-3,4,6,16-tetrol 3.	$\begin{array}{c} C_{20}H_{30}O_4\\ 30460\text{-}60\text{-}7\\ 334.45\\ 1.237\ \pm\\ 0.462\\ 193\text{-}196^\circ\text{C} \end{array}$	4	HO HO HO OH	10-exomethylene, 16-exomethylene	<i>L. grayana</i> ^(b) (Li et al., 2013), <i>R. molle,</i> <i>R. micranthum, R. auriculatum,</i> <i>R. dauricum</i> (Chai et al., 2020; Peng et al, 2020, Zhang et al., 2015)	
	1. Grayanotoxin IX 2. [(1S,3R,4R,6S,8S,10S,13S,16R)-3,4,6- trihydroxy-5,5,14-trimethyl-9-methylidene-16- tetracyclo[11.2.1.01,10.04,8]hexadec-14-enyl] acetate 3. Andromedienol acetate	$\begin{array}{c} C_{22}H_{32}O_5\\ 30460\text{-}58\text{-}3\\ 376.49\\ 2.129\ \pm\\ 0.467\\ \text{No\ exp.\ value} \end{array}$	3	HO OH OAc	10-exomethylene, 14-acetate, 15,16- dehydro	<i>L. grayana^(b)</i> (Li et al., 2013); <i>R. micranthum</i> (Zhang et al., 2015; Chai et al., 2020)	
	1. Grayanotoxin X 2. [(1R,3R,4R,6S,8S,10S,13S,16R)-3,4,6- trihydroxy-5,5-dimethyl-9,14-dimethylidene- 16-tetracyclo[11.2.1.01,10.04,8]hexadecanyl] acetate 3.	$\begin{array}{c} C_{22}H_{32}O_5\\ 36660\text{-}75\text{-}0\\ 376.49\\ 2.173~\pm\\ 0.467\\ \text{No~exp.~value} \end{array}$	3		10-exomethyene, 14-acetate, 16- exomethylene	<i>L. grayana</i> ^(b) (Li et al., 2013), <i>R. micranthum</i> (Chai et al., 2020, Zhang et al., 2015)	
	1. Grayanotoxin XI 2. (1S,3R,4R,6S,8S,10S,12R,13R,14R,16R)- 5,5,14-trimethyl-9-methylidenetetracyclo [11.2.1.01,10.04,8]hexadecane- 3,4,6,12,14,16-hexol 3.	$\begin{array}{c} C_{20}H_{32}O_6\\ 33880\mathchar`-98\mathchar`-98\mathchar`-735\mathchar`-0.735\mathchar`-0.503\\ No\ exp.\ value \end{array}$	6	HO HO HOH OH OH	10-exomethylene	<i>L. grayana^(b)</i> (Li et al., 2013)	
	 Grayanotoxin XII 1,1,8-trimethyl-4-methylidenedodecahydro- 4a,7-methanocyclopenta[b]heptalene- 2,5,8,11,11a,12(1h)-hexol 3. 	$\begin{array}{c} C_{20}H_{32}O_6\\ 35928\text{-}08\text{-}6\\ 368\text{-}46\\ -1.083\pm\\ 0.504\\ \text{No exp. value} \end{array}$	6		10-exomethylene	<i>L. grayana^(b)</i> (Li et al., 2013)	



No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Grayanotoxin XIII 2. [(1S,3R,4R,6S,8S,10S,12R,13R,14R,16R)- 3,4,6,12,14-pentahydroxy-5,5,14-trimethyl-9- methylidene-16-tetracyclo [11.2.1.01,10.04,8] hexadecanyl] acetate 3.	$\begin{array}{c} C_{22}H_{34}O_7\\ 35928\text{-}07\text{-}5\\ 410.50\\ 0.303\ \pm\\ 0.494\\ \text{No\ exp.\ value} \end{array}$	5	HO HO HO OAc OH	10-exomethylene, 14-acetate	<i>L. grayana^(b)</i> (Li et al., 2013)
	1. Grayanotoxin XIV 2. [(3R,4R,8S,9R,13S,14R)-3,4,9,14- tetrahydroxy-5,5,9,14-tetramethyl-6-oxo-16- tetracyclo[11.2.1.01,10.04,8]hexadecanyl] acetate 3.	$\begin{array}{c} C_{22}H_{34}O_7\\ 39012\text{-}12\text{-}9\\ 410.50\\ -1.437\pm\\ 0.533\\ \text{No exp. value} \end{array}$	4		3-ketone, 14-acetate	<i>R. molle, R. micranthum,</i> <i>R. auriculatum, R. dauricum</i> (Chai et al., 2020, Peng et al, 2020)
	1. Grayanotoxin XV 2. (1R,2R,4S,6R,7R,10S,12S,14S)-6,15,15- trimethyl-11-methylidene-16-oxapentacyclo [8.5.1.14,7.01,12.04,10]heptadecane-2,6,14- triol 3.	$\begin{array}{c} C_{20}H_{30}O_4\\ 39012\text{-}13\text{-}0\\ 334.45\\ 1.290\ \pm\\ 0.590\\ No\ exp.\ value\\ \end{array}$	3	HO HO HO	5,9-epoxide, 10-exomethylene	<i>L. grayana^(b)</i> (Li et al., 2013), <i>P. formosa</i> (Li et al., 2017)
	1. Grayanotoxin XVIII 2. (1R,3R,4R,6S,8S,10R,13R,14R)-5,5,14- trimethyl-9-methylidenetetracyclo [11.2.1.01,10.04,8]hexadecane-3,4,6,14-tetrol 3.	$\begin{array}{c} C_{20}H_{32}O_4 \\ 70474\text{-}76\text{-}9 \\ 336.47 \\ 1.669 \pm \\ 0.463 \\ \text{No exp. value} \end{array}$	4		10-exomethylene	L. grayana ^(b) (Li et al., 2013), P. formosa (Li et al., 2017); P. japonica (Zheng et al., 2019), R. molle, R. micranthum, R. auriculatum, R. dauricum (Chai et al., 2020; Peng et al., 2020)
	 12β-Hydroxygrayanotoxin XVIII 3. 	$\begin{array}{c} C_{20}H_{32}O_5\\ 2402790\text{-}40\text{-}1\\ 352.47\\ 0.126\ \pm\\ 0.476\\ 180\text{-}181 \end{array}$	5	HO HO HO HO HO HO HO HO HO HO HO HO HO H	10-exomethylene	<i>P. japonica</i> (Zheng et al., 2019)



No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	 1. 18-Hydroxygrayanotoxin XVIII 2. 3. 	C ₂₀ H ₃₂ O ₅ 352.47 No exp. value	5		10-exomethylene	<i>R. molle</i> (Zhou et al., 2018)
	1. Grayanotoxin XIX 2. (1R,3R,4R,6S,8S,10S,13R)-5,5,14-trimethyl- 9-methylidenetetracyclo [11.2.1.01,10.04,8] hexadec-14-ene-3,4,6-triol 3.	$\begin{array}{c} C_{20}H_{30}O_{3} \\ 75829\text{-}08\text{-}2 \\ 318.45 \\ 3.184 \ \pm \\ 0.448 \\ \text{No exp. value} \end{array}$	3		10-exomethylene, 15,16-dehydro	<i>L. grayana^(b)</i> (Li et al., 2013)
	 1. 17-Hydroxygrayanotoxin XIX 2. 3. 	$\begin{array}{c} C_{20}H_{30}O_4\\ 2402790\text{-}35\text{-}4\\ 334.45\\ 1.708\pm0.558\\ 154\text{-}155^\circ\text{C} \end{array}$	4		10-exomethylene, 15,16-dehydro	<i>P. japonica</i> (Zheng et al., 2019)
	1. Asebotoxin II 2. [(3R,4R,6S,8S,10S,14R,16S)-3,4,6,14- tetrahydroxy-5,5,14-trimethyl-9-methylidene- 16-tetracyclo[11.2.1.01,10.04,8]hexadecanyl] propanoate 3.	$\begin{array}{c} C_{23}H_{36}O_6\\ 23984\text{-}18\text{-}1\\ 408.53\\ 1.889\ \pm\\ 0.481\\ \text{No\ exp.\ value} \end{array}$	4	HO OH OCOEt	10-exomethylene, 14-propionate	<i>P. japonica</i> (Li et al., 2013)
	1. Asebotoxin IV 2. [(1R,2S,3R,4R,6S,8S,9R,10R,13R,14R,16R)- 2,4,6,9,14,16-hexahydroxy-5,5,9,14- tetramethyl-3-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] propanoate 3.	$\begin{array}{c} C_{23}H_{38}O_8\\ 33476\text{-}73\text{-}2\\ 442.54\\ 0.691\ \pm\\ 0.534\\ No\ exp.\ value \end{array}$	6	HO H H OH OH	6-propionate	<i>P. japonica</i> (Li et al., 2013)



No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Asebotoxin V 2. [(1R,2S,3R,4R,6S,8S,9R,10R,13R,14R,16R)- 3-acetyloxy-4,6,9,14,16-pentahydroxy- 5,5,9,14-tetramethyl-2-tetracyclo [11.2.1.01,10.04,8]hexadecanyl] propanoate 3.	$\begin{array}{c} C_{25}H_{40}O_9\\ 33476\text{-}74\text{-}3\\ 484\text{.}58\\ 1.873\ \pm\\ 0.523\\ \text{No\ exp.\ value} \end{array}$	5	HO H HO H HO H HO H HO H HO H HO H HO H	6-acetate, 7-propionate	<i>P. japonica</i> (Li et al., 2013)
	1. Asebotoxin VIII 2. [(1S,2S,3R,4R,6S,8S,9R,10R,13R,14R,16R)- 3-acetyloxy-2,4,6,9,14-pentahydroxy-5,5,9,14- tetramethyl-16-tetracyclo [11.2.1.01,10.04,8] hexadecanyl] propanoate 3.	$C_{25}H_{40}O_9$ 63529-89-5 484.58 No value No exp. value	5	HO HH HH OCOEt	6-acetate, 14-propionate	<i>P. japonica</i> (Li et al., 2013)
	 Craiobiotoxin I . . 	$\begin{array}{c} C_{20}H_{32}O_5\\ 864679\text{-}90\text{-}3\\ 352.47\\ 1.822\ \pm\\ 0.551\\ No\ exp.\ value \end{array}$	5	HO HO HO HO	10-exomethylene	<i>C. yunnanense</i> (Zhang et al., 2005); <i>R. auriculatum</i> (Sun et al., 2019)
	 Craiobiotoxin V . . 	$\begin{array}{c} C_{20}H_{32}O_5\\ 864679\text{-}93\text{-}6\\ 352.47\\ 0.649\ \pm\\ 0.509\\ 137\text{-}139^\circ\text{C} \end{array}$	5		10-exomethylene	<i>C. yunnanense</i> (Zhang et al., 2005), <i>P. japonica</i> (Zheng et al., 2019)
	1. Kalmitoxin II 2. (1R,2S,3R,4R,6S,8S,10S,13R,14R,16R)- 5,5,14-trimethyl-9-methylidenetetracyclo [11.2.1.01,10.04,8]hexadecane-2,3,4,6,14,16- hexol 3.	$\begin{array}{c} C_{20}H_{32}O_6 \\ 63529\mathchar`-90\mathchar`-80\mathchar`-90\mathchar^-90\mathchar`-90\mathchar`-90\mathchar`-90\$	6	HO HOH OH OH	10-exomethylene	<i>K. latifolia</i> (Li et al., 2013), <i>P. formosa</i> (Niu et al., 2018)



No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Kalmitoxin III 2. [(1R,2S,3R,4R,6S,8S,9R,10R,13R,14R,16R)- 2,4,6,9,14,16-hexahydroxy-5,5,9,14- tetramethyl-3-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] acetate 3.	$\begin{array}{c} C_{22}H_{36}O_8 \\ 72514\text{-}64\text{-}8 \\ 428.51 \\ 0.181 \pm \\ 0.534 \\ \text{No exp. value} \end{array}$	6		6-acetate	<i>K. latifolia</i> (Li et al., 2013)
	1. Kalmitoxin IV 2. [(1R,2S,3R,4R,6R,8S,9S,10R,11R,14R,15R, 17R)-2,4,10,15,17-pentahydroxy-5,5,10,15- tetramethyl-7-oxapentacyclo [12.2.1.01,11.04,9.06,8]heptadecan-3-yl] acetate 3.	$\begin{array}{c} C_{22}H_{34}O_8 \\ 72514\text{-}65\text{-}9 \\ 426.50 \\ 1.425 \pm \\ 0.591 \\ \text{No exp. value} \end{array}$	5	HO H HO H OH OH OH OH OH	2,3-epoxide, 6-acetate	<i>K. latifolia</i> (Li et al., 2013)
	1. Kalmitoxin V 2. [(1R,2S,3R,4R,6R,8S,9S,10R,11R,14R,15R, 17R)-3-acetyloxy-4,10,15,17-tetrahydroxy- 5,5,10,15-tetramethyl-7-oxapentacyclo [12.2.1.01,11.04,9.06,8]heptadecan-2-yl] acetate 3.	$\begin{array}{c} C_{24}H_{36}O_9 \\ 72514\text{-}66\text{-}0 \\ 468.5 \\ 2.246 \pm \\ 0.582 \\ \text{No exp. value} \end{array}$	4	HO H H OH ACO OAC OH OH	2,3-epoxide, 6,7-diacetate	<i>K. latifolia</i> (Li et al., 2013)
	1. Lyoniol D 2. [(1R,2S,3R,4R,6R,7R,8S,9R,10R,13R,14R)- 2,4,6,7,9,14-hexahydroxy-5,5,9,14- tetramethyl-3-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] acetate 3.	$\begin{array}{c} C_{22}H_{36}O_8\\ 38011\mathchar`-92\mathchar`-6\\ 428.5\\ 0.313\ \pm\\ 0.552\\ No\ exp.\ value \end{array}$	6		6-acetate	<i>Lyonia ovalifolia</i> (Li et al., 2013)
	 Pierisformosoid E 3. 	C ₂₀ H ₃₄ O ₈ 402.48 No exp. value	8		3-epi-hydroxy	<i>P. formosa</i> (Niu et al., 2018)



No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Pieristoxin F 2. [(2S,3R,4R,6S,8S,9R,14R,16R)-3-acetyloxy- 2,4,6,9,14-pentahydroxy-5,5,9,14-tetramethyl- 16-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] propanoate 3.	$\begin{array}{c} C_{25}H_{40}O_9\\ 56662-60-3\\ 484.58\\ 1.948\pm\\ 0.525\\ 190-195^\circ C\end{array}$	5	HO HH OCOEt	6-acetate, 14- propionate	<i>P. japonica</i> (Li et al., 2013)
	1. Pieristoxin J 2. [(1R,2S,3R,4R,6R,8S,9R,10S,11S,14R,15R, 17R)-2,4,10,11,15,17-hexahydroxy-5,5,10,15- tetramethyl-7-oxapentacyclo [12.2.1.01,11.04,9.06,8]heptadecan-3-yl] acetate 3.	$\begin{array}{l} C_{22}H_{34}O_9\\ 34183-49-8\\ 442.5\\ 0.636\ \pm\\ 0.645\\ No\ exp.\ value\\ \end{array}$	6	HO H OH OH OH OH OH OH OH	2,3-epoxide, 6- acetate	<i>P. japonica</i> (Li et al., 2013)
	1. Pieristoxin R 2. (1R,2S,3R,4R,6R,7R,8R,10S,13S,16R)- 5,5,14-trimethyl-9-methylidenetetracyclo [11.2.1.01,10.04,8]hexadec-14-ene- 2,3,4,6,7,16-hexol 3.	$\begin{array}{c} C_{20}H_{30}O_6 \\ 1836210\text{-}55\text{-}9 \\ 366.4 \\ 5\text{-}0.944 \pm \\ 0.549 \\ \text{No exp. value} \end{array}$	6		10-exomethylene, 15,16-dehydro	<i>P. formosa</i> (Niu et al., 2016)
	1. Pieristoxin S 2. (1R,3R,4R,6S,8S,10R,13S,14R)-5,5,14- trimethyl-9-methylidenetetracyclo [11.2.1.01,10.04,8]hexadecane-3,4,6,13,14- pentol 3.	$\begin{array}{l} C_{20}H_{32}O_5 \\ 1836210\text{-}57\text{-}1 \\ 352.47 \\ 0.479 \ \pm \\ 0.483 \\ \text{No exp. value} \end{array}$	5	HO HO HO HO	10-exomethylene	<i>P. formosa</i> (Niu et al., 2016), <i>P. japonica</i> (Zheng et al., 2019)
	1. Rhodojaponin I 2. [(1S,3R,4R,6R,8S,9S,10R,11R,14R,15R, 17S)-17-acetyloxy-4,10,15-trihydroxy- 5,5,10,15-tetramethyl-7-oxapentacyclo [12.2.1.01,11.04,9.06,8]heptadecan-3-yl] acetate 3.	$\begin{array}{c} C_{24}H_{36}O_8\\ 26116\text{-}88\text{-}1\\ 452.5\\ 2.337\ \pm\\ 0.557\\ 248.5\text{-}250^\circ\text{C} \end{array}$	3		2,3-epoxide, 6,14- diacetate	<i>R. japonicum</i> (Li et al., 2013), <i>R. molle</i> (Zhou et al., 2018; Zou et al., 2014)



No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Rhodojaponin II 2. [(15,3R,4R,6R,8S,9S,10R,11R,14R,15R, 17R)-4,10,15,17-tetrahydroxy-5,5,10,15- tetramethyl-7-oxapentacyclo [12.2.1.01,11.04,9.06,8]heptadecan-3-yl] acetate 3.	$\begin{array}{c} C_{22}H_{34}O_7\\ 26116892\\ 410.50\\ 1.595\ \pm\\ 0.560\\ 294.5296^\circ\text{C} \end{array}$	4		2,3-epoxide, 6- acetate	<i>R. japonicum</i> (Li et al., 2013); <i>R. molle</i> (Zhou et al., 2018)
	1. Rhodojaponin IV 2. [(1S,3R,4R,6S,8S,9R,10R,13R,14R,16R)-16- acetyloxy-4,6,9,14-tetrahydroxy-5,5,9,14- tetramethyl-3-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] acetate 3. 6-O-Acetylgrayanotoxin I	$\begin{array}{c} C_{24}H_{38}O_8\\ 30460\mathchar`$	4	HO HO ACO	6,14-diacetate	<i>R. formosum, R. japonicum</i> (Li et al., 2013)
	1. Rhodojaponin VII 2. [(15,3R,4R,6R,7R,8S,9R,10R,13R,14R, 16R)-16-acetyloxy-4,6,7,9,14-pentahydroxy- 5,5,9,14-tetramethyl-3-tetracyclo [11.2.1.01,10.04,8]hexadecanyl] acetate 3.	$\begin{array}{c} C_{24}H_{38}O_9\\ 38290\text{-}01\text{-}6\\ 470.6\\ 1.045\ \pm\\ 0.540\\ \text{No\ exp.\ value} \end{array}$	5		6,14-diacetate	<i>R. formosum, R. japonicum</i> (Li et al., 2013)
	1. Rhodomolin A 2. (1S,3R,4R,6R,7S,8R,10S,13R,14R,16R)-7- methoxy-5,5,14-trimethyl-9-methylidene- tetracyclo[11.2.1.01,10.04,8]hexadecane- 3,4,6,14,16-pentol 3.	$\begin{array}{c} C_{21}H_{34}O_6\\ 858940\text{-}89\text{-}3\\ 382.5\\ 0.594\ \pm\\ 0.547\\ \text{No\ exp.\ value} \end{array}$	5		2-methoxy, 10- exomethylene	<i>R. molle</i> (Li et al., 2013)
	1. Rhodomolin B 2. [(1S,3R,4R,6R,7S,8R,10S,13R,14R,16R)- 3,4,6,7,14-pentahydroxy-5,5,14-trimethyl-9- methylidene-16-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] acetate 3.	$\begin{array}{c} C_{22}H_{34}O_7\\ 858940\text{-}90\text{-}6\\ 410.50\\ 0.358\ \pm\\ 0.526\\ 282\text{-}284^\circ\text{C} \end{array}$	5	HO H H H H H H H H H H H H H H H H H H	10-exomethylene, 14-acetate	<i>R. molle</i> (Li et al., 2013)



No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)	
	1. Rhodomollein I 2. (3R,4R,6R,7S,8R,10S,14R,16S)-5,5,14- trimethyl-9-methylidenetetracyclo [11.2.1.01,10.04,8]hexadecane-3,4,6,7,14,16- hexol 3.	$\begin{array}{c} C_{20}H_{32}O_6 \\ 126223\text{-}68\text{-}5 \\ 368.46 \\ -0.628 \pm \\ 0.529 \\ \text{No exp. value} \end{array}$	6	HO H HO H	10-exomethylene	<i>P. japonica</i> (Zheng et al., 2019), <i>R. molle</i> (Li et al., 2013; Zou et al., 2014)	
	1. Rhodomollein III 2. Grayanotoxane-2,3,5,6,10,14,16-heptol,6- acetate, (2b,3b,6b,14R)- 3.	$\begin{array}{c} C_{22}H_{36}O_8 \\ 142060\text{-}05\text{-}7 \\ 428.5 \\ 0.013 \ \pm \\ 0.543 \ \text{No exp.} \\ \text{value} \end{array}$	7		6-acetate	<i>R. molle</i> (Li et al., 2013, Zou et al., 2014)	
	1. Rhodomollein XI 2. [(1S,3R,4R,6R,7R,8S,9R,10R,13R,14R, 16R)-4,6,7,9,14,16-hexahydroxy-5,5,9,14- tetramethyl-3-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] acetate 3.	$\begin{array}{c} C_{22H_{36}O_8} \\ 798558$	6		6-acetate	<i>R. molle</i> (Zhou et al., 2018)	
	1. Rhodomollein XIII 2. [(1S,3R,4R,6S,8S,9R,10R,13R,14R,16R)- 4,6,9,14,16-pentahydroxy-5,5,9,14- tetramethyl-3-tetracyclo[11.2.1.01,10.04,8] hexadecanyl] acetate 3. 6-O-Acetylgrayanotoxin III	$\begin{array}{l} C_{22}H_{36}O_7 \\ 54781\text{-}72\text{-}5 \\ 412.5 \\ 0.809 \pm \\ 0.499 \\ > 255^\circ\text{C} \end{array}$	5		6-acetate	<i>R. molle</i> (Li et al., 2013)	



No	Name: 1. Trivial2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Rhodomollein XIX 2. (1S,3R,4R,6R,7R,8R,10S,13S,16R)-5,5,14- trimethyl-9-methylidenetetracyclo [11.2.1.01,10.04,8]hexadec-14-ene- 3,4,6,7,16-pentol 3.	$\begin{array}{c} {\sf C}_{20}{\sf H}_{30}{\sf O}_5 \\ {\sf 629615}\text{-}33\text{-}4 \\ {\sf 350.45} \\ {\sf No\ exp\ value} \\ {\sf 0.158\ \pm} \\ {\sf 0.508} \end{array}$	5		10-exomethylene, 15,16-dehydro	<i>P. formosa</i> (Niu et al., 2018), <i>P. japonica</i> (Zheng et al., 2019), <i>R. molle</i> (Li et al., 2013)

(a): In this table all references are from Li et al., 2013, unless otherwise stated.
(b): Leucothoe grayana and not Lyonia grayana for which the name is unresolved. Some sources state that Lyonia grayanane may be a synonym of Leucothoe grayanane (see http://www.theplantlist.org/tpl1.1/record/kew-2349974).



Grayananes considered in this opinion

Table A.3: STRUCTURE ALERTS: (i) Position 10β -CH₃ or (ii) position 6β -OH or (iii) position 3β -OH or (iv) still to be classified

No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	 1-Epigrayanotoxin II (1S,3R,4R,6S,8R,10S,13R,14R,16R)-5,5,14-trimethyl-9- methylidenetetracyclo [11.2.1.01,10.04,8]hexadecane- 3,4,6,14,16-pentol Principinol D 	C ₂₀ H ₃₂ O ₅ 352.47 239–240°C	5		epi-grayanane structure, 10-exomethylene	<i>R. principis, R. micranthum</i> (Li et al., 2019)
	1. 1-Epigrayanotoxin IV 2. [(15,3R,4R,6S,8R,10S,13R,14R,16R)-3,4,6,14- tetrahydroxy-5,5,14-trimethyl-9-methylidene-16-tetracyclo [11.2.1.01,10.04,8]hexadecanyl] acetate 3.	C ₂₂ H ₃₄ O ₆ 394.50 0.9 239–240°C	4		epi-grayanane structure, 10-exomethylene, 14-acetate	<i>R. micranthum</i> (Zhang et al., 2015)
	1. Grayanotoxin XVI2. [(3R,4R,6S,8S,10S,14R,16S)- 4,6,14,16-tetrahydroxy-5,5,14-trimethyl-9-methylidene-3- tetracyclo[11.2.1.01,10.04,8]hexadecanyl] acetate 3. 6-O-Acetylgrayanotoxin II	$\begin{array}{l} C_{22}H_{34}O_6 \\ 59236\text{-}87\text{-}2 \\ 394.50 \\ 1.353 \pm 0.472 \\ \text{No exp. value} \end{array}$	4		6-acetate, 10- exomethylene	R. molle, R. micranthum, R. auriculatum, R. dauricum (Peng et al, 2020)
	1. Grayanotoxin XVII2. (1R,4R,8S,9R,10R,13R,14R,16R)- 4,9,14,16-tetrahydroxy-5,5,9,14-tetramethyltetracyclo [11.2.1.01,10.04,8]hexadecane-3,6-dione 3.	$\begin{array}{l} C_{20}H_{30}O_6\\ 59740\mathchar`-2.558 \pm 0.584\\ \text{No exp. value} \end{array}$	4		3,6-diketone	<i>L. grayana^(b)</i> (Li et al., 2013)
	 1-Epigrayanotoxin XVIII 2. 3. Pierisformosin A 	C ₂₀ H ₃₂ O ₄ 336.47 No exp. value	4		epi-grayanane structure 10-exomethylene	<i>P. formosa</i> (Li et al., 2013) <i>P. japonica</i> (Zheng et al., 2019)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	 Grayanotoxin XX 3. 	$\begin{array}{c} C_{20}H_{28}O_4 \\ 78534\text{-}58\text{-}4 \\ 332.43 \\ -0.967 \pm 0.544 \\ \text{No exp. value} \end{array}$	3		3-ketone, 10- methylene, 15,16- dehydro	
	 Grayanotoxin XXI 3. 	$\begin{array}{c} C_{22}H_{34}O_6 \\ 850494\text{-}13\text{-}2 \\ 394.50 \\ 1.247 \pm 0.502 \\ 150\text{-}152 \ ^\circ\text{C} \end{array}$	3		1,5-seco- grayanane structure	<i>R. micranthum</i> (Zhang et al., 2015)
	 Grayanotoxin XXII . . 	$\begin{array}{c} C_{23}H_{36}O_{6} \\ 1252585866 \\ 408.53 \\ 1.757 \pm 0.502 \\ \text{No exp. value} \end{array}$	3	HO HO HO R = OCOCH ₂ CH ₃	1,5-seco- grayanane structure	<i>P. formosa</i> (Li et. al., 2017)
	1. Craiobiotoxin VII 2. 3.	$\begin{array}{c} C_{20H_{34}O_6} \\ 864679\text{-}95\text{-}8 \\ 370.48 \\ 0.172 \pm 0.539 \\ 165\text{-}167 \ ^\circC \end{array}$	6		epi-grayanane structure	C. yunnanense (Li et al., 2013)



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	 Craiobiotoxin VIII . . 	$\begin{array}{l} C_{20}H_{32}O_5 \\ 864680\text{-}12\text{-}6 \ 352.47 \\ 0.649 \ \pm \ 0.509 \\ \text{No exp. value} \end{array}$	5	HO H HO H HO HOH	epi-grayanane structure, 10-exomethylene	<i>C. yunnanense</i> (Zhang et al., 2005), <i>P. japonica</i> (Zheng et al., 2019), <i>R. micranthum</i> (Zhang et al., 2015), <i>R. molle</i> (Li et al., 2019)
	1. Grayathol A 2. (1R,3R,4R,6S,8R,10S,13R)-5,5,14-trimethyl-9- methylidenetetracyclo [11.2.1.01,10.04,8]hexadec-14-ene- 3,4,6-triol 3.	$\begin{array}{c} C_{20}H_{30}O_{3} \\ 71493\text{-}02\text{-}2 \\ 318.45 \\ 3.184 \pm 0.448 \\ \text{No exp. value} \end{array}$	3		epi-grayanane structure, 10-exomethylene, 15,16-dehydro	<i>L. grayana</i> ^(b) (Li et al., 2013)
	1. Kalmanol 2. (1R,2R,3S,5S,7R,8R,10S,12R,13R)-2,6,6,12- tetramethyltetracyclo [8.5.1.03,7.013,16]hexadecane- 2,5,7,8,10,12-hexol 3.	$\begin{array}{c} C_{20H_{34}O_6} \\ 118997\text{-}92\text{-}5 \\ 370.48 \\ 0.010 \pm 0.461 \\ \text{No exp. value} \end{array}$	6		kalmane structure	K. angustifolia, R. molle (Li et al., 2013; Qjang et al., 2011); R. micranthum (Zhang et al., 2015)
	1. Leucothol A 2. (1R,3R,6S,8S,10R,13R,14R)-6,14-dihydroxy-5,5,14- trimethyl-9-methylidene-tetracyclo[11.2.1.01,10.03,8] hexadecan-4-one 3.	$\begin{array}{l} C_{20}H_{30}O_{3} \\ 39012\text{-}11\text{-}8 \\ 318.4 \\ 1.150 \pm 0.506 \\ \text{No exp. value} \end{array}$	2		leucothane structure, 4- ketone	L. grayana ^(b) , C. yannanense (Li et al., 2013, Wang et al., 2014)
	1. Leucothol B 2. (1R,3S,4S,6S,8R,10S,13R,14R,16R)-5,5,14-trimethyl-9- methylidenetetracyclo [11.2.1.01,10.03,8]hexadecane- 3,4,6,14,16-pentol 3.	$\begin{array}{l} C_{20}H_{32}O_5 \\ 38302\text{-}26\text{-}0 \\ 352.4 \\ 0.485 \pm 0.484 \\ \text{No exp. value} \end{array}$	5		leucothane structure	<i>L. grayana^(b)</i> (Li et al., 2013)

EFSA Journal 2023;21(3):7866



No	Name: 1. Trivial 2. IUPAC 3. Synonyms	Elem. Comp CAS number MW LogP (25°C) mp (°C)	No of OH	Chemical structure	Remark	Examples of occurrence in Ericaceae ^(a)
	1. Rhodomollein XII 2. [(15,3R,4R,6R,7R,8R,10S,13R,14R,16R)-4,6,7,14,16- pentahydroxy-5,5,14-trimethyl-9-methylidene-3-tetracyclo [11.2.1.01,10.04,8] hexadecanyl] acetate 3.	$\begin{array}{c} C_{22}H_{34}O_7 \\ 798558\text{-}83\text{-}5 \\ 410.50.281 \pm 0.514 \\ 7577^\circ\text{C} \end{array}$	5	HO H H OH OH OH	6-acetate, 10- exomethylene	<i>R. molle</i> (Li et al., 2013; Chen et al., 2004)
	1. Rhodomollein XIV 2. (1R,2R,3S,4R,5R,7R,8R,10S,12R,13R,16R)-2,6,6,12- tetramethyltetracyclo [8.5.1.03,7.013,16]hexadecane- 2,4,5,7,8,10,12-heptol 3.	$\begin{array}{c} C_{20H_{34}O_7} \\ 798558\text{-}84\text{-}6 \\ 386.48 \\ \mathbf{-1.083} \pm 0.512 \\ 133\text{-}\mathbf{135^\circ C} \end{array}$	7		kalmane structure	<i>R. molle</i> (Li et al., 2013)
	1. Rhodomollein XV 2. (1R,2R,3S,7R,8R,10S,12R,13R,16R)-2,7,8,10,12- pentahydroxy-2,6,6,12-tetramethyltetracyclo [8.5.1.03,7.013,16]hexadecan-5-one 3.	$\begin{array}{l} C_{20}H_{32}O_{6} \\ 292164\text{-}17\text{-}1 \\ 368.5 \\ -1.897 \pm 0.489 \\ \text{No exp. value} \end{array}$	5		kalmane structure, 2-keto	<i>R. molle</i> (Zhong et al., 2005)
	1. Rhodomollein XVII 2. (1R,3R,4R,6S,8R,9S,10R,13R,14R)-5,5,9,14- tetramethyltetracyclo [11.2.1.01,10.04,8]hexadecane- 3,4,6,9,14-pentol 3.	$\begin{array}{c} C_{20}H_{34}O_5\\ 292164\text{-}19\text{-}3\\ 354.48\\ 0.906 \pm 0.491\\ \text{No exp. value} \end{array}$	5		epi-grayanane structure	<i>R. molle</i> (Li et al., 2013; Qjang et al., 2011)

(a): In this table all references are from Li et al., 2013, unless otherwise stated.

(b): Leucothoe grayana and not Lyonia grayana for which the name is unresolved. Some sources state that Lyonia grayanane may be a synonym of Leucothoe grayanane (see http://www. theplantlist.org/tpl1.1/record/kew-2349974).

Appendix B – Other toxicological studies in experimental animals

Table B.1: Summary of acute toxicity (LD₅₀) of naturally occurring grayananes of potential concern (Route of exposure other than oral or *i.p.*). *Compounds are ranked on the table based on their potency*

Compound				LD ₅₀		
Trivial name	Origin	Species/ strain/sex	Route	(mg/kg bw)	Reference	
GTX I	R. hymenanthes	Mice, M&F	Subcutaneous	0.15 ± 0.01	Kohanawa ^(a) (1956) [§]	
Rhodojaponin III	P. japonica	Mice, M	Subcutaneous	0.65	Meguri (1959), Scott (1971)	
GTX I	R. hymenanthes	Frog	Into lymphsack	$\textbf{3.9} \pm \textbf{0.12}$	Kohanawa (1956) [§]	
Asebotoxin III	P. japonica	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	0.02	Hotta et al. $(1980)^{(b),\$}$	
Asebotoxin I	P. japonica	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	0.04	Hotta et al. $(1980)^{(b),\$}$	
Lyoniol B	L. ovalifolia	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	0.09	Hotta et al. $(1980)^{(b),\$}$	
Rhodojaponin III	R. japonicum	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	0.14	Hotta et al. (1980) ^{(b),§}	
GTX III	L. grayana	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	0.4	Hotta et al. $(1980)^{(b),\$}$	
Lyoniol A	L. ovalifolia	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	0.4	Hotta et al. (1980) ^{(b),§}	
Pieristoxin G	P. japonica	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	1.0	Hotta et al. (1980) ^{(b),§}	
GTX I	L. grayana	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	1.3	Hotta et al. (1980) ^{(b),§}	
Rhodojaponin IV ^(c)	L. grayana	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	2.5	Hotta et al. (1980) ^{(b),§}	
GTX IV	L. grayana	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	3.1	Hotta et al. (1980) ^{(b),§}	
Rhodojaponin I	R. japonicum	Guinea Pig, Hartley, M	<i>i.v</i> . (dorsalis penis)	3.1	Hotta et al. $(1980)^{(b),\$}$	

[§]: Isolated from plant material or semi-synthesised, purity not provided.

(a): To derive an LD50, Kohanawa (1956) exposed mice subcutaneously to rhodotoxin isolated from *R. hymenanthes* and andromedotoxin from *P. japonica* (synonym: *Andromeda japonica*). The author used the names rhodotoxin and andromedotoxin, both synonyms of GTX I, to identify two different substances that actually provide two different LD50s. Based on the melting point of rhodotoxin and andromedotoxin, as reported by the authors, it is possible to identify rhodotoxin as GTX I (data reported in the table) while the nature of andromedotoxin remains unidentified.

(b): 1 h.

(c): Synonym of 6-acetyl-GTX I.



Table B.2:	Reports on sin	gle dose toxicity	studies with Eric	caceae honey and	l other matrices al	llowing to estimate	the intake of grayananes
		, ,		,		5	5,

Animals	Type of honey as described (geographical and botanical origin when reported)	Concentration of grayananes in honey or other matrices	Exposure and doses of honey or other matrices administered	Estimated single doses of grayananes	Findings	Reference
Rat, Sprague Dawley ^(a) Female (250–300 g) 6-8 Mo n = 12 <i>per</i> experimental group	Rhododendron honey from beekeepers, mixture of 6 samples (≥ 45% Rhododendron pollen, contains also pollen from Apiaceae, Asteraceae, Fabaceae, Fagaceae (<i>Castanea sativa</i>), Rosaceae collected in June-July 2012 from different parts of Black Sea region of Turkey	GTX III: between 2.11 and 11.4 mg/kg (average 6.225 mg/kg; n = 6 different <i>Rhododendron</i> honey samples)	Gavage 0.1; 0.5; 2.5 g/ kg bw 1h	Average GTX-III: 0.62; 3.11; 15.6 μg/kg bw	Oxidative stress Dose dependent decrease in activity of CAT, GSH-Px, SOD coupled to an increase of malondialdehyde (MDA) in erythrocytes, liver, kidney, lung, heart, spleen, testicle, epididymis and brain. The number of affected enzymes is dependent on <i>Rhododendron</i> honey dose and tissue type. 2.5 g/kg bw <i>Rhododendron</i> honey was able to affect all the tested enzymes in all analysed tissues. 0.1 g/kg bw <i>Rhododendron</i> honey decreased CAT, GSH-Px and SOD only in erythrocytes and kidney No histopathology is reported	Sibel ^(b) et al. (2014)
Wistar rats Male (150–200 g) n = 10 control, 12 treated group	'Mad honey', Black Sea region, from a beekeeper	GTX I: 34.05 mg/kg GTX III: 6.5 mg/kg	Gavage 12.5 g/kg bw 24 h	GTX I: 429 µg/ kg bw GTX III: 81.9 µg/kg bw	Increased MDA, NO and decreased CAT in liver, kidney, brain, testes, heart, and plasma or erythrocytes. Decreased SOD in liver, kidney, brain (increased in testes, no effect in heart and erythrocytes) Increased GSH-Px in liver and decrease in kidney, brain, testes, heart (no effect in erythrocytes) Increased serum 4-hydroxynonenal No histopathology is reported	Eraslan et al. (2018)



Animals	Type of honey as described (geographical and botanical origin when reported)	Concentration of grayananes in honey or other matrices	Exposure and doses of honey or other matrices administered	Estimated single doses of grayananes	Findings	Reference
Rat, Sprague Dawley Female (250–300 g) 6-8 Mo n = 12 <i>per</i> experimental group	Rhododendron honey from beekeepers, (58.3% Rhododendron pollen, contains also pollen from Apiaceae, Asteraceae, Fabaceae, Fagaceae (Castanea sativa), Rosaceae collected in June-July 2012 from different parts of Black Sea region of Turkey	GTX III: 5.9 mg/kg	administered Gavage 0.1; 0.5; 2.5 g/ kg bw 1 h	GTX III: 0.59; 2.95; 14.7 μg/kg bw	Clinical biochemistry 0.1 and 0.5 g/kg bw: significant reduction in blood phosphorous (no effect. at 2.5 g/kg bw) 2.5 g/kg bw and 0.5 g/kg bw: significant increase of blood urea nitrogen (BUN) and T-bilirubin, respectively No effect on blood levels of glucose, triglyceride, cholesterol, tot. protein, ALT, AST, ALP, LDH, uric acid, Ca, Na, Cl, mg, K, creatinine, HDL cholesterol, LDL cholesterol, CPK, pseudocholinesterase Kidney histology: normal structure with minimal changes in proximal and distal tubules (no control sections are provided) No liver histopathology Summary: suggested kidney toxicity due to BUN increase at the highest dose, T-bilirubin may	Silici ^(b) et al. (2016)
					only parameter altered among liver function tests (e.g. ALT, AST, ALP, tot. protein)	



Animals	Type of honey as described (geographical and botanical origin when reported)	Concentration of grayananes in honey or other matrices	Exposure and doses of honey or other matrices administered	Estimated single doses of grayananes	Findings	Reference
Mice Male (20–25 g) 8-12 weeks n = 12 <i>per</i> experimental group	Rhododendron honey, Duzce, western Black Sea region of Turkey	Analysed by Kaplan et al. (2014) GTX I: 18.4 mg/kg GTX III: 7.4 mg/kg	Gavage 25; 50; 75 mg/kg bw 24 h	GTX I: 0.46- 0.92; 1.38 μg/kg bw GTX III: 0.185; 0.37; 0.555 μg/ kg bw	<i>Liver:</i> Analysed different parameters of liver injury 25 mg/kg bw: no effect (congestion and steatosis occurs at 48 h after repeated exposure) 50 mg/kg bw: significant congestion and steatosis (5-33%) no effect on necrosis, sinusoidal dilatation, and inflammation (sinusoidal dilatation, and inflammation occurs at 48 h after repeated exposure) 75 mg/kg bw: significant congestion, steatosis (5-33%), sinusoidal dilatation and scattered presence of inflammatory cells	Kukner et al. (2016)
Mice Male (20–25 g) 8-12 weeks n = 6 <i>per</i> experimental group	<i>Rhododendron</i> honey from beekeepers (96% <i>R. ponticum</i> pollen), Duzce, western Black Sea region of Turkey	GTX I: 32 mg/kg GTX III: 8 mg/kg	Gavage 25; 50; 75 mg/kg bw 24 h	GTX I: 0.8; 1.6; 2.4 µg/kg bw GTX III: 0.2; 0.4; 0.6 µg/kg bw	<i>Liver</i> : All doses decreased C–O/protein and C–O/lipid ratios significantly in liver. No dose response. No effect on the other parameters affected by 75 mg/kg 75 mg/kg: decrease in CH2/lipid and carbonyl/lipid ratios and increase in CH3/lipid ratio, interpreted as sign of lipid peroxidation. Ratio of Amide I/ Amide II decrease indicate disturbance of conformation and structure of the proteins. Changes in lipid/protein ratio No histopathology is reported Summary: an altered energy metabolism is suggested	Cakmak-Arslan et al. (2020a)



Animals	Type of honey as described (geographical and botanical origin when reported)	Concentration of grayananes in honey or other matrices	Exposure and doses of honey or other matrices administered	Estimated single doses of grayananes	Findings	Reference
Mice Male (20–25 g) 8-12 weeks n = 6 <i>per</i> experimental group	<i>Rhododendron</i> honey from beekeepers (≥ 45% <i>R. ponticum</i> pollen), Duzce, western Black Sea region of Turkey	GTX I: 32 mg/kg GTX III: 8 mg/kg	Gavage 25; 50; 75 mg/kg bw 24 h	GTX I: 0.8; 1.6; 2.4 µg/kg bw GTX III: 0.2; 0.4; 0.6 µg/kg bw	Heart: 25 mg/kg: increase in the unsaturated/saturated fatty acid ratio with no significant effect on the amount of unsaturated and saturated lipids 50 mg/kg: increase amount of unsaturated and saturated lipids and unsaturated/saturated fatty	Cakmak-Arslan et al. (2020b)
					acid ratio No histopathology is reported Summary: lipid metabolism disturbance and lipids accumulation	
Sprague Dawley rats Female (200–250 g) n = 4 per experimental group	'Mad honey', Duzce, Turkey from a beekeeper	GTX III: 39.9 \pm 0.02 mg/kg	Gavage 0.3; 0.6; 1.2; 2.4 g/kg bw day 8 days	GTX III: 12; 24; 48 and 96 µg/kg bw	Dose dependent increase in AST, ALT, LDH, ALP, CK. CK-MB, in the absence of altered Na ⁺ , K ⁺ , Cl ⁻ No histopathology is reported	Sahin (2016)
Sprague Dawley rats Male (250–300 g) Control group: n = 6 for ventricular function assessment; n = 8 for biochemical analysis Each treated group: n = 8 for ventricular function assessment; n = 12 for biochemical analysis	<i>Rh. Mollis</i> Flos (RMF) extract	Mixture containing 0.37% rhodojaponin I, 0.53% rhodojaponin II, 0.11% rhodojaponin III	Oral 21 and 113 mg/ kg bw	Rhodojaponin I: 0.08 and 0.42 mg/kg bw Rhodojaponin II: 0.11 and 0.6 mg/kg bw Rhodojaponin III: 0.02 and 0.12 mg/kg bw	High dose: vomiting, muscle rigidity, convulsion and death of 6 out of 20 rats Both doses: heart rate, left ventricular systolic and diastolic pressure decrease; plasma increase in LDH, CK-MB and AST No histopathology is reported	Dong et al. (2014)

(a): Data in the figures in the study refer to Wistar rats. EFSA contacted the study authors who clarified that the study was performed with Sprague Dawley rats. (b): Sibel and Silici are the same author.

Appendix C – BMD analysis

The text below describes the benchmark dose (BMD) analysis for the variation of heart rate following *i.p.* injection of GTX III in rats (Turkmen et al., 2013), using model averaging. BMD analysis was performed according to the EFSA guidance (EFSA Scientific Committee, 2017a and 2022).

Selection of the BMR

The BMR (benchmark response) used is a 10% change in mean response compared to the controls. The BMD (benchmark dose) is the dose corresponding with the BMR of interest.

A 90% confidence interval around the BMD was estimated, the lower bound is reported by BMDL and the upper bound by BMDU.

Software Used

Results were obtained using the EFSA web-tool for BMD analysis, which uses the R-package PROAST, version 70.0.

C.1. Specification of Deviations from Default Assumptions

Dose-response models

Default set of fitted models:

Model	Number of parameters	Formula
Null	1	y = a
Full	no. of groups	y = group mean
Exp model 3	3	$y = a \cdot exp(bx^d)$
Exp model 4	4	$y = a \cdot \left(c - (c - 1)exp(-bx^d)\right)$
Hill model 3	3	$\mathbf{y} = \mathbf{a} \cdot \left(1 - \frac{\mathbf{x}^{d}}{\mathbf{b}^{d} + \mathbf{x}^{d}} \right)$
Hill model 4	4	$y = a \cdot \left(1 - \frac{(c-1) \cdot x^d}{b^d + x^d}\right)$
Inverse Exponential	4	$y = a \cdot \left(1 + (c - 1)exp(-bx^{-d})\right)$
Log-Normal Family	4	$\textbf{y} = \textbf{a} \cdot (\textbf{1} + (\textbf{c-1}) \Phi(\textbf{lnb} + \textbf{dlnx}))$

Procedure for selection of BMDL



Flowchart for selection of BMDL

Data description

The endpoint to be analysed is the variation in heart rate (bpm) Data used for analysis:

GTX III dose (μ g/kg bw per day)	Heart rate (bpm)	Std Error	Number of animals
0	363	28	6
179	322	35	6
358	229	4	6
716	191	20	6

C.2. Final BMD values

	BMDL ₁₀ (µg/kg bw)	BMDU ₁₀ (µg/kg bw)
Results at 60 min of exposure	15.3	261

C.3. Confidence intervals for the BMD are based on 200 bootstrap data sets

C.3.1. Fitted Models

model	converged	loglik	npar	AIC
full model	yes	6.10	5	-2.20
null model	yes	-6.58	2	17.16
Expon. m3-	yes	5.27	4	-2.54
Expon. m5-	yes	6.10	5	-2.20
Hill m3-	yes	5.28	4	-2.56
Hill m5-	yes	6.10	5	-2.20
Inv.Expon. m3-	yes	5.43	4	-2.86
Inv.Expon. m5-	yes	5.68	5	-1.36
LN m3-	yes	5.36	4	-2.72
LN m5-	yes	5.83	5	-1.66

C.3.2. Estimated Model Parameters

EXP

estimate for var-: 0.03773 estimate for a-: 361.3 estimate for CED-: 77.41 estimate for d-: 0.8478

HILL

estimate for var-: 0.03772 estimate for a-: 361.3 estimate for CED-: 77.74 estimate for d-: 0.8527

INVEXP

estimate for var-: 0.03723 estimate for a-: 360.5 estimate for CED-: 96.56 estimate for d-: 0.1883

LOGN

estimate for var-: 0.03744 estimate for a-: 360.9 estimate for CED-: 88.41 estimate for d-: 0.3216

C.3.3. Weights for Model Averaging

EXP	HILL	INVEXP	LOGN
0.23	0.24	0.27	0.26



efsa JOURNAL



Appendix D – Uncertainty analysis

The purpose of the uncertainty analysis was to identify and quantify sources of uncertainty affecting the risk assessment and combine them to assess the overall certainty of the final conclusions, as recommended in EFSA's guidance on uncertainty analysis (EFSA, 2018a).

In a first step, sources of uncertainties related to hazard identification and characterisation of grayananes and consumption of honey containing grayananes in the EU were listed and discussed.

Hazard identification for genotoxic effects followed EFSA's guidance for genotoxicity assessment (EFSA Scientific Committee, 2011, 2017b), which can be considered as a standardised procedure for the purposes of uncertainty analysis (Section 3 in EFSA Scientific Committee, 2018a). As there was a positive result from a valid *in vivo* MN study with GTX III and *Rhododendron* honey, the Panel concluded that these substances should be considered as *in vivo* genotoxins (and potential genotoxic carcinogens) and therefore an HBGV cannot be established. The Panel identified no non-standard uncertainties affecting these conclusions, so no further uncertainty analysis was required for them (Section 3.1 in EFSA Scientific Committee, 2018a).

The assessment of acute toxicity and risk were affected by major uncertainties, which are assessed in detail in the following sections. In summary, the Panel quantified by expert judgement the impact of the identified uncertainties on the RP for acute toxicity for the sum of GTX I and III, the relative potency of other grayananes and the P95 of honey consumption for EU toddlers and adults. These judgements were elicited by semi-formal structured methods of Expert Knowledge Elicitation (semi-formal EKE, Annex B.8 of EFSA (2018b)). The protocol for the EKE including detailed methods and results is presented in Annex D of this opinion.

The quantified uncertainties affecting the RP for acute toxicity and P95 consumption were then combined by probability bounds analysis (section 14.1 of EFSA, 2018a) to quantify the uncertainty of the MOEs calculated by the Panel for the exposure at selected concentrations for the sum of GTX I and GTX III based on acute consumption from the EFSA database. The Panel also *calculated highest concentrations* for honey not resulting in acute effects based on the RP for acute toxicity, an MOE of 100 and data on acute consumption (i) from the EFSA database, and (ii) from consumption of 'certain honey' as folk medicine.

The assessed impact of the uncertainties was quantified using % probability, both when making the expert judgements and in the calculations combining them, as recommended by EFSA's guidance on uncertainty analysis (EFSA Scientific Committee, 2018a). For communication purposes, % probability for the more probable outcome is expressed as % certainty when reporting the conclusions, following advice in EFSA's guidance on communication of uncertainty (EFSA, 2019).

• Identification of uncertainties

A complete list of the sources of uncertainty identified by the Panel is presented in Annex D. The most important uncertainties are listed in order of importance in Appendix E, together with 'low' and 'high' scenarios describing the Panel's qualitative evaluation of their potential impact on the assessment. Reference is made in these Tables to the relevant Sections in the Opinion and/or relevant publications which provide the evidence required to support expert judgements quantifying the combined impact of uncertainties affecting different parts of the assessment (see below).

• Quantification of uncertainties affecting the RP for acute toxicity for the sum of GTX I and GTX III

Major uncertainties affecting the RP for acute toxicity for the sum of GTX I and GTX III are presented and qualitatively evaluated in Appendix E. The combined impact of these was quantified by eliciting expert judgements of the probability that the RP for acute toxicity would be higher than the assessed value of 15.3 μ g/kg bw if the uncertainties were resolved. This provides a measure of the conservativism of the assessed value. The judgements were made by the three toxicologists in the WG, who were provided with training on probability judgements and the semi-formal EKE procedure to be used. When making their judgements, the experts were asked to take into account the evidence and reasoning in the draft Opinion, together with their assessment of the identified uncertainties.

The assessment was conducted twice, first considering only the animal data and then taking into account also the evidence on human intoxications. For the first assessment, the experts were asked to make judgements on the following question: 'The proposed RP for acute toxicity is 15.3 μ g/kg bw for the sum of GTX I and III. What is your probability (expressed as %) that the RP for acute toxicity would be higher if the uncertainties affecting it were resolved, considering only animal data?'

18314722, 2023, 3, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903j.efsa.2023.7866 by CochraneItalia, Wiley Online Library on [0203/2023]. See the Terms and Conditions (https://onlinelibrary.wiley.com/rems-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons
The experts were asked to work separately to make a first judgement, considering only the major uncertainties listed in Appendix E, Table E.1A, and to provide a written summary of their reasoning. The judgements and reasoning of all 3 experts were then displayed together for a discussion facilitated by an experienced elicitor. Each expert was then asked to review their judgement and reasoning in the light of the discussion and make any changes that they considered to be appropriate, taking into account also the minor uncertainties affecting this question (listed in Appendix E, Table E.2).

The assessment was then repeated, following the same procedure and addressing the same question with the addition that the evidence on human intoxications was also taken into account (see Appendix E, Table E.1D). As interpretation of the human evidence requires consideration of the occurrence and potency of grayananes other than GTX I and III in honey, a fourth WG member with specialist expertise on these subjects was included in this second assessment. The experts concluded that, due to the large uncertainties affecting interpretation of the human evidence, it did not change the level of certainty based on the animal data alone.

The revised judgements of the four experts were all in the range 70-90% probability. After further discussion, the experts agreed on a consensus of 85% probability, considering the balance between the most important sources of uncertainty: the potential for lower values of the RP for acute toxicity in females than males, higher values for oral compared to *i.p.* exposure, and the lower bioavailability of GTX I and III when administered via honey instead of as pure substances. On this basis, taking account of all the associated sources of uncertainty, the Panel is 85%²⁸ certain that the RP for acute toxicity of 15.3 μ g/kg bw for the sum of GTX I and III is protective (i.e. would not be lower if the uncertainties were resolved).

• Quantification of uncertainties affecting potency factors for the acute toxicity of grayananes

Four groups of relative potency factors (RPFs) for the acute toxicity of grayananes are presented in Table 11 (Section 3.1.5). Major uncertainties affecting these RPFs are presented and qualitatively evaluated in Appendix E. The combined impact of these was quantified by eliciting expert judgements of the probability that the RPF for each group is conservative for all grayananes in Table 1 of the Opinion that would fall in that group. Specifically, experts were asked to make judgements on the following question: 'What is your % probability that the actual potency of all grayananes listed in Table 1 that qualify for a specified group in Table 11 would not be higher than implied by the RPF shown for that group in Table 11 of the draft opinion, if the uncertainties affecting it were resolved, not considering human data?' This was assessed separately for each of the 4 groups in Table 11.

Judgements on this question were made by four experts: the three toxicologists in the WG and a fourth WG member with expertise on assessment of relative potency. The judgements were elicited following the same process as that used for the RP for acute toxicity (described in the preceding section): initial judgements and reasoning based on the major uncertainties in Table D.1B, followed by facilitated discussion and revision, taking into account also the minor uncertainties in Appendix E.

The Panel took as its assessment the envelope (overall range) of the final judgements of the four experts. On this basis, the Panel's certainty that the RPFs shown in Table 11 are conservative is 70–80% for Group 1A, 90-95% for Group 1B, and 80-90% for Groups 2 and 3. This takes into account all the associated sources of uncertainty. The difference in certainty between groups reflects the differing amounts of data on which their RPFs are based.

• Quantification of uncertainties affecting the P95 of honey consumption

Maximum values for the P95 of acute consumption of honey for toddlers and adults who are consumers of honey are presented in Table 17. The value for each age group is the maximum of P95 values from surveys in the EFSA Comprehensive Database from a subset of EU countries. Taking the maximum of the available values is intended to provide a conservative estimate of the P95 for each age group in the whole of the EU.

Major uncertainties affecting these estimates of P95 consumption are presented and qualitatively evaluated in Appendix E, Table E.1C. The combined impact of these was quantified by eliciting expert judgements on the following question: 'What is your estimate of the 95th percentile of consumption of honey (in the toxicologically relevant period related to grayananes) by healthy people in the specified age group in the EU who are consumers of honey, excluding consumption as folk medicine, such that

²⁸ i.e. judges with 85% probability (EFSA, 2019).

you judge with 90% probability that the true 95th percentile is lower?' The question refers to the 'toxicologically relevant period' to enable the uncertainty analysis to take into account that, while the maximum P95 values shown in Table 17 relate to consumption over a full day, the period that would ideally be relevant for assessing acute exposure to grayananes may be less.

Judgements on this question were made separately for toddlers and adults, by three exposure experts in the WG. The judgements were elicited following a similar process to that used for the RP for acute toxicity (described above). First, the experts worked separately on their initial judgements and reasoning, considering the major uncertainties in Appendix E. This was followed by facilitated discussion which resulted in consensus on a probability of at least 90% that the P95 consumption of honey consumers in each age group for the whole EU is not higher than the maximum estimate shown in Table 17. The experts subsequently considered the minor uncertainties in Annex D and confirmed that these would not change their judgements.

The Panel took as its assessment the consensus judgements of the three experts. On this basis, for each of the two age groups the Panel is 90% or more certain that the 95th percentile of acute consumption of honey by EU consumers of honey, is not higher than the assessed value, i.e. 3 g/kg bw per day for toddlers and 1.22 g/kg bw per day for adults. This takes account of all identified sources of uncertainty but excludes consumption of honey as folk medicine.

The Panel did not report maximum estimates for the P95 of acute honey consumption for populations other than toddlers and adults for the reasons detailed in Section 3.4. However, based on the data that were available for other age groups (Annex C, worksheet C.2 and C.3), the experts judged with 90-95% probability that the P95 of acute honey consumption for infants, adolescents and other children in the EU who are consumers of honey would be lower than 3 g/kg (the estimated value for toddlers). Similarly, the experts judged with 95-99% probability that the P95 of acute honey consumption for the elderly and very elderly in the EU who are consumers of honey would be lower than 3 g/kg bw, because the available estimates for these age groups are much lower. The Panel did not quantify the uncertainty of the P95 for pregnant or lactating women or vegetarians, because the few surveys available (0-2), did not contain enough consumption events for honey.

• Quantification of uncertainties affecting the MOEs for acute toxicity

The limitations described under Section 2.2 did not allow the use of data on the occurrence of grayananes in 'certain honey' for risk characterisation. Regarding the limited data on occurrence and consumption of *Rhododendron* honey, the Panel calculated MOEs for acute exposure of adults and toddlers using selected concentrations for the sum of GTX I and III reflecting concentrations measured in 'certain honey' (Table 16).

These MOEs were calculated as the ratio of the RP of 15.3 μ g/kg bw to the estimated exposure, based on selected concentrations and the P95 of acute honey consumption for each age group from dietary surveys (see Section 3.4.1.1). Therefore, the uncertainty of the RP must be combined with the uncertainty of the P95 consumption for the EU population in each age group to assess the uncertainty of each MOE. The selected concentrations are fixed values (not subject to uncertainty), as the calculated MOEs are conditional on them.

The combination of uncertainties relating to the RP and acute consumption was performed by probability bounds analysis (Section 14.1 of EFSA (2018b)). This method for combining uncertain quantities is described in section 14.1 of EFSA (2018a) and in more detail in Annex B.13 of EFSA (2018b). The advantage of this method is that it requires only a probability bound for each quantity: it does not require elicitation of complete probability distributions and allows for any degree of dependence of uncertainty between the quantities. It is conservative in the sense that it overestimates the probability of values outside the resulting bounds for the MOE, compared to a more refined probabilistic assessment, but avoids the need for refined methods when the conservative probabilities are sufficient for decision-making (EFSA Scientific Committee (2018b), page 180).

The MOE is monotonically increasing with respect to the RP and the reciprocal of P95 consumption (1/P95), therefore the probabilities for lower values of each can be combined to obtain a probability for lower values of the MOE using the second of the two simplest methods for probability bounds analysis, described on page 177 of EFSA (2018b). Specifically,

- a) The Panel's assessment of 85% probability that the RP is above the assessed value was subtracted from 100%, to obtain the Panel's probability that it is lower: 15%.
- b) The Panel's % probability of \geq 90% for each age group (toddlers and adults) that the P95 consumption is below the assessed value was subtracted from 100%, to obtain the Panel's

probability that it is higher: \leq 10%. The Panel's probability for higher values of the reciprocal, 1/P95, is therefore also \leq 10%.

- c) Applying the method of probability bounds analysis (EFSA, 2018b), the probabilities from steps (a) and (b) are summed to obtain a maximum probability that the MOE would be lower than the assessed value if the uncertainties affecting the RP for acute toxicity and the estimated P95 consumption were resolved: $15\% + \le 10\% = \le 25\%$.
- d) The probability from step c) was subtracted from 100% to obtain a minimum probability, for each age group, that the MOE would be higher than the assessed value: \geq 75%.

Therefore, taking account of the identified uncertainties, the Panel is 75% or more certain for toddlers and for adults that the MOE shown in Table 16 is conservative, i.e. would not be lower if the uncertainties affecting the RP and P95 consumption estimates were resolved. The Panel noted that the result of the probability bounds analysis is a minimum probability: a higher probability than \geq 75% would be obtained if full distributions were elicited for the RP and the estimated P95 consumption and combined by Monte Carlo calculations, assuming them to be independent (as is likely in this case). However, this would require additional work, which is unnecessary if the minimum probability is sufficient for decision-making.

The Panel did not calculate MOEs for age groups other than toddlers and adults. However, the Panel judged with 90-95% probability that the P95 of acute honey consumption for infants, other children and adolescents in the EU would be lower than 3 g/kg bw, and 95–99% probability for the elderly and very elderly (see earlier). Substituting the judgement of 90–95% probability for infants, other children and adolescents into the probability bounds calculation described above, in place of the > 90% probability that the P95 for toddlers is below the same value (3 g/kg bw), results in a probability of 75% or more that the MOE for toddlers is also conservative for infants, other children and adolescents. Similarly, substituting the judgement of 95–99% probability for the elderly and very elderly into the probability bounds calculation results in a probability of 80% or more that the MOE for toddlers is also conservative for the elderly and very elderly. The Panel noted that the calculated MOEs are based on acute toxicity of GTX I and III and do not consider the presence of other grayananes in 'certain honey'. In addition, they may not be protective for the genotoxicity endpoint.

• Quantification of uncertainties affecting the *highest concentrations* in honey below which no acute effects are expected

The Panel calculated *highest concentrations* of GTX I and III in 'certain honey' below which *no acute effects are expected* based on the P95 of acute honey consumption. For toddlers and adults these were 0.05 mg/kg honey and 0.12 mg/kg honey, respectively. These were calculated using the RP for acute toxicity, an MOE of 100 incorporating intra and inter-species variability (EFSA Scientific Committee, 2012) and the estimated P95 of acute honey consumption for each age group, based on dietary surveys (see Section 3.5.1). Therefore, the uncertainty of the RP for acute toxicity must be combined with the uncertainty of the P95 consumption to assess the uncertainty of the *highest concentration*. This combination was performed by the same probability bounds analysis calculation as was described for the MOEs in the preceding section. (Section 14.1 of EFSA (2018b)). Since the Panel's probabilities for the RP for acute toxicity are the same, the combined probability for the *highest concentrations* is the same.

Consequently, taking account of the identified uncertainties, the Panel is 75% or more certain for both toddlers and adults that the *highest concentrations* shown in Table 17 are protective, i.e. would not lead to a concern for acute effects at the P95 of normal honey consumption by that age group in the EU. The Panel noted again that the result of the probability bounds analysis is a minimum probability.

The Panel did not calculate *highest concentrations* for age groups other than toddlers and adults. However, substituting the experts' judgements about P95 consumption for other age groups into the probability bounds analysis conducted for toddlers and adults, in the same way as was done for MOEs (above), the Panel was 75% or more certain that the *highest concentration* of 0.05 mg/kg for toddlers is also protective for infants, adolescents and other children, and 80% or more certain that it is also protective for the elderly and very elderly.

The Panel also calculated *highest concentrations* for indicative portion sizes of 'certain honey' used as folk medicine (Table 18). These highest concentrations are indicative because they refer to portion sizes recommended on the labels of selected products. There were insufficient data for the Panel to quantify uncertainties regarding variation of recommended portion sizes on product labels or the actual portion sizes consumed by users of those products. Instead, the Panel quantified the uncertainty of the *highest concentrations* for folk medicine use conditional on the specific portion sizes considered. Therefore, the specified portion size was considered as fixed and the Panel's certainty that the corresponding *highest concentration* is protective is determined solely by the Panel's certainty of 85% that the RP for acute toxicity combined with an MOE of at least 100 is conservative. Consequently, the same level of certainty applies to each of the indicative portion sizes shown in Table 18. For example, for a portion size of 66 g, the Panel is 85% certain that concentrations of up to the calculated *highest concentration* of 0.16 mg/kg would not lead to acute effects, taking into account the uncertainty of that level. For portion sizes larger than 66 g, proportionately lower highest concentrations would be needed.

The Panel noted that the calculated *highest concentrations* are based on acute toxicity of GTX I and III and do not consider the presence of other grayananes in 'certain honey'. In addition, they may not be protective for the genotoxicity endpoint.

183

Appendix E – Lists of uncertainties

Table E.1: Uncertainties impacting the result from acute exposure to 'certain honey' (ranked per order of priority)

A. Uncertainties relating to the accuracy of the RP for acute toxicity derived for the sum of GTX I and III (ranked per order of priority)				
Uncertainty	Scenario of maximum probability that the RP for acute toxicity is higher (lower risk)	Scenario of maximum probability that the RP for acute toxicity is lower (higher risk) Possibility that the <i>i.p.</i> and oral routes are equivalent		
ADME:	Bioavailability via oral route might be lower than via the			
Insufficient information on the absorption of GTX I and	<i>i.p.</i> route which would result in lower levels in blood and a higher RP for acute toxicity.	considering that injection via <i>i.p.</i> leads the substance through the liver and that non-lipophilic compounds		
	Oral bioavailability of GTX III when the exposure to the substance is from honey would be lower than when experimental animals are exposed to the pure substance.	(such as GTX III) tend to be less absorbed by <i>i.p.</i> (see Turner et al., 2011; Al Shoyaib et al., 2020).		
Toxicity studies - critical study design:		Experience shows that females can be slightly more		
Studies carried out only in one gender		sensitive to the acute toxic effect than males (OECD Guidelines 420). This would result in the derivation of a		
See Section 3.1.2, Tables 6 and 7		lower RP for acute toxicity.		
Toxicity studies - critical endpoints:	LD_{50} for GTX I and III (oral) is approx. 5 mg/kg bw, or	Results from <i>i.p.</i> and oral exposure to GTX I and III in		
Extrapolation from <i>i.p.</i> to oral administration and uncertainties associated with it, i.e. species difference and mortality	\sim 5× LD ₅₀ for GTX I and III (oral) which is approx. 1 mg/kg bw (Hikino et al., 1976, 1979).	the study by Hikino et al. (1976 and 1979) suggest that $LD_{50}s$ of GTX I and III are of the same order of magnitude. Therefore, if studies were not based on mortality, but lower doses were used in the study, then		
Supporting evidence: – Critical studies by Hikino et al. (1976) – <i>i.p</i> . injection to mice, and		the true <i>i.p.</i> /oral ED ₅₀ s for a specific effect might not differ that much (for oral and for <i>i.p.</i> , exposures at 0.5 and 0.1 μ g/kg bw, respectively), i.e. for exposures		
Hikino et al., (1979) – oral administration of GTX III to mice		two values are similar.		
– Table 6				
Dose-response analysis of critical endpoints:	Statistics and confidence intervals (CI) are reported only	Statistics and confidence intervals (CI) are reported only in graphs and not in Tables, therefore, data were extracted manually. If response values were overestimated, then they would give a lower RP for acute toxicity.		
Lack of raw data points	In graphs and not in Tables in the paper, therefore, data were extracted manually from the graph. If the response			
Supporting evidence: – Critical study by Turkmen et al., 2013 – <i>i.p.</i> injection of GTX III in rats	values were underestimated, then this would result in a higher RP for acute toxicity.			



A. Uncertainties relating to the accuracy of the RP for acute toxicity derived for the sum of GTX 1 and 111 (ranked per order of priority)						
Uncertainty	Scenario of maximum probability that the RP for acute toxicity is higher (lower risk)	Scenario of maximum probability that the RP for acute toxicity is lower (higher risk)				
Dose-response analysis of critical endpoints	If the analysis was performed at 90 min exposure time	If the analysis was performed with a BMR of 5% (instead of 10%), then the BMDL would have been 3.89 μ g/kg bw or around 4× lower and would result in a lower RP for acute toxicity.				
Choice of a BMR of 10% for continuous data instead of 5% and an exposure time of 60 min, considering the transient nature of the effect. The analysis was based the reduction of heart rate after 60 min of <i>i.p.</i> exposure to GTX III.	(instead of 60 min), then the BMDL ₁₀ would be 65.8 μ g/kg bw or around 4× higher than the BMDL ₁₀ of 15.3 μ g/kg bw (the cut-off criteria on the ratios BMDU/BMDL and Lowest dose (LD)/BMD are met for both BMD calculations). This would have resulted in a higher RP for acute toxicity.					
Supporting evidence: – Section 3.1.6.1 – Appendix C – Turkmen et al., 2013	For 60 and 90 min of <i>i.p.</i> exposure to GTX III, the values for heart rate (bpm) are comparable for the two highest doses (400 and 800 μ g/kg bw) but for the lowest dose (200 μ g/kg bw), the values for bpm \pm SE* after 90 min are closer to the ones of the control group than after 60 min of exposure.					
	*SE: Standard Error					
B. Uncertainties relating to the potency of all oth	er grayananes found in `mad honey', relative to GT	K I and III (ranked per order of priority)				
Uncertainty	Scenario of maximum probability that the RPF per group of grayananes in Table 11 should be higher than the one in the table (lower risk)	Scenario of maximum probability that the RPF per group of grayananes in Table 11 should be lower than the one in the table (higher risk)				
ADME:	If the total grayanane bioavailability is lower than the	If the total grayanane bioavailability is higher than the				
Insufficient information on the absorption of the different grayananes	bioavailability of GTX III, the combined RP for acute toxicity and highest concentration will be higher.	bioavailability of GTX III, the combined RP for acute toxicity and highest concentration will be lower.				
Supporting evidence: Section 3.1.1						
Mixture group membership and interactions:	The kaurane structure is stable and no cleavage of the	The kaurane structure is stable and no cleavage of the				
Lack of information on toxicity of metabolites of	Viayanane is expected (see Section 5.1.1).	grayanane is expected (see Section 3.1.1). Metabolic conversion from grayananes with less or no structural alerts to grayananes with structural alerts would increase potency.				
Supporting evidence: Section 3.1.1	decrease potency.					

A. Uncertainties relating to the accuracy of the RP for acute toxicity derived for the sum of GTX I and III (ranked per order of priority)



Uncertainty	Scenario of maximum probability that the RPF per group of grayananes in Table 11 should be higher	Scenario of maximum probability that the RPF per group of grayananes in Table 11 should be lower than the one in the table (higher risk) Grayananes identified in honey have been ranked in the	
-	than the one in the table (lower risk)		
Substances used and read across:	Grayananes identified in honey have been ranked in the		
Calculation/use of potency factors	tollowing groups:	Tollowing groups:	
Supporting evidence:	 Subgroup 1A: rhodojaponin III, craiobiotoxin II and rhodomollein XVIII. An LD₅₀ is available for 	 Subgroup 1A: rhodojaponin III, craiobiotoxin II and rhodomollein XVIII. An LD₅₀ is available for rhodojaponin III suggesting that the substance is twice more toxic than GTX I and III. No experimental data are available on craiobiotoxin II and rhodomollein XVIII (ranked to this group based on their structural alerts). If the two substances are more toxic than rhodojaponin III then they should have a higher RPF than the one in Table 11. 	
 Table 1, Section 1.2.1, Section 3.1.5, Table 11 Appendix A and Section 3.1.2, Table 6 	rhodojaponin III suggesting that the substance is twice more toxic than GTX I and III. No experimental data are available on craiobiotoxin II and rhodomollein XVIII (ranked to this group based on their structural alerts). If the two substances are less toxic than rhodojaponin III then they should have a lower RPF than the one in Table 11.		
	 Subgroup 1B: GTX I, III, VI, Kalmitoxin I, and rhodojaponin VI. LD₅₀ values are available for these grayananes and all of them appear to exhibit lower toxicity than GTX I and III, with the exception of GTX VI. 	 Subgroup 1B: GTX I, III, VI, Kalmitoxin I, and rhodojaponin VI. LD₅₀ values are available for these grayananes and all of them appear to exhibit lower toxicity than GTX I and III, with the exception of GTX VI which appears to be as toxic as GTX III. 	
	– Subgroup 2: GTX II, IV, VII, VIII, XVII, rhodomollein XIII and XIX, craiobiotoxin I and V. LD ₅₀ value is available for GTX II only, all other substances were ranked based on their structural alerts. This may overestimate the potency for these substances.	 Subgroup 2: GTX II, IV, VII, VIII, XVII, rhodomollein XIII and XIX, craiobiotoxin I and V. LD₅₀ value is available for GTX II only, all other substances were ranked based on their structural alerts. This may underestimate the potency for these substances. 	
	 Subgroup 3: GTX XVI, craiobiotoxin VIII, and rhodomollein XI. LD₅₀ information is available for GTX XVI but not for the other substances which were ranked based on their structural alerts. This may overestimate the potency for these substances but still, their overall contribution would be negligible. 	 Subgroup 3: GTX XVI, craiobiotoxin VIII, and rhodomollein XI. LD₅₀ information is available for GTX XVI but not for the other substances which were ranked based on their structural alerts. This may underestimate the potency for these substances but still, their overall contribution would be negligible. 	



1831

C. Uncertainties relating to the P95 for acute consumption of normal honey used to calculate *the highest concentration(s)* for the sum of GTX I and III in honey not resulting in a concern for acute effects (ranked per order of priority)

Uncertainty	Scenario of maximum probability that consumption is lower (lower risk)			consumption is	Scenario of maximum probability that consumption is higher (higher risk)
Representativeness of EU surveys to the EU population	Twenty-two (22) adults which wou adult honey cons surveys for toddle honey consumpti particularly subje percentiles for ad one estimated in Table 17 are over	c) (22) surveys from 22 EU countries are available for ch would suggest a reliable estimate for the overall y consumption in the EU. The few available national r toddlers (5 surveys) cannot capture the overview of sumption in these EU populations and thus it is y subjected to uncertainty. If the true highest 95th is for adults and toddlers in the EU are lower than the ated in this opinion, then the highest consumptions in re overestimated and so it is the risk.		ries are available for nate for the overall w available national oture the overview of s and thus it is true highest 95th J are lower than the nest consumptions in risk.	Twenty-two (22) surveys from 22 EU countries are available for adults which would derive a reliable estimate for the overall adult honey consumption in the EU. The few available national surveys for toddlers (5 surveys) cannot capture the overview of honey consumption in these EU populations and thus it is particularly subjected to uncertainty. If the true highest 95th percentiles for adults and toddlers in the EU are higher than the one estimated in this opinion, then the highest consumptions in Table 17 are underestimated and so it is the risk.
	Consumption sum included in the as robust (EFSA, 20	veys with less tha ssessment since re 11).	n 60 sub esults ma	vject days were not ay not be statistically	Consumption surveys with less than 60 subject days were not included in the assessment since results may not be statistically robust (EFSA, 2011).
One day consumption vs one eating occasion	EFSA assesses ac amount consume occasion due to u occasions/meals. between wide int may be lower. Ba after a certain tin less than 1 day th overestimate the <i>highest concentra</i> Consumption per associated with in per day.	esses acute consumption based on the total food onsumed per day rather than each individual eating due to uncertainties in distinguishing between eating /meals. If honey is consumed in smaller portions wide intervals during the day the risk of intoxication wer. Based on data on intoxications, subjects recover rtain time period which is unknown. When this period is 1 day then data from 1-day consumption would hate the risk and result in the estimation of lower <i>oncentrations</i> . tion per one-eating occasion (which is the one mainly d with intoxications) is slightly lower than consumption		n the total food h individual eating ing between eating smaller portions risk of intoxication ns, subjects recover n. When this period is umption would nation of lower ch is the one mainly rer than consumption	EFSA assesses acute consumption based on the total food amount consumed per day rather than each individual eating occasion due to uncertainties in distinguishing between eating occasions/meals. If honey is consumed in smaller portions between wide intervals during the day the risk of intoxication may be lower. Based on data on intoxications, subjects recover after a certain time period which is unknown. When this period is more than 1 day then data from 1-day consumption would underestimate the risk and result in the estimation of higher <i>highest concentrations</i> .
	Sub-population No of surveys Max P95 (g/kg bw)		95 (g/kg bw) /eating occasion		
	Infant	1	0.83	0.83	
	Toddler	5	3.00	2.00	
	Adult	22	1.22	1.18	



183

C. Uncertainties relating to the P95 for acute consumption of normal honey used to calculate *the highest concentration(s)* for the sum of GTX I and III in honey not resulting in a concern for acute effects (ranked per order of priority)

Uncertainty	Scenario of maximum probability that consumption is lower (lower risk)			ario of maximum probability that consumption is ar (higher risk)
Data reporting Uncertainties related to the methodologies of the dietary surveys, over/under-reporting, etc.	In some surveys consumers might have reported honey consumption per teaspoon or tablespoon and not in grams. Conversion to grams is subject to uncertainty due to differen conversion factors that could be used (e.g. search on the we regarding standard portions of honey, gave values for 1 teaspoon (7.08 g) and 1 tablespoon (21.25 g) which are diffe from the values of 8 g and 22 g, respectively, used in this assessment. It is, therefore, possible that the overall honey consumption for the EU population is lower and that the risk overestimated.		In some surveys consumers might have reported honey consumption per teaspoon or tablespoon and not in grams. Conversion to grams is subject to uncertainty due to different conversion factors that could be used (e.g. search on the web regarding standard portions of honey, gave values for 1 teaspoon (7.08 g) and 1 tablespoon (21.25 g) which are different from the default values of 8 g and 22 g, respectively, used in this assessment. It is, therefore, possible that the overall honey consumption for the EU population is higher and that the risk is underestimated.	
D. Uncertainties relating to human	observations	(ranked per order of priority)		
Uncertainty		Scenario of maximum probability that the RP for acute toxicity when combined with an MOE of at least 100 will be higher (lower risk)		Scenario of maximum probability that the RP for acute toxicity when combined with an MOE of at least 100 will be lower (higher risk)
Chemical composition: Composition of grayananes in the ingested material is incomplete (no/limited information on the identity/ levels of grayananes other than GTX I and III in honey). Supporting evidence: Section 3.2.1, Table 14		Total grayanane conc. in honey > conc. of the sum of GTX I and III. This suggests that the 'LOAEL' of 4.8 μ g/kg bw for GTX I and III based on data from Aygun et al. (2015) reflects the toxicity of all grayananes in honey and not only the toxicity of GTX I and III. Therefore, the 'LOAEL' for the sum of all GTX present should be the same (if no other grayananes in addition to GTX I and II are present) or higher (if other grayananes are present i addition to GTX I and III). This would result in the same or higher RP for acute toxicity for the sum of GTX I and III. Based on semi-quantitative determinations, total grayananes in honey may occur at levels up to 3 or 7 times higher than the levels for the sum of GTX I and III.		



D. Uncertainties relating to human observations (ranked per order of priority)				
Uncertainty	Scenario of maximum probability that the RP for acute toxicity when combined with an MOE of at least 100 will be higher (lower risk)	Scenario of maximum probability that the RP for acute toxicity when combined with an MOE of at least 100 will be lower (higher risk)		
Substances used and read across:	With the exception of rhodojaponin III, all grayananes	Some honeys have high levels of rhodojaponin III (see		
Calculation of a group effect of no concern for acute toxicity/use of potency factors	to have similar or lower potency than GTX I and III (see Section 3.1.5 and Appendix A).	Rhododendron honeys from Nepal may contain high		
 Table 1, Section 1.2.1, Section 3.1.5, Table 11 and Appendix A Section 3.2.1, Table 15 	<i>Rhododendron</i> honeys from the Turkish Black Sea region are rich in GTX I and III and contain low or no levels of rhodojaponin III.	levels of rhodojaponin 111.		
Severity of the effect: Uncertainty relating to low dose-effects after consumption of 'certain honey'		The available data refer to intoxication reports, i.e. after severe life-threatening effects following consumption of mad honey; relevant reports on acute adverse effects in humans from consumption of 'mad honey' which do not end to the hospital are not found. If adverse (but not life-threatening) effects are manifested in lower doses, then a lower RP for acute toxicity would apply for grayananes.		
		Adverse low dose effects may be unspecific and not seen in causal relationship with honey consumption by affected persons.		

Table E.2: List of minor uncertainties

Main group	Sub-group	Overarching questions	Sources of uncertainty in the opinion on grayananes in `certain honey'
Chemical composition and analytical methods	Dosing and chemical composition	Uncertainty in the chemical composition of the material administered to the experimental animals.	a) Purity of the single substance tested not always defined in the studies.b) The impurities and unidentified compounds of the tested substance, and/or botanical material and/or the honey tested
			Physicochemical properties of GTX III versus the hemi (ethyl acetate), e.g. the free GTX III (may be less soluble and bioavailable) (read across).
	Analytical methods		Certified reference standards not commercially available for most grayananes.



Main group	Sub-group	Overarching questions	Sources of uncertainty in the opinion on grayananes in `certain honey'
Hazard identification and characterisation	ADME	Is there uncertainty in any aspect of ADME?	Uncertainty related to the elimination of grayananes.
	Mixture group membership and interactions	Is there uncertainty on the extent and profile of effects due to co-exposure (e.g. metabolites, interaction of chemicals, combined effects)?	Interaction/combined effects between grayananes and with other substances.
	Selection of reference point from Animal studies	Are there uncertainties in the selection of the RP that are not covered by the BMD confidence interval e.g. is model uncertainty covered?	Uncertainty in the confidence intervals of the selected BMR.
Consumption data	Data reporting	Is there uncertainty in the consumption data due to e.g. classification, bw, memory errors etc.?	Memory errors and capacity to report details in dietary surveys, possible under and over-reporting.
		Is there uncertainty in consumption data, e.g. due to methodology of the dietary survey, weekdays, national recipes?	 Dietary survey methodology (dietary record vs 24-hour recall), dietary software, interview options (place, face to face vs telephone and background of the interviewers) and use of portion-size measurement aids for the estimation of portion sizes. Sampling frame, method and design of the dietary surveys. Representativeness over different weekdays and seasons within dietary survey. Sample size and response rate of the dietary surveys.
	Representativeness of the data	Is there uncertainty in consumption data, e.g. representativeness of the countries, special population groups, sample size and response rates.	Sampling frame, method and design of the dietary surveys.
			Representativeness over different weekdays and seasons within dietary surveys.
			Availability of adequate number of consumption surveys and adequate number of subject days per age group.
			Availability of food consumption data for the general population, including consumers only of specific foods of special interest.
			Sample size and response rate of the dietary surveys.
Exposure estimates	Statistical models	Is there uncertainty in consumption data, e.g. due to methodology of the dietary survey, weekdays, national recipes?	Consumers loyalty to specific brands or products not considered, local consumption.
Risk characterisation	Uncertainty factors	Is there uncertainty on the selected (default or specific) UFs?	Inter-species differences not correctly covered by standard UFs (for animal data only).
			Use of uncertainty factors to cover data gaps.

Annex A – Protocol

The file containing the protocol on grayananes in 'certain honey' is available under the Supporting Information section on the online version of the scientific output.

Annex B – Plant list of Ericaceae species

The excel file containing the plant list of Ericaceae species that have been investigated for grayanane content or that have been listed for ethnopharmacological use is available on the EFSA Knowledge Junction community on Zenodo at: https://doi.org/10.5281/zenodo.7673435

Annex C – Consumption data statistics

The excel file containing the information on the dietary surveys included in the assessment, the detailed acute consumption statistics per country, survey and age group calculated in mg/kg bw per day and in mg/kg bw per eating event is available on the EFSA Knowledge Junction community on Zenodo at: https://doi.org/10.5281/zenodo.7673456

Annex D – Uncertainty analysis: EKE Protocol

The file containing the EKE protocol on grayananes in 'certain honey' is available under the Supporting Information section on the online version of the scientific output.